MORRISON & FOERSTER LLP

1290 Avenue of the Americas

New York, New York 10104

Telephone: (212) 468-8000 Facsimile: (212) 468-7900

Larren M. Nashelsky

Gary S. Lee

Lorenzo Marinuzzi

Proposed Counsel for the Debtors and Debtors in Possession

## IN THE UNITED STATES BANKRUPTCY COURT FOR THE SOUTHERN DISTRICT OF NEW YORK

	)	
In re:	)	Case No. 12-
	)	
RESIDENTIAL CAPITAL, LLC, et al.,	)	Chapter 11
, ,	)	•
Debtors.	)	Joint Administration Pending
	)	_

DEBTORS' MOTION FOR INTERIM AND FINAL ORDERS UNDER BANKRUPTCY CODE SECTIONS 105(a) AND 363(b) AUTHORIZING RESIDENTIAL CAPITAL, LLC TO ENTER INTO A SHARED SERVICES AGREEMENT WITH ALLY FINANCIAL INC. NUNC PRO TUNC TO THE PETITION DATE FOR THE CONTINUED RECEIPT AND PROVISION OF SHARED SERVICES NECESSARY FOR THE OPERATION OF THE DEBTORS' BUSINESSES

The debtors and debtors in possession in the above-captioned cases (collectively, the "Debtors")<sup>1</sup> hereby move for entry of interim and final orders, under sections 105(a) and 363(b) of title 11 of the United States Code (the "Bankruptcy Code") authorizing Residential Capital, LLC ("ResCap") to enter into a shared services agreement (the "Agreement") with its indirect non-debtor parent, Ally Financial Inc. ("AFI") *nunc pro tunc* to the Petition Date for the

The names of the Debtors in these cases and their respective tax identification numbers are identified on <a href="Exhibit 1">Exhibit 1</a> to the Whitlinger Affidavit (defined below). Additional subsidiaries and affiliates of the Debtors may file Chapter 11 petitions on a rolling basis. As used herein, the term "Debtors" includes any such entities.



continued receipt and provision of shared services between ResCap and AFI necessary for the continued operation of the Debtors' businesses (the "Motion").<sup>2</sup> In support of the Motion, the Debtors rely upon and incorporate by reference the Affidavit of James Whitlinger, Chief Financial Officer of Residential Capital, LLC, in Support of Chapter 11 Petitions and First Day Pleadings, filed with the Court concurrently herewith (the "Whitlinger Affidavit"). In further support of the Motion, the Debtors, by and through their undersigned counsel, respectfully represent:

#### **JURISDICTION**

1. This Court has jurisdiction to consider this Motion under 28 U.S.C. §§ 157 and 1334. This is a core proceeding under 28 U.S.C. § 157(b). Venue of these cases and this Motion in this district is proper under 28 U.S.C. §§ 1408 and 1409. The statutory predicates for the relief requested herein are Bankruptcy Code sections 105(a) and 363(b).

#### **BACKGROUND**

- 2. On the date hereof (the "Petition Date"), each of the Debtors filed a voluntary petition in this Court for relief under Chapter 11 of the Bankruptcy Code. The Debtors are managing and operating their businesses as debtors in possession pursuant to Bankruptcy Code sections 1107(a) and 1108. No trustee, examiner or statutory creditors' committee has been appointed in these Chapter 11 cases.
- 3. The Debtors are a leading residential real estate finance company indirectly owned by AFI, which is not a Debtor. The Debtors and their non-debtor affiliates operate the fifth largest servicing business and the tenth largest mortgage origination business in

Creditors and parties-in-interest with questions or concerns regarding the Debtors' Chapter 11 cases or the relief requested in this Motion may refer to <a href="http://www.kccllc.net/rescap">http://www.kccllc.net/rescap</a> for additional information.

the United States. A more detailed description of the Debtors, including their business operations, their capital and debt structure, and the events leading to the filing of these bankruptcy cases, is set forth in the Whitlinger Affidavit.

- 4. The Debtors' primary and most valuable business operations consist of servicing mortgage loans for investors, including loans originated by the Debtors, Ally Bank, and other third parties. As of March 31, 2012, the Debtors were servicing over 2.4 million domestic mortgage loans with an aggregate unpaid principal balance of approximately \$374.2 billion. To preserve and realize the value of these assets and achieve the goals of these Chapter 11 cases, the Debtors developed and are prepared to implement a strategy that provides maximum value to the Debtors' estates.
- 5. The Debtors negotiated and entered into two separate asset purchase agreements. The first, with Nationstar Mortgage LLC as the proposed stalking horse bidder ("Nationstar") for the sale of their mortgage loan origination and servicing businesses (the "Platform Sale"), and the second, with AFI as the proposed stalking horse bidder for the sale of their legacy portfolio consisting mainly of mortgage loans and other residual financial assets (the "Legacy Sale" and collectively with the Platform Sale, the "Asset Sales").
- 6. In furtherance of their restructuring strategy, and contemporaneous with the commencement of these Chapter 11 cases, the Debtors have filed a motion for authority to, among other things, establish auction and sale procedures for the Asset Sales,<sup>3</sup> and for approval

See Debtors' Motion Pursuant to 11 U.S.C. §§ 105, 363(b), (f), and (m), 365 and 1123, and Fed. R. Bankr. P. 2002, 6004, 6006, and 9014 For Order: (A)(I) Authorizing and Approving Sale Procedures, Including Break-Up-Fee and Expense Reimbursement; (II) Scheduling Bid Deadline and Sale Hearing; (III) Approving Form and Manner of Notice Thereof; and (IV) Granting Related Relief and (B)(I) Authorizing the Sale of Certain Assets Free and Clear of Liens, Claims, Encumbrances, and Other Interests; (II) Authorizing and Approving Asset Purchase Agreements Thereto; (III) Approving the Assumption and Assignment of Certain Executory Contracts and Unexpired Leases Related Thereto; and (IV) Granting Related Relief.

to consummate the Asset Sales under a plan. If, however, the Debtors do not obtain confirmation of a plan by deadlines to be determined, then the Sale Motion allows the Debtors to pursue an alternative course of action and immediately move forward with the Asset Sales under Bankruptcy Code section 363(b) and outside of a plan.

#### PRELIMINARY STATEMENT

- The Agreement is critical to the continued operation of the Debtors' businesses during the pendency of these Chapter 11 cases, specifically in connection with the Asset Sales discussed above. The services that the Debtors seek to continue providing to and receiving from AFI during these Chapter 11 cases pursuant to the Agreement include, among other things: (i) information technology services; (ii) employee benefits administration and other human resources functions; (iii) accounting, tax and internal audit services; (iv) treasury and collateral management; (v) risk management functions; (vi) supply chain management, including procurement of goods and services from third parties; (vii) government and regulatory relations and compliance services; (viii) facilities management services; (ix) marketing services; and (x) capital markets services relating to managing the value of certain of the Debtors' loan servicing rights. Given the integrated nature of the Debtors' and AFI's businesses, the continuation of these services pursuant to the Agreement is both warranted and absolutely necessary to avoid any disruption to the Debtors' day-to-day operations.
- 8. As described above, the Debtors are seeking authority to sell substantially all of their assets pursuant to a plan, or in the alternative, through a section 363 sale. As set forth in the asset purchase agreements, Nationstar and AFI require as a condition to the closing of the Asset Sales that the Debtors operate their businesses in the ordinary course and use commercially reasonable efforts to preserve the value of their business enterprise. The Debtors are optimistic

that the sale of substantially all of their assets will yield a significant return for the Debtors' estates. To comply with the provisions of the asset purchase agreements with Nationstar and AFI and preserve the value of their businesses, the Debtors must avoid significant disruptions to their operations during this critical postpetition period. Accordingly, the Debtors seek authority for ResCap to enter into and perform under the Agreement with AFI to ensure the continued receipt and provision of the essential services included in the Agreement. As explained in detail below, the relief requested is in the best interests of the Debtors' estates and amply warranted under the circumstances.

#### **RELIEF REQUESTED**

- 9. By this Motion, the Debtors seek entry of interim and final orders pursuant to Bankruptcy Code sections 105(a) and 363(b) authorizing ResCap to (i) enter into the Agreement substantially in the form attached hereto as <a href="Exhibit A">Exhibit A</a> nunc pro tunc</a> to the Petition Date; (ii) provide services to AFI in accordance with the Agreement; and (iii) pay AFI for services received in accordance with the Agreement. The Debtors are seeking Court approval of the Agreement nunc pro tunc to the Petition Date; however, by this Motion, the Debtors are also seeking to perform under the Agreement as of the Petition Date upon the entry of an interim order and subject to a final hearing. Only ResCap and AFI are parties to the Agreement; however, each party is required to work with its respective subsidiaries to ensure that all of the services are provided and each of the ResCap's Debtor affiliates and AFI's non-debtor affiliates receive services as applicable.
- 10. The Debtors request that, to the extent necessary, the relief sought by this Motion apply to any future debtor (a "Future Debtor") in these jointly-administered cases without any further order of the Court. The Debtors propose that a debtor be deemed to be a

Future Debtor upon the Court's entry of an order authorizing the joint administration of such Future Debtor's Chapter 11 case with the Chapter 11 cases of the Debtors.

#### A. The Agreement

- and certain of their affiliates provided various financial, operational and administrative services to each other, which the Debtors seek to continue on and after the Petition Date pursuant to the Agreement. Historically, the services that are now subject of the Agreement were provided on an undocumented basis. Although these arrangements were suitable, the Debtors determined it was in their best interests to compile a comprehensive and integrated agreement, to become effective postpetition, to ensure that (i) the Debtors obtain and pay for only those services that are necessary during their Chapter 11 cases; (ii) the services that are being provided between ResCap and AFI are specifically identified; and (iii) ResCap may reduce or terminate the receipt of services at any time, including, without limitation, following the closing of a sale of substantially all of the Debtors' assets.
- operations of AFI are an integrated business involving the Debtors and Ally Bank. The Debtors believe the services covered by the Agreement are common among enterprises similar to those of the Debtors and their non-debtor affiliates, especially given the integrated nature of their businesses, because these arrangements eliminate redundant functions, reduce costs and allow for the realization of operational synergies. The Debtors are seeking the Court's authority to allow ResCap to enter into the Agreement with AFI so that these services can be continued postpetition to allow for a smooth transition to operating as a debtor in possession, facilitate the Asset Sales, and complete a wind-down of the Debtors' businesses following a sale.

13. The following are the salient terms of the Agreement:<sup>4</sup>

#### **Term**

The initial term of the Agreement begins on the date the Agreement is approved by the Bankruptcy Court (the "Effective Date")<sup>5</sup> and continues through midnight Eastern Time on the one (1) year anniversary of the Petition Date (the "Initial Term"), unless earlier terminated or extended in accordance with the terms of the Agreement and except for any Service that has been terminated in accordance with the terms of the Agreement. The Initial Term will be automatically extended for additional periods of one (1) year each unless either Party provides to the other Party written notice of nonrenewal at least three (3) months prior to the expiration of the then-current Term. The Initial Term as extended by any such additional period(s) (if any) shall be referred to as the "Term."

#### Services

AFI will provide the Services within each of the Functional Service Areas listed in Schedule A-1 (the "Parent Services"), and ResCap will provide the Services within each of the Functional Service Areas listed in Schedule A-2 (the "Reverse Services"), in each case beginning on the Effective Date. Services provided by a Supplier under the Agreement may be provided by that Supplier directly or through any of its Affiliates and/or Subcontractors at Supplier's discretion. A Supplier will not be relieved of any of its obligations under the Agreement as a result of the provision of Services by any of Supplier's Subsidiaries or other Supplier Personnel. Supplier agrees that it will not enter into any new Subcontracting arrangement that would involve the transfer to a Subcontractor or another third party of Recipient's personal information, without the prior written consent of the other Party. The Supplier Personnel providing Services will at all times be qualified to provide the Services assigned to them.

### Additional Services

A Party may, from time to time during the Term, upon at least ninety (90) days prior written notice to the other Party, request that the other Party provide additional services, functions and responsibilities not within the scope of the Services provided by the performing Party ("Additional Services"). Any such Additional Services will be provided under supplements to Schedule A-1 or Schedule A-2, as applicable, entered into by the Parties ("Supplements") for the charges set forth therein and mutually agreed upon. Supplements shall be in the form of Schedule E to the Agreement. During the ninety (90) day period after the Effective Date, the Parties will work together to define a process reasonably satisfactory to the Parties that will be used to (a) submit and prioritize requests for Additional Services and (b) create and execute Supplements for Additional

Capitalized terms used in the summary and not otherwise defined shall have the meaning ascribed to them in the Agreement. The contents of this summary are qualified in their entirety by the terms of the Agreement. In the event of a conflict between this summary and the Agreement, the Agreement shall govern.

As noted above, the Debtors are seeking Court approval of the Agreement *nunc pro tunc* to the Petition Date. Upon entry of a final order, as requested by the Debtors, the Petition Date will operate as the Effective Date of the Agreement.

Services.

Changes to
Services/
Change Control
Procedures

A Party may, from time to time during the Term, upon at least ninety (90) days prior written notice to the other Party, request that the other Party provide Customized Services. If the Parties mutually agree on the provision of Customized Services, such request for Customized Services and such Customized Services will be provided under Supplements to Schedule A-1 or Schedule A-2, as applicable. Before Supplier is required to provide any Customized Services, the Parties shall jointly agree on the applicable Charges for any agreed Customized Services, including any Charges that may be required to equitably compensate Supplier for any additional costs it may reasonably incur in connection with any changes to the Services.

# Ownership of Data

As between Recipient and Supplier, Recipient owns and will continue to own all right, title and interest in and to all Recipient Data. To the extent that Recipient Data is embedded or incorporated into reports and other documentation, analyses, compilations and other materials (including code or software) owned or licensed by Supplier ("Supplier Materials") and provided to and for use by Recipient as part of the Services, Supplier will not be deemed to have assigned or transferred any of its right, title or interest in or to any underlying Intellectual Property Rights thereto. Supplier grants to Recipient a perpetual, irrevocable, royalty free, transferable (to a Buyer of all or substantially all of the core business of Recipient) license to use, reproduce, display and perform (whether publicly or otherwise) and modify (and have others exercise such rights on behalf of Recipient (or Buyer)) such Supplier Materials solely as necessary for Recipient's (or Buyer's) use in the ordinary course of its mortgage related business. Recipient's (or Buyer's) use (including without limitation, the use by any third party on behalf of Recipient (or Buyer)) of the Supplier Materials is subject to the confidentiality obligations set forth in Section 9 of the Agreement and any third party restrictions imposed on any Supplier Materials of which Supplier makes Recipient aware. Additionally, Recipient's ownership of the Recipient Data reflected in Supplier Materials shall not serve to transfer or otherwise affect any of Supplier's right, title and/or interest in and/or to any of underlying Intellectual Property Rights in any Supplier Materials. Supplier may not use Recipient Data for any purpose except: (x) to provide the Services; (y) as required by AFI as the parent of ResCap to meet its financial and regulatory reporting requirements; or (z) as otherwise permitted under the Agreement, nor may Supplier sell, assign, lease or otherwise dispose of or commercially exploit Recipient Data. Supplier acknowledges that Recipient is not restricted from using or disseminating Recipient Data in the format in which it is provided by Supplier or any different format.

#### Compensation

ResCap will pay to AFI the Charges as set forth in <u>Schedule C-1a</u> and elsewhere in <u>Schedule C</u>, and AFI will pay to ResCap the Charges as set forth in <u>Schedule C-2a</u> and elsewhere in <u>Schedule C</u>. Supplier will provide Recipient with estimated invoices (with information linking the Services delivered to the invoice

amounts) on a monthly basis on or before the last day of each calendar month for all Services performed by Supplier and all related Charges incurred by Recipient during that month. The last business day of the calendar month is the cut-off for delivered Services and related Charges to be invoiced in the next calendar month. The estimated charges will be adjusted to actual charged by the last day of the following month. Recipient will pay the estimated and adjusted actual invoices within forty-five (45) days of the receipt of an invoice from Supplier. Any payment by Recipient is without prejudice of its right to contest the accuracy of any Charges.

# Recipient Indemnification

Recipient will indemnify, hold harmless and defend Supplier and its Affiliates and their respective directors, officers and employees ("Indemnified Parties"), from and against any and all liabilities, damages, penalties, judgments, assessments, losses, costs and expenses in any case, whether arising under strict liability or otherwise (including reasonable outside attorneys' fees) (collectively, "Damages") suffered or otherwise incurred due to a Third Party Claim arising from or out of or relating to the Agreement, including the performance (or failure to perform) by Supplier of its obligations under the Agreement; provided, however, that to the extent and in the proportion Damages also arise out of or relate to the gross negligence or willful misconduct of any Indemnified Party, then such indemnity will not apply.

# Supplier Indemnification

Supplier will indemnify, hold harmless and defend the Recipient Indemnified Parties from and against any and all Damages suffered or otherwise incurred due to a Third Party Claim to the extent arising from or out of or relating to the gross negligence or willful misconduct of any Supplier Parties; <u>provided</u>, <u>however</u>, that to the extent and in the proportion Damages also arise from or out of or relate to the performance (or failure to perform) by Recipient of its obligations under the Agreement, then such Supplier indemnity will not apply.

# Other Indemnification Provisions

The Agreement also contains indemnification provisions relating to the infringement or misappropriation of intellectual property rights and third party software, as well as procedures for seeking indemnification and certain limitations on liability, as set forth in more detail in the Agreement.

#### Termination

Without limiting the rights of the Parties under any other provision of the Agreement, any Service to be provided under the Agreement may be terminated as follows:

- (a) by either Party acting in its capacity as "Supplier", upon written notice to the other Party if, after Recipient fails to pay the Charges for such Service when due in accordance with the Agreement, the terminating Party sends to the other Party an initial notice of such failure and the other Party fails to pay such Charges within forty-five (45) days of receipt of such initial notice; or
- (b) by either Party acting in its capacity as "Supplier", upon written

notice to the other Party if, following a material breach by Recipient of the Agreement other than a payment default, the terminating Party sends to the other Party a notice of such material breach and the other Party fails to cure such material breach within thirty (30) days; <u>provided however</u>, that if cure cannot reasonably be accomplished within such thirty (30) day period, the Agreement may not be terminated by reason of such breach for so long as the other Party commences a cure within such thirty (30) day period and pursues such cure diligently to completion and such completion occurs within ninety (90) days of such written notice; or

- (c) by either Party acting in its capacity as "Recipient", upon written notice to the other Party if, following a material breach by Supplier of the Agreement, the terminating Party sends to the other Party an initial notice of such material breach and the other Party fails to cure such material breach within thirty (30) days of receipt of such initial notice; provided, however, that if cure cannot reasonably be accomplished within such thirty (30) day period, the Agreement may not be terminated by reason of such breach for so long as the other Party commences a cure within such thirty (30) day period and pursues such cure diligently to completion and such completion occurs within ninety (90) days of such written notice; or
- (d) by either Party acting in its capacity as "Recipient", upon at least ninety (90) days prior written notice to the other Party; or
- (e) otherwise upon mutual agreement of the Parties.

## Termination Following Sale

Upon the final approval by the Bankruptcy Court of a Sale, either Party in its capacity as a Supplier may within fourteen (14) days of such final approval elect to terminate the provision of the Services that such Party is required to provide to the other Party under the Agreement, effective upon the closing of such Sale. In the event of such termination, Recipient will provide Supplier with notice of any Termination Assistance Services required by Recipient within twenty-one (21) days after receipt of such notice.

Rights upon
Termination or
Expiration/
Termination
Assistance
Services

Commencing six (6) months prior to expiration of the Term or any extension thereof, or upon either Party receiving any notice of termination from the other Party pursuant to <u>Section 14</u> of the Agreement:

(a) At Recipient's option, Supplier will provide to Recipient the termination assistance reasonably requested by Recipient to allow the Services to continue, and to facilitate the orderly transfer of responsibility for performance of the Services, including any migration of Recipient Data, to Recipient (collectively, "Termination Assistance Services"), for a period of up to six (6) months (or, with respect to information technology Services, eighteen (18) months) after the effective date of such termination of Services or expiration of the Term (the "Termination Services Periods");

<u>provided</u> <u>however</u>, that, if mutually agreed upon in writing by the Parties (including with respect to adjustments in Charges), the Termination Service Periods may be extended until the effective date of a plan of reorganization of ResCap and its affiliated Debtors.

(b) The Termination Assistance Services to be provided during the Termination Services Periods may include, as and if reasonably required by Recipient, the following, among other Services: (i) developing, together with Recipient, a plan for the orderly transition of the performance of the Services, including the migration of Recipient Data from Supplier to Recipient or to a third party designated by Recipient; (ii) providing reasonable training for personnel of Recipient in the performance of the Services then being transitioned to Recipient or the third party designated by Recipient, to the extent Services are being assumed by that third party; and (iii) providing cooperation to Recipient to facilitate the transition of the performance of the Services to Recipient or to a third party designated by Recipient, in accordance with a mutually agreed upon transition plan for the applicable Services.

ResCap will pay to AFI Charges for the Termination Assistance Services as set forth in <u>Schedule C</u> to the Agreement, including during the Termination Services Periods. AFI will pay to ResCap Charges for the Termination Assistance Services as set forth in <u>Schedule C</u> to the Agreement, including during the Termination Services Periods. In the event of any termination of Services by Supplier for cause, if approved by the Bankruptcy Court, Charges for the related Termination Assistance Services will be paid by Recipient monthly in advance.

As set forth in <u>Schedule C</u> to the Agreement, the price of Termination Assistance Services will be documented in a Supplement to <u>Schedule A</u>. In addition to any Charges otherwise provided for in the Agreement or <u>Schedule C</u>, Recipient will reimburse Supplier for all incremental additional resource and other costs and expenses required or incurred by Supplier to provide Termination Assistance Services.

Transition
Services
Agreement

Upon the approval by the Bankruptcy Court of any Sale, or as otherwise mutually agreed between the Parties to facilitate a Sale, AFI and ResCap each agree to promptly enter into good faith negotiations with the Buyer(s) in such Sale for an agreement pursuant to which AFI, ResCap and/or the Buyer(s) would provide or receive, as applicable, reasonable specified transition services in connection with such Sale (the "Transition Services Agreement"). The services to be provided to the Buyer and the terms relating to the transition of certain Services are to be addressed in the Transition Services Agreement. For the avoidance of doubt, while AFI agrees to negotiate in good faith with ResCap and/or a Buyer with respect to entering into a Transition Services Agreement, AFI is not obligated to provide services to a Buyer or after a Sale, other than Termination Assistance Services as contemplated in Section 14.5(a)(2) of the Agreement unless otherwise agreed by AFI.

### Total Shared Services Charge

Total Shared Services Charges (as described below) will begin on the Effective Date and are set forth in <u>Schedule C-1a</u> and <u>Schedule C-2a</u> to the Agreement. Charges for Additional Services, Customized Services, and Termination Assistance Services shall begin on the date such services begin to be performed by Supplier.

"Total Monthly Shared Services Charge" means, in respect of any individual Service, the monthly Charge for that Service, which shall be applicable from the Effective Date until the termination date for such Service, as such Charges are listed on Schedule C-1a or Schedule C-2a to the Agreement and may be adjusted in accordance with any Price Adjustment Event and/or Price Adjustment Process (each as described below). The Total Shared Services Charge represents the aggregate of the Total Base Costs, Third Party Costs and IT Projects as referenced on Schedule C-1a and Schedule C-2a. The Total Base Costs are equal to an amount representing Employee Costs, IT Costs, Platform Costs and the Indirect Support Costs for that Functional Service Area, as described below:

- (i) Employee Costs means compensation, benefits, travel and other non-compensation related personnel costs;
- (ii) IT Costs means those costs associated with IT services under <u>Schedule A-1</u> and <u>Schedule A-2</u> to the Agreement;
- (iii) Platform Costs means the Charges to the Recipient for use of the IT services on Supplier's platform related to depreciation and amortization in respect of IT infrastructure; and
- (iv) Indirect Support Costs means, for AFI, an amount equal to the percentage of the total of Employee Costs, IT Costs and Platform Costs set forth in Schedule C-1b, and for ResCap means an amount equal to the percentage of the total of Employee Costs, IT Costs and Platform Costs set forth in Schedule C-2b. Such percentages in each case are those referenced in the formulas in the Pricing Methodology Spreadsheets titled "Shared Services Global Functions Exhibit v.11.pdf" (AFI to ResCap) and "ResCap to AFI Shared Services Pricing 05-13-12.pdf" (ResCap to AFI).

These amounts are set forth in <u>Schedule C-1a</u> and <u>Schedule C-2a</u> to the Agreement on a monthly basis as the Total Monthly Shared Services Charge. The Total Monthly Shared Services Charges do not include any (i) Charges for Additional Services or Customized Services, (ii) Termination Assistance Services or (iii) Pass-Through Expenses.

Pricing for the Services set forth on Schedule C-1a and Schedule C-2a to the

Agreement can change on a monthly basis, depending upon whether the Service is priced on the basis of Monthly Fixed Charges or Monthly Variable Charges, as indicated on <u>Schedule C-1b</u> and <u>Schedule C-2b</u> to the Agreement. Any such calculations will be made by reference to and in accordance with the formulas in the Pricing Methodology Spreadsheets titled "ResCap to AFI Shared Services Pricing 05-13-12.pdf" (ResCap to AFI).

Price
Adjustment
Event

As set forth on <u>Schedule C</u> to the Agreement, "Price Adjustment Event" means any of the following: (i) a Business Change Event<sup>6</sup> occurs; or (ii) a Service has been terminated pursuant to <u>Section 3</u> (Additional Services or Customized Services) or Section 14 (Termination) of the Agreement.

Price
Adjustment
Process

The Price Adjustment Process described in <u>Schedule C</u> to the Agreement will be followed by the Parties upon either Party's written request in the event of a Price Adjustment Event. The Price Adjustment Process for Business Change Events is subject to any approval right or termination right that Supplier may have with respect to such change in Services pursuant to <u>Section 3</u> (Additional Services or Customized Services) or <u>Section 14</u> (Termination) of the Agreement.

- (a) The Price Adjustment Process will be initiated immediately following a Price Adjustment Event and the resulting change to the Charges will take effect as soon as practical and will be retroactive to the point in time that the Price Adjustment Event occurs (or, in the case of adjustments under subsection (c) below, retroactive to the point in time that the New Cost Allocation (as described below) is determined).
- (b) Unless the Price Adjustment Process is initiated for a Business Change Event, during the Price Adjustment Process the Parties will mutually determine in good faith the cost impact of the Price Adjustment Event through the methodologies set forth in the exhibits to Schedule C including all of the sub-attachments. When a Service is being terminated by a Party, the starting assumption is that Recipient will no longer be charged for that Service; provided, however, that Supplier may be entitled to reimbursement from Recipient of all costs previously paid by Recipient for such Service that Supplier is not able to eliminate, for example, costs for physical assets which cannot be reduced (i.e. leasing expenses or rental expenses which cannot be eliminated), third party costs (i.e. minimum revenue commitments which are required under a third party agreement) or any other non-internal costs ("Stranded Costs") which the Parties using appropriate due diligence and commercially reasonable efforts can not eliminate. If, after using such efforts the Parties are unable

<sup>&</sup>quot;Business Change Event" means any change in the products or services offered by Recipient or the expansion or closure of any Facility of any Recipient, or a decrease or increase in Recipient's level of consumption of the Services from the historical usage levels of its business, in either case that has an impact on the cost to Supplier of providing any impacted Services.

to eliminate 100% of any Stranded Costs, the Parties will meet and discuss the remaining Stranded Costs to be reimbursed by Recipient to Supplier.

- (c) The Total Shared Services Charges set forth in Schedule C-1a and Schedule C-2a were determined by allocating to Recipient a portion of Supplier's underlying costs of performing the applicable Service (such cost allocation, the "Initial Cost Allocations"). In the event the Price Adjustment Process is initiated solely due to a Business Change Event and does not involve a termination of the applicable Service, then during the Price Adjustment Process, Supplier will (i) use the cost allocation methodology that was used to determine the Initial Cost Allocation for such Service, to determine a new cost allocation reflecting Recipient's then-current level of consumption of such Service, taking into account material increases or decreases (if any) in Supplier's overall costs that are a direct result of the change in Recipient's level of consumption of such Service (a "New Cost Allocation"), and (ii) if the New Cost Allocation differs from the Initial Cost Allocation for such Service, adjust the Total Shared Services Charge for such Service accordingly. At Recipient's request, Supplier will provide Recipient with all supporting calculations of the effects of such Business Change Event on Supplier's costs of performing the Services.
- 14. As noted above, the services to be provided by AFI and ResCap pursuant to the Agreement are broken down into categories of services, as set forth on Schedules A-1 and A-2 to the Agreement (each a "Functional Service Area"), the most significant of which are summarized as follows:
  - (a) Information Technology. AFI and ResCap provide information technology development, maintenance and servicing functions to each other for each of the systems used in the operation of their respective businesses. Likewise, AFI and ResCap provide information technology services at their respective facilities for employees of the other affiliates located at such facilities. Many of the information technology services provided by AFI involve creation and maintenance of systems that are critical to Debtors' daily business operations and the integration of the Debtors' and Ally Bank's systems to ensure seamless operation of the Debtors' mortgage origination, servicing and securitization operations.
  - (b) Human Resources. ResCap and AFI also provide various global human resources functions to each other. For instance, ResCap provides various employee relations, staffing, workforce planning, and compensation design, administration and monitoring functions to AFI. AFI, on the other hand, provides

ResCap services relating to compensation plan monitoring, benefits design and administration, talent management and performance review functions.

- (c) Finance, Tax & Audit. AFI and ResCap provide various finance, tax and audit services on a global basis for the consolidated organization. More specifically, ResCap provides Ally Bank with many accounting services relating to general ledger recordation and reconciliation, mortgage loan-level payment accounting, and real estate owned accounting. The Debtors also rely upon AFI for accounts payable and payroll processing, tax services and internal audit procedures and frameworks. In addition, AFI provides ResCap with various regulatory reporting and control services. These services are critical to the Debtors' ability to continue to operate their businesses in the ordinary course and ensure compliance with the requirements of governmental regulators.
- (d) Compliance. AFI provides the Debtors with global compliance services, including development of compliance and procedure frameworks, training employees regarding the compliance policies and procedures, and assessing and testing compliance programs. The Debtors also provide to AFI various compliance support services. These services enable the Debtors and AFI to comply with recent agreements and settlements with federal and state authorities and ensure that the Debtors and AFI continue to operate in compliance with the requirements of governmental units that regulate the lending industry.
- (e) Risk Management. ResCap provides AFI with mortgage credit and servicing policy, risk management, and quality assurance services, as well as certain services related to the repurchase of loans from correspondent lenders and brokers on behalf of Ally Bank. AFI provides to the Debtors enterprise policies and governance, operational risk management, business continuity planning, market and counterparty risk management, and reports and analytics.
- (f) Treasury. AFI provides the Debtors with various cash management and treasury services, <sup>7</sup> including cash and wire desk services, as well as liquidity and collateral management services. Each of these services is critical to the Debtors' ability to maintain liquidity and funding so that they can continue to operate their businesses and continue activities related to the origination, servicing and securitization of mortgage loans.
- (g) Legal. ResCap provides various banking and mortgage related legal services to AFI, while AFI provides general legal services to the Debtors. The legal services that are provided are in the form of certain AFI legal employees allocating a percentage of their time to ResCap legal matters, and vice versa for ResCap's legal employees working on legal matters relating to AFI.

By separate motion, filed concurrently herewith, the Debtors seek to continue their current cash management systems with modifications necessitated by their postpetition financing needs.

- (h) Supply Chain. AFI's global supply chain functional group provides ResCap with assistance in negotiating contracts with and transitioning to suppliers, as well as monitoring and off-boarding existing suppliers.
- (i) Capital Markets. The Debtors provide various capital markets related functions to Ally Bank, including services related to securitization of mortgages with the GSEs, portfolio management and accounting, and management of mortgage servicing rights. AFI provides various investment management and capital markets IT services to the Debtors.
- (j) <u>Marketing</u>. AFI provides the Debtors with global brand and product marketing services across mass media, direct, digital, sponsorship/events, and sales support materials. Such services include strategic marketing planning and management of day-to-day marketing operational responsibilities.
- (k) Facilities. AFI provides the Debtors with various facilities related services, mostly through contracts with third parties, including, without limitation, monitoring and contracting for cost of space, mail and food services; providing space planning; and coordinating any construction or renovations.
- 15. The specific services to be performed in each Functional Service Area are set forth in various Statements of Work. The specific services detailed in the Statements of Work are documented in a summary of the Statements of Work attached hereto as <a href="Exhibit B">Exhibit B</a>.
- 16. As discussed above, the Agreement also provides for the provision of certain Customized Services, Additional Services and Termination Assistance Services. For the avoidance of doubt, by seeking the authority to enter into and perform under the Agreement, the Debtors are seeking the authority to provide to, receive from and/or compensate AFI for such further services, as and if necessary in the ordinary course of business, without the need for further order of this Court.
- 17. The prices of the services being provided to AFI and ResCap pursuant to the Agreement, which are set forth in more detail on Schedules C-1a and C-2a to the Agreement and described above, are generally in the form of monthly service charges by Functional Service Area for services provided by ResCap to AFI and services provided by AFI to ResCap. The

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 17 of 126

anticipated aggregate monthly cost to ResCap for the services received from AFI is approximately \$10.2 million. The anticipated aggregated monthly cost for services provided by ResCap to AFI is approximately \$4.4 million. The initial charges under the Agreement are based on projected operations at postpetition levels, which are expected to continue through one or more sales of the Debtors' assets. The Debtors are not seeking to pay any prepetition claims through or pursuant to the Agreement.

- 18. The quantity and scope of services to be received by the Debtors pursuant to the Agreement has been scaled back from historical levels to reflect only those services that the Debtors believe are necessary to ensure the efficient operation of their businesses during their Chapter 11 cases. As noted above, either party in its capacity as a recipient may terminate the receipt of services, with pricing to be adjusted based on such termination, at any time pursuant to the terms of the Agreement, including in connection with the closing of a sale of substantially all of the Debtors' assets. Any such termination as a recipient does not relieve the party from its obligations as a supplier of services. The Debtors anticipate that the services they will need under the Agreement will be reduced significantly following a sale; however, the Debtors will likely require the continuation of certain limited services from AFI during a wind-down period. Pursuant to the Agreement, the Debtors will be able to secure the performance of such limited services, at reduced costs, to complete the wind-down of their businesses following a sale.
- 19. The Agreement was negotiated at arm's length over an extended period of time by senior personnel from AFI and ResCap with the assistance of independent counsel to each of the Parties, which is reflected in the flexibility of the services provided and the pricing and termination rights. On May 13, 2012, ResCap's board of directors approved ResCap's entry into the Agreement.

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 18 of 126

### B. The Urgent Need for the Agreement

- 20. As described above, ResCap and AFI provide services to each other that are critical to the continued operations of their affiliated businesses. If the Debtors are not authorized to continue receiving and compensating AFI for all of the services as well as providing services to AFI pursuant to the Agreement, there will be immediate and widespread disruption of the Debtors' operations which would, in all likelihood, significantly diminish the value of the Debtors' estates and impede the Debtors' ability to consummate any going concern sale with a buyer.
- 21. For example, failure to receive these services would disrupt the Debtors' use of information technology. The Debtors rely upon AFI to maintain, develop and service many of the integrated information technology systems utilized, among other things, to seamlessly transmit loan applications and data among various AFI affiliates. Any disturbance to the operation of these systems could cause an interruption in the Debtors' origination and servicing businesses and result in a significant decline in the value of the Debtors' businesses.
- 22. Moreover, discontinuing the services would disrupt fundamental administrative aspects of the Debtors' businesses. For example, many of the Debtors' human resources services are managed by AFI. Without AFI's human resource services, the Debtors would not be able to offer basic health benefits to employees, continue certain other employee benefit plans and administer payroll, potentially causing an exodus of the Debtors' most valuable employees and irreparable harm to employee morale and trust. In an industry where enterprise value is so heavily contingent on employees, such repercussions would be devastating to the Debtors' operations.
- 23. Likewise, without AFI services, the Debtors cannot continue to ensure compliance with the myriad of regulatory requirements governing the mortgage origination and

servicing industry. Ultimately, failing to comply with regulatory requirements could potentially result in revocation of the licenses that each state requires the Debtors to maintain in order to continue origination activities and mortgage loan servicing. At a minimum, failure to satisfy compliance requirements would cause the Debtors to incur substantial costs and administrative burdens to address any such non-compliance.

AFI to preserve the value of their bankruptcy estates. The Debtors' and AFI's loan origination and securitization businesses are integrated. In connection with origination activities, across a 47 state platform, the Debtors, in their capacities as licensed mortgage brokers, broker loan applications and supporting materials with Ally Bank. Ally Bank in turn, then underwrites, originates and funds loans based on the loan application packages. The substantial majority of these loans are pooled and securitized as part of GSE backed securities for which the Debtors typically act as loan servicers. Thus, it is essential to the Debtors' day-to-day operations and their business model that AFI and its affiliates are able to continue to rely upon and have the benefits of the services historically performed by ResCap. Any material disruption to AFI's operations caused by ResCap's failure to provide services would necessarily disrupt the Debtors' operations to the detriment of the Debtors and their creditors.

#### APPLICABLE AUTHORITY

- A. The Agreement Has a Sound Business Purpose and Should Be Approved Pursuant to Sections 363(b) and 105 (a) of the Bankruptcy Code
- 25. Bankruptcy Code section 363(b)(1) provides that after notice and a hearing a debtor may use, sell or lease property of the estate outside the ordinary course of business. 11 U.S.C. § 363(b)(1). Additionally, under Bankruptcy Code section 105(a), this Court "may issue any order...that is necessary or appropriate to carry out the provisions of [the

Bankruptcy Code]." 11 U.S.C. § 105(a). Courts in this Circuit have consistently held that a debtor may use, sell or lease property of the estate outside the ordinary course of business where a sound business purpose justifies such actions. Contrarian Funds LLC v. Aretex LLC (In re WestPoint Stevens, Inc.), 600 F.3d 231, 248 n.8 (2d Cir. 2010); Comm. of Equity Sec. Holders v. Lionel Corp. (In re Lionel Corp.); 722 F.2d 1063, 1070 (2d Cir. 1983); In re Borders Group, Inc., 453 B.R. 459, 473 (Bankr. S.D.N.Y. 2011); In re Boston Generating, LLC, 440 B.R. 302, 321 (Bankr. S.D.N.Y. 2010). Specifically, once a debtor articulates a valid business justification for a particular form of relief, the court reviews the debtor's request under the "business judgment rule." "The business judgment rule 'is a presumption that in making a business decision the directors of a corporation acted on an informed basis, in good faith and in the honest belief that the action taken was in the best interests of the company." Official Comm. of Subordinated Bondholders v. Integrated Res., Inc. (In re Integrated Res., Inc.), 147 B.R. 650, 656 (S.D.N.Y. 1992) (quoting Smith v. Van Gorkom, 488 A.2d 858, 872 (Del. 1985), rev'd on other grounds, Gantler v. Stephens, 965 A.2d 695, 713 (Del. 2009)); In re Quigley Co., 437 B.R. 102, 156-57 (Bankr. S.D.N.Y. 2010). "Courts are loath to interfere with corporate decisions absent a showing of bad faith, self-interest, or gross negligence." In re Quigley Co., 437 B.R. at 157 (quoting In re Integrated Res., Inc., 147 B.R. at 656). Consequently, a debtor's business decision "should be approved by the court unless it is shown to be so manifestly unreasonable that it could not be based upon sound business judgment, but only on bad faith, or whim or caprice." In re Aerovox, Inc., 269 B.R. 74, 80 (Bankr. D. Mass. 2001) (internal quotations omitted); cf. In re Global Grossing Ltd., 295 B.R. 726, 742 (Bankr. S.D.N.Y. 2003). The burden of rebutting the presumption in favor of a debtor exercising its business judgment under section 363(b) falls on a party opposing the requested relief. See In re Integrated Res., Inc., 147 B.R. at 656 (citing

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 21 of 126

<u>Aronson v. Lewis</u>, 473 A.2d 805, 812 (Del. 1984), <u>rev'd on other grounds</u>, <u>Brehm v. Eisner</u>, 746 A.2d 244, 254 (Del. 2000)).

- 26. The decision to document and enter into the Agreement is in the sound business judgment of the Debtors because it ensures that the Debtors are able to continue to receive necessary services that will permit them to operate their businesses smoothly in Chapter 11 through a sale and a subsequent wind-down. As discussed above, the Agreement was negotiated at arm's length and the amount of services to be received by the Debtors pursuant to the Agreement has been scaled back from historical levels to reflect only those services that the Debtors believe are necessary during their Chapter 11 cases.
- 27. Likewise, the Debtors submit that the pricing terms in the Agreement were heavily negotiated and reasonably calculated. The Agreement was also drafted to provide step-down pricing so that following the sale of the Debtors' businesses, ResCap will pay for only those services that the Debtors require to wind-down their estates, which will permit the Debtors to reduce costs following the closing of a sale.
- 28. The cost of the services to be provided to ResCap and its affiliates under the Agreement is eminently reasonable, particularly when juxtaposed against the additional costs and administrative burden the Debtors would no doubt incur if they attempted to transfer these services to new third-party suppliers or perform them in house, assuming they were able to do so in either case. Given the type and volume of services received and provided by the Debtors, it would be highly impracticable for the Debtors to duplicate these services through other sources. Even if they could do so, shifting to outside service providers would likely require significant expenditures to customize the suppliers' services to address the Debtors' particularized needs. Further, the Debtors' employees are already experiencing disruption in their day-to-day

operations. Sourcing these services to new third-party providers would only amplify the disruption and add additional layers of unnecessary complexity to the ability of employees to complete their daily tasks.

29. In light of the foregoing, the Debtors submit that they have demonstrated more than a sound business justification to continue to provide and receive the services under the Agreement. Allowing the Debtors to enter into and provide services under the Agreement without interruption is absolutely necessary to preserve, enhance and maximize the value of the Debtors' estates and is in the best interests of the Debtors, their creditors, and all other parties in interest. Accordingly, the Debtors submit that entry into the Agreement represents an exercise of their sound business judgment and request authorization, pursuant to Bankruptcy Code section 363(b), to enter into the Agreement and to continue to provide the services set forth in the Agreement.

### B. Request for Immediate Relief and Waiver of Stay

- 30. Bankruptcy Rule 6003 generally precludes the Court from authorizing certain relief until twenty-one days after the petition is filed, except to the extent necessary to prevent "immediate and irreparable harm." Fed. R. Bankr. P. 6003. The Debtors submit that Bankruptcy Rule 6003 has been satisfied because the concerns raised above demonstrate that the interim relief requested in this Motion is necessary to avoid immediate and irreparable harm to the Debtors and their estates. Accordingly, the Debtors request that an order granting the relief requested in this Motion be entered on an interim basis.
- 31. To successfully implement the foregoing, the Debtors seek a waiver of the notice requirements under Bankruptcy Rule 6004(a) and the fourteen-day stay under Bankruptcy Rule 6004(h).

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 23 of 126

#### SCOPE OF MOTION

32. This Motion is limited to the Agreement between ResCap and AFI and the associated services provided by ResCap and AFI thereunder. The Motion does not cover all aspects of the Debtors' transactions with AFI or other affiliates. Other ordinary course agreements and payment arrangements exist between and among the Debtors and their non-debtor affiliates, including, without limitation, those that are disclosed in publicly available filings with the Securities and Exchange Commission and other regulatory bodies. Whether or not a particular transaction with an affiliate is described in the Agreement or this Motion, and whether or not the Debtors seek an order with respect to the continuation of such transaction, the Debtors intend to continue ordinary course transactions with affiliates after the Petition Date pursuant to the authority granted to them by Bankruptcy Code section 363(c). The Debtors do not believe that any of the relief sought herein limits their right to conduct other transactions with Debtor and non-debtor affiliates in the ordinary course of business without prior Court approval.

#### **NOTICE**

33. Notice of this Motion will be given to the following parties, or in lieu thereof, to their counsel: (a) the Office of the United States Trustee for the Southern District of New York; (b) the office of the United States Attorney General; (c) the office of the New York Attorney General; (d) the office of the United States Attorney for the Southern District of New York; (e) the Internal Revenue Service; (f) the Securities and Exchange Commission; (g) each of

One intercompany item that is not covered by this Motion, or the pricing in the Agreement, is the charge for AFI's and Ally Bank's use of buildings or spaces that are owned or leased by the Debtors, as well as charges for the Debtors' use of space owned or leased by AFI or Ally Bank. In 2012, the Debtors expect to pay, in the ordinary course of business, up to \$3 million to AFI for the use of AFI and/or Ally Bank space. ResCap, on the other hand, expects to charge AFI up to \$12 million in 2012 for AFI's and Ally Bank's use of the Debtors' space.

the Debtors' prepetition lenders, or their agents, if applicable; (h) each of the indenture trustees for the Debtors' outstanding notes issuances; (i) Ally Financial Inc. and its counsel; (j) counsel to the administrative agent for the Debtors' proposed providers of debtor in possession financing; (k) Nationstar Mortgage LLC and its counsel; and (l) the parties included on the Debtors' list of fifty (50) largest unsecured creditors (collectively, the "Initial Notice Parties"). The Debtors submit that, in view of the facts and circumstances, such notice is sufficient and no other or further notice need be provided.

- 34. Within two (2) days after entry of an interim order, the Debtors propose to serve a copy of the Motion and the interim order upon the Initial Notice Parties. The Debtors request that the Court schedule the final hearing on the Motion for a date that is as soon as practicable, but in no event later than forty-five (45) days following the entry of the interim order, and establish the date prior to the final hearing for parties to file objections to the Motion.
- 35. Any objections to the relief requested in the Motion must be filed with the Clerk of the Bankruptcy Court and served upon and received by: (a) proposed counsel for the Debtors, Morrison & Foerster LLP (Attn: Larren M. Nashelsky, Gary S. Lee and Lorenzo Marinuzzi); (b) the Office of the United States Trustee for the Southern District of New York, 33 Whitehall Street, 21<sup>st</sup> Floor, New York, NY 10004 (Attn: Tracy Hope Davis, Linda A. Riffkin and Brian S. Masumoto); (c) counsel for AFI, Kirkland and Ellis, LLP, Citigroup Center, 601 Lexington Avenue, New York, NY 10022 (Attn: Richard Cieri); (d) counsel for the administrative agent for the Debtors' proposed providers of debtor in possession financing; and (e) counsel for any statutory committee appointed in the Debtors' cases. If no objections are filed to the Motion, the Court may enter the order without further notice or hearing.

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 25 of 126

#### CONCLUSION

36. WHEREFORE, the Debtors respectfully request that the Court: (i) enter an interim order substantially in the form attached hereto as <u>Exhibit C</u>, granting certain of the relief sought herein immediately; (ii) enter a final order substantially in the form attached hereto as <u>Exhibit D</u>, granting the relief sought herein on a final basis and authorizing ResCap to enter into the Agreement *nunc pro tunc* to the Petition Date; and (iii) grant such other and further relief to the Debtors as the Court may deem just and proper.

Dated: May 14, 2012

New York, New York

/s/ Larren M. Nashelsky

Larren M. Nashelsky Gary S. Lee Lorenzo Marinuzzi MORRISON & FOERSTER LLP 1290 Avenue of the Americas New York, New York 10104 Telephone: (212) 468-8000

Facsimile: (212) 468-7900

Proposed Counsel for the Debtors and Debtors in Possession

## EXHIBIT A

12-12020-mg	Doc 41	Filed 05/14/12	Entered 05/14/12 14:50:46	Main Document
Pg 27 of 126				

**EXECUTION VERSION** 

**Shared Services Agreement** 

by and between

Ally Financial Inc. and Residential Capital, LLC

Effective as of May \_\_\_\_, 2012

<i>1</i> .	<b>DEF</b>	INITIONS	1
	1.1	Definitions.	
<i>2</i> .	TER	M AND RENEWAL	
	2.1	Initial Term.	
	2.2	Renewal.	
<i>3</i> .	SER	VICES	5
	3.1	Services	5
	3.2	Additional Services.	6
	3.3	Changes to Services/Change Control Procedures.	
	3.4	Recipients' Obligations.	
	3.5	Required Consents.	
,	3.6	Security Level; Additional Security Measures.	
<i>4</i> .	USE	OF AFFILIATES AND SUBCONTRACTORS	8
	4.1	Affiliates and Subcontractors	8
	4.2	Compliance.	
<i>5</i> .	REL	ATIONSHIP MANAGEMENT	9
	5.1	Relationship Managers.	9
	5.2	Governance Model.	
	5.3	Reports.	9
	5.4	Regulatory Review.	10
	5.5	Books and Records; Audit.	
	5.6	Books and Records; Audit.	
	5.7	Informal Dispute Resolution Procedures.	
	5.8	Continued Performance.	
<i>6</i> .	FACI	ILITIES	11
	6.1	Use of Recipient Facilities.	
	6.2	Supplier Facilities and Systems.	
_	6.3	Physical Security for Facilities.	
<i>7</i> .	INTE	ELLECTUAL PROPERTY AND PROPRIETARY RIGHTS	12
	<b>7.1</b>	Ownership of Pre-Existing Intellectual Property.	12
	7.2	Development of Intellectual Property.	
	7.3	Limited License to Use Supplier Work Processes and Software	
_	7.4	No Implied Licenses.	
8.	REC	IPIENT DATA	13
	8.1	Definition	13
	8.2	Ownership.	13
	8.3	Data Security.	
<i>9</i> .	CON	FIDENTIALITY	14
	9.1	Obligations.	14
	9.2	Excluded Information	
	9.3	Compelled Disclosure.	15
	9.4	Return of Information.	
	9.5	Exception.	15

# 12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 29 of 126

	9.6	Obligations	15
	9.7	Loss of Confidential Information.	16
	9.8	No Implied Rights.	16
<i>10</i> .	COM	PENSATION	16
	10.1	General.	16
	10.2	Taxes.	16
	10.3	Invoicing and Payment.	16
<i>11</i> .	REPI	RESENTATIONS AND WARRANTIES	
	11.1	Services.	17
	11.2	Maintenance.	
	11.3	Authorization.	
	11.4	Viruses.	
	11.5	Disclaimer.	
<i>12</i> .	INSU	TRANCE	17
	12.1	Insurance.	17
	12.2	Risk of Loss.	
			10
<i>13</i> .	INDE	EMNIFICATION AND LIMITATIONS ON LIABILITY	18
	13.1	Indemnification.	18
	13.2	Limitations on Liability.	
	13.3	Indemnification and Limitations on Liability Relating to Negligence and	
		Strict Liability.	22
	13.4	Exclusive Remedy.	
	13.5	Insurance.	23
<i>14</i> .	TERA	MINATION	23
	14.1	Termination.	23
	14.2	Termination Following Sale.	
	14.3	Election of Terminating Party.	24
	14.4	Survival.	
	14.5	Rights Upon Termination or Expiration; Sale.	
	14.6	Dedicated Assets.	
	14.7	Transition Services Agreement.	
<i>15</i> .	FULI	LY INTEGRATED AGREEMENT	25
<i>16</i> .	<b>GEN</b>	ERAL	26
	16.1	Acknowledgement.	26
	16.2	Binding Effect; No Assignment.	26
	16.3	Counterparts.	27
	16.4	Entire Agreement.	
	16.5	Force Majeure.	
	16.6	Further Assurances.	
	16.7	Governing Law; Jurisdiction and Forum; Waiver Of Jury Trial	
	16.8	Headings.	
	16.9	Independent Contractors.	28

# 12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 30 of 126

<b>16.10</b>	Notices.	28
	Public Announcements.	
16.12	Amendments and Waivers.	29
16.13	Severability.	30
	No Third Party Beneficiaries.	
	Order of Precedence.	

# 12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 31 of 126

## **SCHEDULES**

Schedule A-1	Parent Services
Schedule A-2	Reverse Services
Schedule A-3	Planned IT Projects
Schedule B	Governance Model
Schedule C	Pricing Methodology
Schedule C-1	Pricing for Parent Services
Schedule C-2	Pricing for Reverse Services
Schedule D-1	List of AFI Recipients and Supported Facilities
Schedule D-2	List of ResCap Recipients and Supported Facilities
Schedule E	Form of Supplement

#### SHARED SERVICES AGREEMENT

This Shared Services Agreement (this "**Agreement**") is entered into on this \_\_ day of May , 2012 and effective as of the date this Agreement is approved by the Bankruptcy Court (the "**Effective Date**") by and between Residential Capital, LLC, a Delaware limited liability company ("**ResCap**") and Ally Financial Inc., a Delaware Corporation ("**AFI**").

NOW THEREFORE, in consideration of the mutual promises and covenants contained herein, and for other good and valid consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby agree as follows.

#### 1. DEFINITIONS.

- **1.1** <u>Definitions</u>. As used in this Agreement, the following terms have the following meanings:
- "Additional Services" has the meaning set forth in Section 3.2.
- "Affiliate" means, with respect to any specified Person, (a) those other Persons with respect to which such specified Person directly or indirectly owns or controls more than 50% of the voting equity securities of such other Person, (b) those other Persons that directly or indirectly own or control more than 50% of the voting equity securities of such specified Person, and (c) as to the Persons identified in the preceding clauses (a) and (b), those Persons that directly or indirectly own or control more than 50% of the voting equity securities of such identified Persons; provided, however, that, for purposes of this Agreement: (a) ResCap and its direct and indirect Subsidiaries shall not be deemed to be Affiliates of AFI; and (b) only the direct and indirect Subsidiaries of ResCap shall be deemed to be the Affiliates of ResCap.
- "Bankruptcy Court" means the bankruptcy court in which the chapter 11 case(s) of ResCap and/or its Subsidiaries are pending.
- "Bankruptcy Code" means title 11 of the United States Code.
- "Buyer" means one or more purchasers of all or substantially all of the assets of ResCap and certain of its Subsidiaries pursuant to the Sale.
- "Change" means any change or modification in the Services, or the schedule for performing such Services.
- "Charges" has the meaning set forth in <u>Schedule C</u>.
- "Claim" has the meaning set forth in Section 13.1(d)(1).
- "Claim Notice" has the meaning set forth in Section 13.1(d)(1).
- "Completion" has the meaning set forth in Section 14.3(c)(3).
- "Confidential Information" has the meaning set forth in Section 9.1.
- "Customized Services" has the meaning set forth in Section 3.3(a).
- "Damages" has the meaning set forth in Section 13.1(a).
- "Dedicated Assets" means any Supplier's third party contracts, Supplier Equipment, Supplier

Personnel and Supplier Software primarily dedicated to providing Services to Recipient.

- "Equipment" means computer and telecommunications equipment (without regard to the entity owning or leasing such equipment) including: (a) servers, personal computers, and associated attachments, accessories, peripheral devices, printers, cabling and other equipment; and (b) private branch exchanges, multiplexors, modems, CSUs/DSUs, hubs, bridges, routers, switches and other telecommunications equipment.
- "Force Majeure Event" has the meaning set forth in Section 16.5(a).
- "Functional Service Areas" means the categories of Parent Services or Reverse Services that are set forth under <u>Schedule C-1</u> or <u>Schedule C-2</u>, as applicable
- "Governmental Authority" means any government or political subdivision, board, commission or other instrumentality thereof, whether federal, state, local or foreign.
- "Include" and its derivatives means including without limitation. This term is as defined whether or not capitalized in this Agreement.
- "Indemnified Parties" has the meaning set forth Section 13.1(a).
- "Indemnifying Party" has the meaning set forth Section 13.1(d).
- "Initial Term" has the meaning set forth in <u>Section 2.1</u>.
- "Intellectual Property" or "IP" means any of the following, whether subsisting now or in the future anywhere in the world: (a) patents and pending patent applications; (b) trademarks, service marks, trade names and trade dress, and associated goodwill and rights of publicity and all rights associated therewith; (c) copyrights, including copyrights in works of authorship and computer Software; (d) confidential and proprietary information, including trade secrets and proprietary algorithms; (e) data base rights; (f) design rights and rights in designs; (g) rights in domain names; (h) rights in know-how; (i) all other intellectual property rights subsisting now or in the future, anywhere in the world; and (j) registrations, right to register and pending applications for registration of the foregoing, renewals, extensions, continuations, divisions or reissues thereof now or hereafter in force throughout the world (including rights in any of the foregoing); and (k) any and all causes of action arising from or related to any of the foregoing.
- "Intellectual Property Rights" means any and all common law, statutory and other rights in Intellectual Property honored and/or enforceable under any Laws.
- "Law" means any applicable order, writ, injunction, decree, judgment, ruling, statute, law, rule or regulation of any federal, state, municipal or local government, governmental authority, regulatory or administrative agency, governmental commission, department, board, bureau, court or tribunal or any arbitrator or arbitral body.
- "New Developments" means new Systems implemented by Supplier that are substantially different from the existing Systems, including major replacements to existing Systems and any directly related major replacements to work processes, policies and procedures implemented by Supplier.
- "Parent Services" has the meaning set forth in Section 3.1(a).
- "Parties" means AFI and ResCap.
- "Person" means any individual, partnership, firm, corporation, association, joint venture, limited

liability company, trust or other entity, or any governmental entity.

- **''Plan''** means ResCap's and certain of its Subsidiaries' chapter 11 plan(s) of reorganization or liquidation under the Bankruptcy Code.
- "Planned IT Projects" has the meaning set forth in Section 3.1(g).
- "Pre-Existing Intellectual Property" has the meaning set forth in Section 7.1.
- "Recipient" (i) in the case of Parent Services means ResCap and its Affiliates listed on <u>Schedule D-1</u>, and (ii) in the case of Reverse Services means AFI and its Affiliates listed on <u>Schedule D-2</u>.
- "Recipient Data" has the meaning set forth in Section 8.1.
- "Recipient Equipment" means all Equipment owned or leased by a Recipient and used in connection with the Services.
- "Recipient Facilities" has the meaning set forth in Section 6.1(a).
- "Recipient Software" means all Software owned by, or provided under license to, Recipient and used in connection with the Services (and all modifications, replacements, upgrades, enhancements, documentation, materials and media relating to the foregoing).
- "Recipient System" means an interconnected grouping of Recipient Equipment and/or Recipient Software used in connection with the Services, and all additions, modifications, substitutions, upgrades or enhancements thereto.
- "Relationship Manager" has the meaning set forth in Section 5.1.
- "Required Consents" means the consents, if any, required from third parties in connection with Supplier's provision, and Recipient's receipt, of the Services.
- "Reverse Services" has the meaning set forth in Section 3.1(a).
- "Sale" means a sale of all or substantially all of the assets of ResCap and/or certain of its Subsidiaries pursuant to the Plan or section 363 of the Bankruptcy Code.
- "Services" means all or any of Parent Services, Reverse Services, Customized Services, Termination Assistance Services, provided by a Supplier, as applicable given the context.
- "**Software**" means programs and programming (including the supporting documentation, media, on-line help facilities and tutorials).
- "Statement(s) of Work" means the mutually agreed internal documents that set forth the Services in respect of the Functional Service Areas that are identified in <u>Schedule A-1</u> or <u>Schedule A-2</u>, as applicable.
- "Subcontractors" means Supplier's contractors or other agents of Supplier that perform a portion of the Services.
- "Subsidiaries" means, with respect to any specified Person, those other Persons with respect to which such specified Person directly or indirectly owns or controls more than 50% of the voting equity securities of such other Person.
- "Supplement" has the meaning set forth in Section 3.2.
- "Supplier" means (i) in the case of Parent Services, AFI and any AFI Affiliates that provide

such Services, and (ii) in the case of Reverse Services, ResCap, and any ResCap Affiliates that provide such Services. For purposes of clarity, all references to "Supplier" refer only to the Party that is acting as the "Supplier" with respect to the Equipment, Software and other applicable responsibilities for the Services being performed by such Party in its capacity as the "Supplier".

- "Supplier Equipment" means Equipment owned or leased by the Party that in its capacity as the "Supplier" hereunder, or an Affiliate or Subcontractor of such Party, and used in connection with the Services.
- "Supplier Facilities" has the meaning given in <u>Section 6.2</u>.
- "Supplier Personnel" means those employees, representatives, contractors, Subcontractors and agents of the Party that is acting in its capacity as the "Supplier" hereunder, who perform any Services under this Agreement.
- "Supplier Software" means all software programs and programming owned by, or provided under license to, the Party that is acting in its capacity as the "Supplier" and used to provide the Services (and all modifications, replacements, upgrades, enhancements, documentation, materials and media relating to the foregoing).
- "Supplier System" means an interconnected grouping of Supplier Equipment and/or Supplier Software used in connection with the Services, and all additions, modifications, substitutions, upgrades or enhancements thereto.
- "Systems" means Recipient Systems and Supplier Systems or any of them.
- "Term" has the meaning set forth in Section 2.2.
- "Termination Assistance Services" has the meaning set forth in Section 14.5(a).
- "Termination Service Periods" has the meaning set forth in Section 14.5(a).
- "Third Party Claim" has the meaning set forth in Section 13.1(d)(1).
- "Third Party Claim Notice" has the meaning set forth in Section 13.1(d)(1).
- "Transition Services Agreement" has the meaning set forth in Section 14.5(a).

#### 2. TERM AND RENEWAL

- **2.1** <u>Initial Term.</u> The initial term of this Agreement begins on the Effective Date and continues through and until midnight Eastern Time on the one (1) year anniversary of the date on which ResCap and certain of its Subsidiaries file voluntary petitions ("**Petitions**") for relief under Chapter 11 of the Bankruptcy Code, as amended (the "**Initial Term**"), unless earlier terminated or extended in accordance with the terms of this Agreement and except for any Service that has been terminated in accordance with the terms of this Agreement.
- **2.2** Renewal. The Initial Term will be automatically extended for additional periods of one (1) year each unless either Party provides to the other Party written notice of nonrenewal at least three (3) month prior to the expiration of the then-current Term. The Initial Term as extended by any such additional period(s) (if any) shall be referred to as the "**Term**".

#### 3. SERVICES.

#### 3.1 Services.

- (a) <u>Performance</u>. AFI will provide the Services within each of the Functional Service Areas listed in <u>Schedule A-1</u> (the "Parent Services"), and ResCap will provide the Services within each of the Functional Service Areas listed in <u>Schedule A-2</u> (the "Reverse Services"), in each case beginning on the Effective Date. Services provided by a Supplier under this Agreement may be provided by that Supplier directly or through any of its Affiliates and/or Subcontractors at Supplier's discretion. A Supplier will not be relieved of any of its obligations under this Agreement as a result of the provision of Services by any of Supplier's Subsidiaries or other Supplier Personnel pursuant to this <u>Section 3.1(a)</u>. Supplier agrees that it will not enter into any new Subcontracting arrangement that would involve the transfer to a Subcontractor or another third party of Recipient's personal information, without the prior written consent of the other Party. The Supplier Personnel providing Services will at all times be qualified to provide the Services assigned to them.
- (b) <u>Recipients and Facilities</u>. Supplier will provide the Services to Recipient at the Recipient Facilities for which such Services are provided as of the Effective Date or, with respect to any particular Services, such other locations as may be specifically identified in <u>Schedule D-1</u> or <u>Schedule D-2</u> with respect to such Services. Supplier will negotiate in good faith to provide Services in support of any new Recipient or additional Recipient Facility, but shall not be obligated to provide Services to new Recipients or additional Recipient Facilities unless the pricing and terms for such new Recipients or facilities has been agreed upon by the Parties.
- (c) Service Standards. Supplier will perform or cause to be performed the Services referenced in the applicable Statement(s) of Work, to Recipient (i) with reasonable skill, care and diligence; and (ii) performed in substantially the same manner (including historic levels of service, historical usage levels and geographic provisioning) that such Services were generally performed by Supplier for Recipient immediately prior to the Effective Date and thereafter in substantially the same manner (including historic levels of service, historical usage levels and geographic provisioning) as Supplier generally performs such Services for its own businesses in accordance with its then-current processes and procedures (except to the extent such Services differ because of the need to follow legal corporate formalities and to keep Recipient Data separate from Supplier data, or as may be mutually agreed through the Parties' change control principles pursuant to Sections 3.2 and 3.3). In no event will Supplier be required to make any customization to the Services (or Supplier's associated Systems or processes) that are unique to Recipient, except for customizations that are expressly agreed upon in accordance with **Sections 3.2** and **3.3**. Each Party acting in its capacity as Recipient shall, and such Party shall cause all of its Affiliates that are Recipients to, comply with any applicable terms and conditions of third party contracts used by Supplier in connection with the Services.
- (d) <u>Implied Services</u>. If any services, functions or responsibilities performed by Supplier for Recipient as of the Effective Date are not specifically described in this Agreement but are necessary for Supplier to perform or provide the Services described in this

Agreement in accordance with <u>Section 3.1(c)</u>, such services, functions or responsibilities shall be deemed to be included within the scope of the Services to the same extent and in the same manner as if specifically described in this Agreement, except to the extent otherwise specified in this Agreement and excluding all services, functions and responsibilities of Recipient under this Agreement (including any services, functions or responsibilities not specifically described in this Agreement but are necessary for Recipient to perform or provide such services, functions and responsibilities described in this Agreement). Except as otherwise expressly provided in this Agreement, as between the Parties, Supplier shall be responsible for providing the facilities, personnel and other resources as necessary to provide the Services consistent with the requirements of <u>Section 3.1(c)</u>.

- (e) <u>Not a Requirements Contract</u>. This Agreement will not be construed as a requirements contract and will not be interpreted to prevent Recipient from obtaining from third parties, or providing itself, any or all of the Services or similar services.
- (f) <u>Dedicated Assets</u>. Upon the reasonable request of either Party, the Parties will work together in good faith to identify the Dedicated Assets, and with respect to Supplier Personnel that are Dedicated Assets, on a Service-by-Service basis: (i) the number of individuals providing such Service; (ii) the title of each such individual providing such Service (e.g., senior accountant, deputy general counsel, human resources manager, etc.); and (iii) the amount of time that each such individual spends providing such Service on a full-time equivalent (FTE) basis (e.g., 0.4 FTE).
- (g) <u>Planned IT Projects</u>. <u>Schedule A-3</u> to this Agreement sets forth a list of IT projects that are planned or ongoing as of the Effective Date (the "Planned IT Projects"). ResCap acknowledges and agrees that (i) the Services will be modified during the Term by and in accordance with each Planned IT Project, and (ii) the Planned IT Projects are necessary for the maintenance, upgrade, safety, or security of the AFI environment. ResCap will pay to AFI the portion of each Planned IT Project as applicable to Recipient, in accordance with <u>Schedule A-3</u> so long as ResCap or its Affiliates continue to receive Services that are based on or involve the use of AFI's IT environment. ResCap will provide to AFI cooperation and assistance as necessary or reasonably requested by AFI to allow AFI to implement each Planned IT Project.
- 3.2 Additional Services. A Party may, from time to time during the Term, upon at least ninety (90) days prior written notice to the other Party, request that the other Party provide additional services, functions and responsibilities not within the scope of the Services provided by the performing Party ("Additional Services"). Any such Additional Services will be provided under supplements to Schedule A-1 or Schedule A-2, as applicable entered into by the Parties ("Supplements") for the charges set forth therein and mutually agreed upon. Supplements shall be in the form of Schedule E. During the ninety (90) day period after the Effective Date, the Parties will work together to define a process reasonably satisfactory to the Parties that will be used to (a) submit and prioritize requests for Additional Services and (b) create and execute Supplements for Additional Services.
  - **3.3** Changes to Services/Change Control Procedures.

- (a) <u>Customized Services</u>. Supplier shall only provide Services customized for Recipient ("Customized Services") in accordance with this <u>Section 3.3</u>, and shall not otherwise be required to make customizations to the Services or Supplier Systems.
- (b) Changes Requested by Recipients. Recipient from time to time, upon at least thirty (30) days notice may require that the Supplier decrease or reasonably increase the Services provided to Recipient and the Charges shall be modified in accordance with the procedures set forth in Exhibit C. A Party may, from time to time during the Term, upon at least ninety (90) days prior written notice to the other Party, request that the other Party provide Customized Services. If the Parties mutually agree on the provision of Customized Services, such request for Customized Services and such Customized Services will be provided under Supplements to Schedule A-1 or Schedule A-2, as applicable. Before Supplier is required to provide any Customized Services, the Parties shall jointly agree on the applicable Charges for any agreed Customized Services, including any Charges that may be required to equitably compensate Supplier for any additional costs it may reasonably incur in connection with any changes to the Services.
- (c) Changes Required by Applicable Law. If Customized Services are necessary in order to comply with applicable Laws or changes to Supplier's third party contracts, Supplier will provide such Customized Services unless (i) such Customized Services require a material change to a Supplier System, or the implementation of a new Supplier System, or (ii) providing such Customized Services is not practicable given the then-current characteristics of the Supplier Systems, and the use thereof for Supplier and its Affiliates, and subject to the Parties jointly agreeing on the applicable Charges for such Customized Services. Any such Customized Services will be provided under Supplements to Schedule A-1 or Schedule A-2, as applicable. The Parties will negotiate in good faith in order to promptly agree on the applicable charges for any such necessary Customized Services. If providing such Customized Services is not practicable given the Supplier Systems, Recipient may purchase such services from a third party, provided that the Parties must agree on (A) the activities required to transition the affected Services from Supplier to such third party, (B) the impact on the remaining Services, and (C) the Charges associated with removing such Services and (D) the Charges for the remaining Services.
- 3.4 Recipients' Obligations. Supplier's failure to perform its obligations under this Agreement will be excused (and any rights of the Recipient arising as a consequence of such failure will not be exercised by the Recipient) if and to the extent such Supplier non-performance is caused by (a) the wrongful or tortuous actions of Recipient or a third party contractor performing obligations on behalf of Recipient under this Agreement, or (b) the failure of Recipient or such a third party contractor to perform such Recipient's obligations under this Agreement. Recipient will be responsible for any additional costs incurred by Supplier in connection with providing the Services as a result of any such failure. Supplier will use commercially reasonable efforts to perform its obligations notwithstanding such failure, provided that Recipient works with Supplier through the Relationship Managers to remedy the failure.

#### **3.5** Required Consents.

- (a) Responsibility. Each Party will be responsible for obtaining any Required Consents required under its own third party contracts pertaining to any Software, Equipment, Systems or other materials or associated services required in connection with the Services under this Agreement. Such responsibility shall include the administrative activities necessary to obtain the Required Consents and payment of the fees and/or expenses associated with obtaining the Required Consents. In the event of a pending Sale or upon termination of any Service(s) or this Agreement, the Parties will cooperate in order to determine and obtain any Required Consents necessary in order to transition any Software, Equipment, Systems or other materials to the other Party and/or the Buyer(s).
- (b) <u>Contingent Arrangements</u>. If, despite using commercially reasonable efforts, either Party is unable to obtain a Required Consent for which it is responsible under <u>Section 3.5(a)</u>, such Party will use commercially reasonable efforts to obtain a replacement license, product or right, as applicable. If such replacement cannot be obtained using commercially reasonable efforts, the Parties will work together in good faith to develop a mutually acceptable alternative arrangement that is sufficient to enable Supplier to provide, and Recipient to receive the Services without such Required Consent. The Party responsible for obtaining the Required Consent will be financially responsible for the costs of such alternative arrangement. If the Parties can not reach a resolution under <u>Section 3.5</u>, either Party may require that the affected Services be discontinued in which case the Charges for Services will be equitably adjusted to account for such discontinuation.
- 3.6 Security Level; Additional Security Measures. Supplier may take physical or information security measures (a) that affect the manner in which the Services are provided to maintain Supplier's current level (or, if greater, an industry-standard level) of physical and electronic security (including data security and data privacy) during the Term and (b) that address any new security-related issues, including compliance with applicable law and governmental orders related to security and issues in connection with new technologies or threats. Supplier shall provide Recipient reasonable, prior written notice of any such physical or information security measures that are material to Supplier's delivery of the Services. Recipient shall provide all assistance reasonably requested by Supplier in connection with such security measures. Recipient shall pay to Supplier increased Charges for the applicable Services that may be required to equitably compensate Supplier for any additional costs it may reasonably incur in connection with such security measures with respect to the Services provided to Recipient under this Agreement.

#### 4. USE OF AFFILIATES AND SUBCONTRACTORS.

#### **4.1** Affiliates and Subcontractors.

(a) <u>Use of Affiliates and Subcontractors</u>. Subject to <u>Section 3.1(a)</u>, Supplier will have the right to use Affiliates and Subcontractors to assist Supplier in the provision of the Services. Supplier shall adhere to Supplier's then-current "Third Party Vendor Management" policies when engaging Subcontractors to perform Services under this Agreement. In respect of any material Services for which Supplier desires to engage a Subcontractor, Supplier shall seek and obtain Recipient's approval of such Subcontractor, which approval Recipient shall not unreasonably withhold or delay; provided, however, that

all Subcontractors engaged or otherwise performing services for Supplier as of the Effective Date are hereby deemed approved by Recipient.

- (b) Supplier Responsibility for Affiliates and Subcontractors. Supplier shall remain responsible for the Services performed by its Affiliates and Subcontractors to the same extent as if such Services were performed by such Supplier. In the event that Supplier's subcontractor(s) fails to perform their obligations for which Supplier is utilizing them under this Agreement, Supplier will not be responsible or liable for such failure. However, Supplier will exercise any rights that it may have under its contract with the Subcontractor to cause the Subcontractor to resolve or cure any such failure within a reasonable time period, taking into consideration the circumstances of such failure, the nature of the services and the impact that such failure has on the Recipient. If such failure is not resolved or cured within a reasonable time period, Supplier will exercise any rights that it may have under its contract with the Subcontractor in the same manner that Supplier responds to such failure with respect to the services such Subcontractor provides for Supplier or Supplier's Affiliates businesses, taking into consideration the circumstances of such failure, the nature of the services and the impact that such failure has on the Recipient. Supplier shall be Recipient's sole point of contact regarding the Services, including with respect to payment. In addition, each Party, in its capacity as Supplier, will monitor and manage (including any necessary audits as to data privacy and security) its Subcontractors being used to perform Services for Recipient in compliance with Supplier's third party vendor management policies and procedures throughout the Term and any renewal or extension of the Agreement.
- **4.2** <u>Compliance</u>. Supplier will cause its employees and agents as well as the employees of its Affiliates and Subcontractors while at Recipient Facilities, to comply with the personnel, operational, safety and security procedures, policies, rules and regulations applicable to Recipient employees and agents and the Recipient Facilities, of which they have been given notice in advance by Recipient.

#### 5. RELATIONSHIP MANAGEMENT.

- **5.1** Relationship Managers. Each Party will appoint an individual (each, a "Relationship Manager") who, from the Effective Date until replaced by the appointing Party, will serve as that Party's representative under this Agreement. Each Relationship Manager will (a) have overall responsibility for managing and coordinating the performance of the appointing Party's obligations under this Agreement, and (b) be authorized to act for and on behalf of the appointing Party concerning all matters relating to this Agreement. Neither Party will reassign a Relationship Manager, unless it provides at least ten (10) days prior written notice to the other Party. If a Relationship Manager ceases to be employed by the Party that appointed it or is reassigned by such Party, such Party will promptly appoint a new Relationship Manager and provide written notice to the other Party of the new Relationship Manager so appointed.
- **5.2** <u>Governance Model</u>. The Parties will conduct meetings and manage interactions in accordance with the governance model described in **Schedule B**.

- **5.3** Reports. Each Party will provide to the other Party the reports described in the applicable Statement of Work, in the format and at the frequencies specified therein. In addition, ResCap will provide AFI access to such information and reports in their possession and control about ResCap and its direct and indirect Affiliates as AFI requests from time to time for regulatory reporting, audit, risk management, compliance, corporate governance, bank and/or bank holding company supervision and/or examination, and or other business purposes.
- **5.4** Regulatory Review. Each Party will notify the other promptly of any formal request or order by a governmental agency or regulator or any internal or external audit examination or request to examine records regarding Recipient that are maintained by Supplier or to audit Supplier's performance of the Services. Supplier will cooperate with any such examination or audit. Recipient will reimburse Supplier for the actual and reasonable out-of-pocket costs Supplier incurs in connection with that examination or audit.
- 5.5 Books and Records; Audit. Each Party acting in its capacity as a Supplier will, and will cause its Affiliates providing Services hereunder to, keep books of account and other records, in reasonable detail and in accordance with generally accepted accounting principles, consistently applied, as to the Charges for providing the Services to Recipient pursuant to this Agreement, and shall make such books of account and other records available to Recipient (and to any Governmental Authority that desires to audit Recipient) for inspection during normal business hours in a non-disruptive manner so as not to disrupt Supplier's business operations during the Term and for twenty-four (24) months thereafter for the purpose of performing audits and inspections of Supplier, related to the Services performed under this Agreement to: verify the accuracy of Charges and invoices.
- **5.6** Each Party acting in its capacity as a Supplier shall, and will cause its Affiliates providing Services hereunder to, use commercially reasonable efforts to provide to such auditors, inspectors, regulators, and representatives, at Recipient's sole cost and expense, such assistance as they reasonably require. Recipient's auditors and other representatives shall comply with Supplier's security and confidentiality policies, procedures and requirements. All such inspections or audits may be performed only by an independent third party auditing firm of national standing that has a written agreement with Recipient in which such third party agrees (i) to confidentiality obligations no less protective of Supplier than Recipient's confidentiality obligations under this Agreement and (ii) not to share with Recipient the Supplier information provided in connection with such inspection or audit, other than the findings and conclusions of the audit report.
- **5.7** <u>Informal Dispute Resolution Procedures</u>. The informal dispute resolution procedures applicable to disputes under or in connection with this Agreement are set forth in **Schedule B**.
- **5.8** Continued Performance. Each Party agrees that it will, unless otherwise directed by the other Party, continue performing its obligations under this Agreement while any dispute is being resolved until this Agreement expires or is terminated in accordance with its terms, provided, however, that in the case of a dispute with regards to a Party's alleged failure to pay amounts in excess of two (2) times the average monthly Charges for the Services provided

hereunder to such Party or its Affiliates, Supplier may suspend its performance of the Services until the earlier of such dispute is resolved or this Agreement is terminated; <u>provided</u>, <u>further</u>, that if Recipient pays such disputed amounts, (a) Supplier will continue to perform its obligations under this Agreement and (b) such payment will not constitute a waiver of any claims Recipient may have with respect to such disputed amounts.

#### 6. FACILITIES.

#### **6.1** Use of Recipient Facilities.

- (a) General. Except as expressly set forth otherwise in any applicable Statement(s) of Work, each Party will, and will cause its Affiliates to, acting in its capacity as "Recipient", provide to Supplier, at no charge, the space, office furnishings, janitorial service, telephone service, utilities (including air conditioning) and office-related equipment, supplies, and duplicating services at Recipient's premises that Supplier may reasonably need to provide the Services (collectively, the "Recipient Facilities"). In addition, each Party will, and will cause its Affiliates to, acting in its capacity as "Recipient" provide necessary storage space for backup data files related to the Services and will provide additional storage space that may be required by any change in retention schedules required by Recipient. Supplier's employees will have reasonable access to the Recipient Facilities twenty-four hours a day, seven days a week; provided, however, that in times of emergency, turnaround or significant maintenance or construction activity, access may be restricted or denied if not required in connection with such emergency, turnaround, maintenance or construction. In such an event, Supplier will be excused from its performance of the Services to the extent Supplier is unable to provide the Services in accordance with the requirements of this Agreement as a result of such restricted access.
- (b) Relocation. If Recipient contemplates or makes a final decision to alter or relocate any of the Recipient Facilities or if a Change of Control is pending that will require Supplier to relocate any of its personnel or Equipment from any Recipient Facility and the alteration or relocation could reasonably be expected to impact the Services (including the cost to perform, timing, ability to perform or level of performance) then Recipient will provide Supplier with sufficient advance notice of that fact to allow Supplier a reasonable amount of time to prepare for and implement the alteration or relocation as it impacts Supplier. The Parties acknowledge that in the case of a Change of Control it is likely that personnel and Equipment of each Party will be required to be relocated from the other Party's facilities. The Parties will work together in good faith for a planned and orderly process for such relocation, timed as much as possible to coincide with the scheduled completion of migration of all Services provided by each of them for the other from such facility, and each Party will bear its own relocation costs and expenses. Before requiring Supplier to relocate from any Recipient Facilities, the Parties will agree on any adjustments to (i) the Services that may be required as a result of the alteration or relocation, and (ii) Supplier's Charges that may be required to equitably compensate Supplier for any additional costs it may reasonably incur in connection with the alteration or relocation.
- (c) <u>Supplier's Obligations</u>. Supplier will (i) keep the Recipient Facilities in good order, and (ii) not commit waste or damage to those facilities or use those facilities for

any purpose other than providing the Services (and appropriate incidental use for internal Supplier administrative tasks unrelated to other customer accounts or Supplier marketing efforts).

(d) Access to Recipient Systems. Supplier will limit its, and will require that all Supplier Personnel who have access to Recipient Systems will limit their, access to those portions of such Recipient Systems for which they are authorized in connection with the Services. Supplier will limit such access to those Supplier Personnel who are needed in order to provide the Services. Supplier will cooperate with Recipient in the investigation of any apparent unauthorized access by Supplier Personnel to the Recipient Systems.

#### **6.2** Supplier Facilities and Systems.

- (a) <u>Supplier Facilities</u>. Supplier may perform the Services in such facilities maintained by or for Supplier or its Affiliates or Subcontractors (collectively, "**Supplier Facilities**") as Supplier reasonably deems appropriate, so long as appropriate security procedures have been implemented and are being observed at the Supplier Facilities. While at Supplier Facilities, each Party will, and will cause its and its Affiliates' personnel to, in its capacity as "Recipient", comply with Supplier's reasonable security requirements and other relevant policies of which they have been given notice.
- (b) Access to Supplier Systems. Each Party in its capacity as Recipient will limit its, and will require that all Recipient personnel who have access to Supplier's computer or electronic data storage systems to, limit their, access to those portions of such systems for which they are authorized in connection with their receipt and use of the Services. Each Party in its capacity as Recipient will (i) limit such access to those Recipient personnel who are authorized to use the Services, (ii) subject to Recipient's record retention policy, make available, upon Supplier's request, to Supplier a written list of the names of each individual who has been granted such access, and (iii) adhere to Supplier's security rules and procedures for use of Supplier's systems. All user identification numbers and passwords disclosed to the Recipients to permit Recipient personnel to access the Supplier systems will be deemed to be, and will be treated as, Supplier's Confidential Information. Recipient will cooperate with Supplier in the investigation of any apparent unauthorized access by Recipient personnel to Supplier's systems.
- **6.3** Physical Security for Facilities. Supplier will be responsible for all security procedures at any Supplier Facilities. Each Party as Recipient will provide all necessary security personnel and security equipment at the Recipient Facilities. While at the Recipient Facilities, each Party as Supplier will cause all Supplier Personnel to comply with Recipient's physical security procedures, as made known to Supplier's Relationship Manager.

# 7. INTELLECTUAL PROPERTY AND PROPRIETARY RIGHTS

**7.1** Ownership of Pre-Existing Intellectual Property. Except as expressly provided in Section 7.2, and 7.3, nothing in this Agreement shall grant or transfer any rights, title or interests in any Intellectual Property invented or created before or after the Effective Date by

or on behalf of a Party and/or its Affiliates or otherwise controlled by or licensed to such Party and/or its Affiliates (the "**Pre-Existing Intellectual Property**"). A Party's use of the other Party's Pre-Existing Intellectual Property shall not modify the ownership rights set forth above.

- 7.2 <u>Development of Intellectual Property</u>. Subject to <u>Section 8</u>, as between the Parties (and without affecting Recipient's right, title and interest in and to Recipient Data or any Confidential Information of Recipient), all Intellectual Property developed or acquired by or for Supplier or any of its Affiliates in connection with providing the Services shall be owned by Supplier. Any services that rely on New Developments are outside the scope of the Services under this Agreement and therefore would have to be separately negotiated and agreed upon by the Parties.
- 7.3 Limited License to Use Supplier Work Processes and Software. Supplier grants to Recipient a limited, non-exclusive, non-assignable license, with the right for Recipient to grant sublicenses to its Affiliates, to use the work processes and Software owned by Supplier and/or its Affiliates that are provided to Recipient in connection with the Services solely to the extent necessary for Recipient to receive Services, and perform its responsibilities under this Agreement during the Term. THE SUPPLIER WORK PROCESSES AND SOFTWARE AND ALL DELIVERABLES ARE PROVIDED BY SUPPLIER ON AN AS-IS BASIS. SUPPLIER EXPRESSLY DISCLAIMS ANY REPRESENTATIONS OR WARRANTIES, EXPRESS OR IMPLIED, AS TO SUCH WORK PROCESSES AND SOFTWARE AND DELIVERABLES.
- **7.4** No Implied Licenses. Except as expressly specified in this Agreement, nothing in this Agreement will be deemed to grant to one Party, by implication, estoppel or otherwise, license rights, ownership rights or any other Intellectual Property Rights in any work processes, Software or other materials, data or information owned by the other Party or any Affiliate of the other Party.

#### 8. RECIPIENT DATA.

- **8.1** <u>Definition</u>. The term "**Recipient Data**" means (i) any data or information of Recipient or its respective vendors, customers or other business partners that is provided to or obtained by Supplier in the performance of its obligations under this Agreement, including data and information regarding Recipient's businesses, customers, operations, facilities, products, consumer markets, assets and finances in whatever form or format including in the form of any analysis or compilation of such data, and (ii) any data or information collected or processed in connection with the Services, even if such data or information is contained in reports, documentation, compilations or analyses provided to Recipient as part of the Services. For the avoidance of doubt, Supplier retains ownership of data pertaining to its performance of Services, including data pertaining to the volume and quality of the Services.
- **8.2** Ownership. As between Recipient and Supplier, Recipient owns and will continue to own all right, title and interest in and to all Recipient Data. To the extent that

Recipient Data is embedded or incorporated into reports and other documentation, analyses, compilations and other materials (including code or software) owned or licensed by Supplier ("Supplier Materials") and provided to and for use by Recipient as part of the Services, Supplier will not be deemed to have assigned or transferred any of its right, title or interest in or to any underlying Intellectual Property Rights thereto. Supplier hereby grants to Recipient a perpetual, irrevocable, royalty free, transferable (to a Buyer of all or substantially all of the core business of Recipient) license to use, reproduce, display and perform (whether publicly or otherwise) and modify (and have others exercise such rights on behalf of Recipient (or Buyer)) such Supplier Materials solely as necessary for Recipient's (or Buyer's) use in the ordinary course of its mortgage related business. Recipient's (or Buyer's) use (including, without limitation, the use by any third party on behalf of Recipient (or Buyer)) of the Supplier Materials is subject to the confidentiality obligations set forth below in Section 9 and any third party restrictions imposed on any Supplier Materials of which Supplier makes Recipient aware. Additionally, Recipient's ownership of the Recipient Data reflected in Supplier Materials shall not serve to transfer or otherwise affect any of Supplier's right, title and/or interest in and/or to any of underlying Intellectual Property Rights in any Supplier Materials. Supplier may not use Recipient Data for any purpose except: (x) to provide the Services; (y) as required by AFI as the parent of ResCap to meet its financial and regulatory reporting requirements; or (z) as otherwise permitted under this Agreement, nor may Supplier sell, assign, lease or otherwise dispose of or commercially exploit Recipient Data. Supplier acknowledges that Recipient is not restricted from using or disseminating Recipient Data in the format in which it is provided by Supplier or any different format.

**8.3** Data Security. Supplier will establish and maintain safeguards against the destruction, loss or alteration of Recipient Data in its possession that are no less rigorous than those for Supplier's operations. If Recipient reasonably requests additional safeguards for Recipient Data, Supplier will provide those additional safeguards as Additional Services under a new Supplement subject to Sections 3.2 and 3.3. Recipient has established backup security for data and keeps backup data files in its possession. If Recipient chooses to establish additional backup security and backup data files, it may do so, except that if such activities increase Supplier's cost of providing the Services or require Additional Services or Customized Services, such activities will be subject to Sections 3.2 and 3.3. In all instances, Recipient will ensure that Supplier will have access to the backup data files as Supplier reasonably needs to provide the Services.

#### 9. CONFIDENTIALITY.

9.1 Obligations. Each Party will retain the other Party's non-public, proprietary and confidential information with the same degree of care as it uses to avoid unauthorized use, disclosure, publication or dissemination of its own confidential information of a similar nature ("Confidential Information") in confidence and not disclose the same to any third party nor use the same, except as expressly permitted in this Section 9. As used in this Agreement, "Confidential Information" shall include all non-public information, in any form, furnished or made available directly or indirectly by one Party to the other that is (a) marked confidential, restricted, proprietary and/or with a similar designation and/or (b) provided under circumstances reasonably indicating that it is confidential, restricted and/or proprietary. The terms and conditions of this Agreement shall be deemed Confidential Information. In the case

of either Party, Confidential Information also shall include, whether or not designated "Confidential Information", (i) Software, materials and other Intellectual Property owned by the disclosing Party, and (ii) all non-public information concerning the operations, affairs and businesses of a Party or its Affiliates, the financial affairs of a Party or its Affiliates, and the relations of a Party or its Affiliates with its customers, employees and service providers (including customer lists, customer information, account information and consumer markets).

- **9.2** Excluded Information. Excepted from the obligations of confidence and non-use under this **Section 9** is that information which:
- (a) the receiving Party is legally required to disclose, which disclosure shall be made in accordance with the below provisions of Section 9.3;
  - **(b)** is independently developed by the receiving Party without reference to Confidential Information of the disclosing Party;
  - (c) was, at the time of disclosure to it, in the public domain;
  - (d) after disclosure to it, is published or otherwise becomes part of the public domain through no fault of the receiving Party;
  - (e) was in the possession of the receiving Party at the time of disclosure to it;
  - **(f)** was received after disclosure to it from a third party who had a lawful right to disclose such information to it without any obligation to restrict its further use or disclosure.
- **9.3** Compelled Disclosure. Notwithstanding the provisions of this **Section 9**, if the receiving Party becomes legally compelled to disclose any of the disclosing Party's Confidential Information, the receiving Party will promptly advise the disclosing Party of such legal requirement to disclose Confidential Information in order that the disclosing Party may seek a protective order, may interpose an objection to such disclosure, take action to assure confidential handling of the Confidential Information, or take such other action as it deems appropriate to protect the Confidential Information in the circumstances. The receiving Party will disclose only that portion of the disclosing Party's Confidential Information that it is legally required to disclose.
- **9.4** Return of Information. Except for Confidential Information for which a continuing license is granted to the receiving Party under this Agreement, upon written request by the disclosing Party or upon Termination or conclusion of the Agreement, all of the disclosing Party's Confidential Information in whatever form will be returned to the disclosing Party or destroyed by the Receiving Party and certified as such to the Disclosing Party, without retaining copies thereof, except that any instances of such Confidential Information in an archived form that are commercially impractical to return may be retained so long as the receiving Party does not access or make use of such Confidential Information after receipt of the written request for return from the disclosing Party.
  - **9.5** Exception. Notwithstanding anything else in this Section 9, Supplier will have a

right to disclose Recipient's Confidential Information to third parties to the extent reasonably necessary for Supplier to accomplish its responsibilities contemplated hereunder or for Supplier to comply with its obligations under applicable law, including its reporting obligations under applicable law, and, in the case of AFI, to use Confidential Information of ResCap to operate AFI's business, consistent with the manner such Confidential Information was used by AFI prior to the Effective Date; provided, however, that such disclosure to third parties will be made under confidentiality terms and conditions that are no less favorable to Recipient than the provisions of this <u>Section 9</u> or, if less favorable, that are consistent with Recipient's customary practice for the nature of the third party receiving Recipient's Confidential Information.

#### **9.6** Obligations.

- (a) Each Party's Confidential Information shall remain the property of that Party except as expressly provided otherwise by the other provisions of this Agreement. Except as otherwise provided in this Agreement, each Party shall each use at least the same degree of care, but in any event no less than a reasonable degree of care, to prevent disclosing to third parties the Confidential Information of the other as it employs to avoid unauthorized disclosure, publication or dissemination of its own information of a similar nature.
- **(b)** In the event of any disclosure or loss of any Confidential Information of the disclosing Party, the receiving Party shall notify the disclosing Party promptly upon become aware thereof.
- **9.7** Loss of Confidential Information. In the event of any disclosure or loss of any Confidential Information of the disclosing Party due to the fault of the receiving Party, the receiving Party shall promptly, at its own expense: (a) notify the disclosing Party in writing; and (b) cooperate in all reasonable respects with the disclosing Party to minimize the violation and any damage resulting therefrom.
- **9.8** No Implied Rights. Nothing contained in this Section 9 shall be construed as obligating a Party to disclose its Confidential Information to the other Party, or as granting to or conferring on a Party, expressly or impliedly, any rights or license to the Confidential Information of the other Party.

#### 10. COMPENSATION.

10.1 General. ResCap will pay to AFI the Charges as set forth in Schedule C-1a and elsewhere in Schedule C, and AFI will pay to ResCap the Charges as set forth in Schedule C-2a and elsewhere in Schedule C. Supplier will provide Recipient with estimated invoices (with information linking the Services delivered to the invoice amounts) on a monthly basis on or before the last day of each calendar month for all Services performed by Supplier and all related Charges incurred by Recipient during that month. The last business day of the calendar month is the cut-off for delivered Services and related Charges to be invoiced in the next calendar month. The estimated charges will be adjusted to actual charged by the last day of the following month. Recipient will pay the estimated and adjusted actual invoices within

45 days of the receipt of an invoice from Supplier. Any payment by Recipient is without prejudice of its right to contest the accuracy of any Charges.

- 10.2 <u>Taxes</u>. In addition to the prices determined pursuant to <u>Schedule C</u>, each Party will pay, and hold the other Party harmless against, all goods and services, sales, use, excise and other taxes, and other fees or assessments imposed by law in connection with the provision of the Services by the other Party, other than taxes measured by the other Party's net income. The Parties will cooperate with each other and use commercially reasonable efforts to assist the other in entering into such arrangements as the other may reasonably request in order to minimize, to the extent lawful and feasible, the payment or assessment of any taxes relating to the transactions contemplated by this Agreement; <u>provided</u>, <u>however</u>, that nothing in this <u>Section 10.2</u> will obligate Supplier to cooperate with, or assist, Recipient in any arrangement proposed by Recipient that would, in Supplier's sole discretion, have a detrimental effect on Supplier or any of Supplier's Affiliates.
- **10.3** Invoicing and Payment. Supplier will invoice Recipient in accordance with **Schedule C**. Payments for amounts past due will bear interest calculated on a per annum basis from the due date to the date of actual payment at a fluctuating interest rate equal at all times to the prime rate of interest announced publicly from time to time by Citibank, N.A. plus two percent (2%), but in no case higher than the maximum rate permitted by Law. Each Party will make payments under this Agreement by electronic funds transfer in accordance with payment instructions provided by the other Party in its capacity as Supplier from time to time.

#### 11. REPRESENTATIONS AND WARRANTIES.

- 11.1 <u>Services</u>. Supplier represents and warrants to Recipient that it will use the level of care in providing the Services required by <u>Section 3.1(c)</u>, or with respect to Services that are dedicated to Recipient (i.e., where Supplier does not provide similar services to itself or its Affiliates), Supplier will provide such Services using commercially reasonable efforts and in a workmanlike manner, and in accordance with commercially reasonable practices.
- 11.2 <u>Maintenance</u>. Supplier represents and warrants to Recipient that it shall provide for the maintenance of the Supplier Systems, Supplier Equipment and Supplier Software in the same manner that such functions were generally performed by Supplier for Recipient immediately prior to the effective Date and thereafter in a manner consistent with the standard of performance required by <u>Section 3.1(c)</u>, for such items to generally operate in accordance with the manner in which they operated in the past or in the manner in which they operate in the future in support of Supplier and its Affiliates.
- 11.3 <u>Authorization</u>. Each Party represents and warrants to the other Party that, subject to approval of this Agreement by the Bankruptcy Court: (a) it has the requisite corporate power and authority to enter into this Agreement and to carry out the transactions contemplated by this Agreement; and (b) the execution, delivery and performance of this Agreement and the consummation of the transactions contemplated by this Agreement have been duly authorized by the requisite corporate action on the part of such Party.
  - 11.4 Viruses. Each Party represents and warrants to the other Party that it shall use the

same efforts that it uses with respect to its own users and Systems, to avoid computer viruses from being introduced into the Systems of the other Party under this Agreement or the Systems that such Party is using to perform Services hereunder.

11.5 <u>Disclaimer</u>. EXCEPT AS EXPRESSLY SET FORTH IN THIS <u>SECTION 11</u>, NONE OF SUPPLIER, ITS AFFILIATES OR ANY OF THEIR RESPECTIVE REPRESENTATIVES MAKE OR HAVE MADE ANY REPRESENTATION OR WARRANTY, EXPRESS OR IMPLIED, AT LAW OR IN EQUITY, IN RESPECT OF THE SERVICES, INCLUDING WITH RESPECT TO (A) MERCHANTABILITY OR FITNESS FOR ANY PARTICULAR USE OR PURPOSE, (B) THE USE OF THE SERVICE BY RECIPIENT AFTER THE RECEIPT THEREOF, OR (C) THE PROBABLE SUCCESS OR PROFITABILITY OF RECIPIENT'S OR ANY OF RECIPIENT'S BUSINESS AFTER THE RECEIPT OF THE SERVICES.

#### 12. INSURANCE.

- 12.1 <u>Insurance</u>. At all times during the Term and any extension hereof, unless and until the effective date of the Plan, ResCap, its Affiliates, and all of their respective directors, officers and employees will be covered under AFI's existing policies of insurance. Upon and following the effective date of the Plan, (i) both Parties will procure and maintain in full force and effect their own insurance policies at their own expense and for their own benefit, as determined by each party independently, but consistent with any and all applicable federal, state and local laws or regulations, and (ii) AFI will use commercially reasonable efforts to continue to renew its current blended directors and officers liability and fiduciary liability insurance program (the "Current Program"), for a period of six years following the effective date of the Plan (the "Run Off Period"), on substantially the same terms and conditions as the Current Program and including prior acts coverage with respect to claims arising from acts or omissions that occurred prior to the Effective Date; provided that if AFI is unable to continue the Current Program for the entire Run Off Period despite its commercially reasonable efforts it shall promptly notify the Debtors and use best efforts to obtain run off coverage for the balance of the Run Off Period.
- 12.2 Risk of Loss. Supplier will be responsible for the risk of loss of, or damage to, any property of Recipient at a Supplier facility, unless and to the extent that such loss or damage was caused by the acts or omissions of Recipient and/or Recipient's Affiliates, and/or its and/or their employees, agents and/or Subcontractors. Recipient will be responsible for the risk of loss of, or damage to, any property of Supplier at a Recipient facility unless and to the extent that such loss or damage was caused by the acts or omissions of Supplier and/or Supplier's Affiliates, and/or its and/or their employees, agents and/or Subcontractors. The risk of loss of, or damage to, property in transit will remain with the Party arranging the shipment.

#### 13. INDEMNIFICATION AND LIMITATIONS ON LIABILITY.

#### 13.1 Indemnification.

(a) <u>Recipient Indemnification</u>. Recipient will indemnify, hold harmless and defend Supplier and its Affiliates and their respective directors, officers and employees

("Indemnified Parties"), from and against any and all liabilities, damages, penalties, judgments, assessments, losses, costs and expenses in any case, whether arising under strict liability or otherwise (including reasonable outside attorneys' fees) (collectively, "Damages") suffered or otherwise incurred due to a Third Party Claim arising from or out of or relating to this Agreement, including the performance (or failure to perform) by Supplier of its obligations under this Agreement; provided, however, that to the extent and in the proportion Damages also arise out of or relate to the gross negligence or willful misconduct of any Indemnified Party, then the indemnity under this Section 13.1(a) will not apply.

**(b)** <u>Indemnification by Supplier</u>. Supplier will indemnify, hold harmless and defend the Recipient Indemnified Parties from and against any and all Damages suffered or otherwise incurred due to a Third Party Claim to the extent arising from or out of or relating to the gross negligence or willful misconduct of any Supplier Parties; <u>provided</u>, <u>however</u>, that to the extent and in the proportion Damages also arise from or out of or relate to the performance (or failure to perform) by Recipient of its obligations under this Agreement, then Supplier's indemnity under this <u>Section 13.1(b)</u> will not apply.

#### (c) Infringement.

- (1) <u>Indemnity</u>. Supplier will indemnify, hold harmless and defend the Recipient Indemnified Parties, and Recipient will indemnify, hold harmless and defend the Supplier Indemnified Parties, from and against any and all Damages due to a Third Party Claim to the extent that the claim alleges that any information or materials provided by the Indemnifying Party to the Indemnified Party under this Agreement infringes or misappropriates any Intellectual Property Right of such third party in any country in which Services are performed or received under this Agreement.
- (2) <u>Exclusions</u>. Neither Party shall have any obligation or liability to the other Party to the extent any infringement or misappropriation is caused by:
- (i) modifications to any such information or materials made by the other Party, its Affiliates, or its third party contractors;
- (ii) any combination of the allegedly infringing information or materials with items not provided by the Indemnifying Party , unless such combination was approved or directed in writing by the Indemnifying Party;
  - (iii) a breach of this Agreement by the other Party;
- (iv) the failure of the other Party to use corrections or modifications provided by the Indemnifying Party offering equivalent features and functionality to the extent the Indemnifying Party notified the other Party that the failure to do so could result in infringement liability;
- (v) the content provided by the other Party and not resulting from the Indemnifying Party's modification of that content without the other Party's approval; or

- (vi) third party Software, except to the extent that such infringement or misappropriation arises from the failure of the Indemnifying Party to obtain the necessary licenses or to abide by the limitations of the applicable third party Software licenses.
- Third Party Software Indemnities. **(3)** As specified Section 13.1(c)(2)(vi), the foregoing infringement indemnification does not apply to third party Software. With respect to third party Software provided by Supplier or its Subcontractors pursuant to this Agreement, Supplier and its Subcontractors shall use commercially reasonable efforts to obtain intellectual property indemnification for Recipient (or obtain intellectual property indemnification for itself and enforce such indemnification on behalf of Recipient) from the suppliers of such Software comparable to the intellectual property indemnification generally available in the industry for the same Software products. With respect to third party Software provided by Recipient pursuant to this Agreement, Recipient shall use commercially reasonable efforts to obtain intellectual property indemnification for Supplier and its Affiliates (or obtain intellectual property indemnification for itself and enforce such indemnification on behalf of the Supplier and its Affiliates) from the suppliers of such Software comparable to the intellectual property indemnification generally available in the industry for the same Software products.
  - (d) <u>Indemnification Procedure</u>. The Party making a claim for indemnification under this <u>Section 13.1</u> is referred to as the "Indemnified Party" and the Party or other Persons against whom such claims are asserted under this <u>Section 13</u> are referred to as the "<u>Indemnifying Party</u>." All claims by any Indemnified Party under this <u>Section 13</u> will be asserted and resolved as follows:
- (1) In the event that any right, demand, claim, action and cause of action, assertion, notice of claim or assertion, complaint, litigation, suit, proceeding, formal investigation, inquiry, audit or review of any nature, civil, criminal, regulatory, administrative or otherwise, or any grievance or arbitration (each, a "Claim") is asserted or instituted in writing by any person or entity other than the Parties or their Affiliates that could give rise to Damages for which an Indemnifying Party could be liable to an Indemnified Party under this Agreement (such Claim, a "Third Party Claim"), the Indemnified Party will promptly send to the Indemnifying Party a written notice specifying the nature of such Third Party Claim, together with all information reasonably available to the Indemnified Party with respect to such Third Party Claim (a "Third Party Claim Notice"); provided, however, that a delay in notifying the Indemnifying Party will not relieve the Indemnifying Party of its obligations under this Agreement, except to the extent that such failure will have caused actual prejudice to the Indemnifying Party.
- (2) In the event of a Third Party Claim, the Indemnifying Party will have ten (10) business days after receipt of the Third Party Claim Notice relating to such Third Party Claim to elect to undertake, conduct and control, through counsel of its own choosing (provided that such counsel is reasonably acceptable to the Indemnified Party) and at its own expense, the settlement or defense of such Third Party Claim (in which case the Indemnifying Party will not thereafter be responsible for the fees and expenses of any separate counsel retained by any Indemnified Party except as set forth below). Notwithstanding an Indemnifying Party's election to appoint counsel to represent an Indemnified Party in connection with a Third Party Claim, an Indemnified Party will have the right to employ separate counsel, and the Indemnifying Party will bear the reasonable fees, costs and expenses of such separate counsel if

(i) the use of counsel chosen by the Indemnifying Party to represent the Indemnified Party would present such counsel with a conflict of interest, or (ii) the Indemnifying Party will not have employed counsel to represent the Indemnified Party within a reasonable time after notice of the institution of such Third Party Claim. If the Indemnifying Party elects to undertake such defense, it will promptly assume and hold such Indemnified Party harmless from and against the full amount of any Damages resulting from such Third Party Claim to the extent provided herein. If the Indemnifying Party elects to undertake such defense, (x) the Indemnified Party agrees to cooperate with the Indemnifying Party and its counsel in contesting such Third Party Claim, and, if appropriate and related to such Third Party Claim, the Indemnifying Party and the Indemnified Party will reasonably cooperate with each other in connection with defending such Third Party Claim, and (y) such Third Party Claim may not be settled or compromised by the Indemnifying Party without the prior written consent of the Indemnified Party, which consent shall not be unreasonably withheld or delayed; provided, however, that in the event any Indemnified Party settles or compromises or consents to the entry of any judgment with respect to any Third Party Claim without the prior written consent of the Indemnifying Party, such Indemnified Party will be deemed to have waived all rights against the Indemnifying Party for indemnification under this **Section 13**. If the Indemnifying Party does not undertake the defense of such Third Party Claim, the Indemnified Party will have the right to contest, settle, compromise or consent to the entry of any judgment with respect to such Third Party Claim, and, in doing so, will not thereby waive any right to recourse therefor pursuant to this Agreement; provided, however, that at any time during the course of the matter the Indemnifying Party may assume the defense of such Third Party Claim by written notice of the same to the Indemnified Party.

(3) In the event of a Third Party Claim, from and after the delivery of a Claim Notice under this Agreement, at the request of the Indemnifying Party, the Indemnified Party will grant the Indemnifying Party and its representatives all reasonable access to the books, records and properties of such Indemnified Party to the extent reasonably related to the matters to which the Claim Notice relates. All such access will be granted during normal business hours and will be granted under conditions that will not unreasonably interfere with the businesses and operations of such Indemnified Party. The Indemnifying Party will not, and will cause its representatives not to, use (except in connection with such Claim Notice or such Third Party Claim) or disclose to any third person or entity other than the Indemnifying Party's representatives (except as may be required by Law) any information obtained pursuant to this Section 13.1(c)(3), which is designated by the Indemnified Party as (or provided under circumstances in reasonably indicating that it is) confidential.

#### **13.2** Limitations on Liability.

(a) Subject to the specific provisions and limitations of this <u>Section 13.2</u>, it is the intent of the Parties that each Party will be liable to the other Party for all Damages incurred by the non-breaching Party arising out of the breach of or failure to perform any of the breaching Party's agreements, covenants or obligations under this Agreement. If a Party has a Claim for such Damages, it will promptly send to the breaching Party a written notice specifying the nature of such Claim, together with all information reasonably available to such non-breaching Party with respect to such Claim (a "Claim Notice"); provided, however, that a delay in notifying the breaching Party will not relieve the breaching Party of its obligations under this Agreement, except to the extent that such failure will have caused

actual prejudice to the breaching Party. In the event of such a Claim, the breaching Party will notify the non-breaching Party within 45 days of receipt of a Claim Notice whether or not the breaching Party disputes such Claim.

- **(b)** The total aggregate liability of Supplier under or in connection with this Agreement, regardless of the form of the action or the theory of the recovery, will be limited to:
- With respect to Damages arising out of or relating to the failure by **(1)** Supplier to meet the standards set forth in Section 11.1, subject to Section 13.2(b)(2), at Recipient's option, which option Recipient shall exercise in its reasonable discretion after reasonably consulting with Supplier and after exhausting the escalation process set forth in Schedule B, the replacement of (if applicable) or re-performance of, or repayment of the price paid for, such Service; provided, however, that if Recipient requests the replacement of or reperformance of such Service, Supplier shall, instead have the option of refunding to Recipient the price paid for such Service in those instances where re-performance or replacement of the Services cannot, in Supplier's good faith opinion be completed in a commercially reasonable manner. In the limited instances where Supplier determines that Supplier cannot replace or reperform the Services in a commercially reasonable manner, Recipient shall have the option to terminate those specific Services from the Agreement without cost to Supplier other than the refund for the Services not performed in accordance with the Agreement discussed above. In any such limited instance, Recipient may elect to terminate and no longer have any obligation to pay for those specific Services under the Agreement, and Supplier's obligation will be limited to the refund discussed above and any reasonable cooperation requested by Recipient in the transition of the terminated services to Recipient or a third party.
- (2) With respect to indemnity Claims under <u>Section 13.1</u> or other Claims arising out of or relating to the gross negligence or willful misconduct of Supplier, the fees actually received by Supplier from Recipient for Services during the twelve (12) months preceding the last act or omission giving rise to such Claims or, in the event such last act or omission occurs during the first twelve (12) months following the date hereof, an amount equal to twelve (12) times the fees paid for Services in the month preceding such last act or omission;
- (3) With respect to all other Damages under or in connection with this Agreement, the fees actually received by Supplier from Recipient for Services during the twelve (12) months preceding the last act or omission giving rise to such Damages or, in the event such last act or omission occurs during the first twelve (12) months following the date hereof, an amount equal to twelve (12) times the fees paid for Services in the month preceding such last act or omission.
- (c) Each Party will use its commercially reasonable efforts to mitigate Damages for which it seeks recourse hereunder, including by promptly pursuing recovery under available insurance policies, <u>provided</u>, <u>however</u>, that the failure of such Party to successfully mitigate such Damages will not affect such Party's right to seek recourse with respect to such Damages so long as such Party will have used its commercially reasonable efforts to mitigate.
- (d) Any amounts payable under <u>Article 13</u> will be calculated after giving effect to any proceeds received from insurance policies covering the Damages.

- (e) IN NO EVENT SHALL EITHER PARTY, ITS AFFILIATES OR ANY OF ITS OR THEIR DIRECTORS, OFFICERS, EMPLOYEES OR AGENTS BE RESPONSIBLE OR LIABLE TO THE OTHER PARTY OR ITS AFFILIATES FOR ANY INDIRECT, INCIDENTAL, CONSEQUENTIAL, SPECIAL, EXEMPLARY, OR PUNITIVE DAMAGES, OR FOR ANY LOSS OF PROFITS, LOSS OF REVENUE, LOSS RESULTING FROM INTERRUPTION OF BUSINESS OR LOSS OF USE OR DATA, EVEN IF THAT PARTY, ITS AFFILIATES OR ANY OF THEIR DIRECTORS, OFFICERS, EMPLOYEES OR AGENTS HAS BEEN ADVISED OF THE POSSIBILITY OF SUCH DAMAGES AND NOTWITHSTANDING ANY FAILURE OF ESSENTIAL PURPOSE OF ANY LIMITED REMEDY OF ANY KIND, UNDER ANY CONTRACT, NEGLIGENCE, STRICT LIABILITY OR OTHER THEORY, ARISING OUT OF OR RELATING IN ANY WAY TO THIS AGREEMENT OR ITS IMPLEMENTATION.
- 13.3 Indemnification and Limitations on Liability Relating to Negligence and Strict Liability. ALL INDEMNITIES AND LIMITATIONS ON LIABILITY CONTAINED IN THIS SECTION 13 WILL APPLY WHETHER OR NOT THE INDEMNITEE OR PARTY CLAIMING DAMAGES WAS OR IS CLAIMED TO BE PASSIVELY, CONCURRENTLY OR ACTIVELY NEGLIGENT, AND REGARDLESS OF WHETHER LIABILITY WITHOUT FAULT IS IMPOSED OR SOUGHT TO BE IMPOSED ON SUCH INDEMNITEE OR PARTY.
- 13.4 Exclusive Remedy. Neither Party will have any remedy arising out of or relating to the breach of this Agreement other than the indemnification remedies set forth in <u>Section</u> 13.1 and Claims for direct damages referred to in <u>Section</u> 13.2, in each case subject to the limitations and exclusions set forth in this **Section** 13.
- **13.5** <u>Insurance</u>. From and after the completion of a Sale, each Party will cause its insurers to waive their rights of subrogation against the other Party with respect to any Claim.

#### 14. TERMINATION.

- **14.1** <u>Termination</u>. Without limiting the rights of the Parties under any other provision of this Agreement, any Service to be provided under this Agreement may be terminated as follows:
- (a) by either Party acting in its capacity as "Supplier", upon written notice to the other Party if, after Recipient fails to pay the Charges for such Service when due in accordance with this Agreement, the terminating Party sends to the other Party an initial notice of such failure and the other Party fails to pay such Charges within forty-five (45) days of receipt of such initial notice; or
- (b) by either Party acting in its capacity as "Supplier", upon written notice to the other Party if, following a material breach by Recipient of this Agreement other than a payment default, the terminating Party sends to the other Party a notice of such material breach and the other Party fails to cure such material breach within 30 days; provided however, that if cure cannot reasonably be accomplished within such 30-day period, this Agreement may not be terminated by reason of such breach for so long as the other Party

commences a cure within such 30-day period and pursues such cure diligently to completion and such completion occurs within 90 days of such written notice; or

- (c) by either Party acting in its capacity as "Recipient", upon written notice to the other Party if, following a material breach by Supplier of this Agreement, the terminating Party sends to the other Party an initial notice of such material breach and the other Party fails to cure such material breach within 30 days of receipt of such initial notice; provided, however, that if cure cannot reasonably be accomplished within such 30-day period, this Agreement may not be terminated by reason of such breach for so long as the other Party commences a cure within such 30-day period and pursues such cure diligently to completion and such completion occurs within 90 days of such written notice;
- (d) by either Party acting in its capacity as "Recipient", upon at least ninety (90) days prior written notice to the other Party; or
  - (e) otherwise upon mutual agreement of the Parties.
- 14.2 <u>Termination Following Sale</u>. Upon the final approval by the Bankruptcy Court of a Sale, either Party in its capacity as a Supplier may within 14 days of such final approval elect to terminate the provision of the Services that such Party is required to provide to the other Party under this Agreement, effective upon the closing of such Sale. In the event of such termination, Recipient will provide Supplier with notice of any Termination Assistance Services required by Recipient within twenty-one (21) days after receipt of such notice.
- 14.3 Election of Terminating Party. In connection with any termination pursuant to this Section 14, the Party exercising such termination rights may elect to continue to receive the Services (other than Services so terminated) that the other Party provides to it.
  - 14.4 <u>Survival</u>. Any provision of this Agreement which contemplates performance or observance subsequent to any termination or expiration of this Agreement will survive any termination or expiration of this Agreement and continue in full force and effect including, but not limited to, the following: <u>Sections 8.2</u> (Ownership), <u>9</u> (Confidentiality), <u>10</u> (Compensation), <u>11</u> (Representations and Warranties), <u>12</u> (Insurance), <u>13</u> (Indemnification and Limitations on Liability), <u>14.4</u> (Survival), <u>14.5</u> (Rights Upon Termination or Expiration; Sale), <u>16.3</u> (Counterparts), <u>16.5</u> (Force Majeure), <u>16.6</u> (Further Assurances), <u>16.7</u> (Governing Law; Jurisdiction and Forum; Waiver of Jury Trial), <u>16.8</u> (Headings), <u>16.11</u> (Public Announcements), and <u>16.13</u> (Severability).

#### **14.5** Rights Upon Termination or Expiration; Sale.

- (a) <u>Termination or Expiration</u>. Commencing six (6) months prior to expiration of the Term or any extension thereof, or upon either Party receiving any notice of termination from the other Party pursuant to the above provisions of <u>Section 14</u>:
- (1) At Recipient's option, Supplier will provide to Recipient the termination assistance reasonably requested by Recipient to allow the Services to continue, and to facilitate the orderly transfer of responsibility for performance of the Services, including any migration of Recipient Data, to Recipient (collectively, "Termination Assistance Services"),

for a period of up to six (6) months (or, with respect to information technology Services, eighteen (18) months) after the effective date of such termination of Services or expiration of the Term (the "<u>Termination Services Periods</u>"); provided however, that, if mutually agreed upon in writing by the Parties (including with respect to adjustments in Charges), the Termination Service Periods may be extended until the effective date of the Plan.

The Termination Assistance Services to be provided during the Termination Services Periods may include, as and if reasonably required by Recipient, the following, among other Services: (i) developing, together with Recipient, a plan for the orderly transition of the performance of the Services, including the migration of Recipient Data from Supplier to Recipient or to a third party designated by Recipient; (ii) providing reasonable training for personnel of Recipient in the performance of the Services then being transitioned to Recipient or the third party designated by Recipient, to the extent Services are being assumed by that third party; and (iii) providing cooperation to Recipient to facilitate the transition of the performance of the Services to Recipient or to a third party designated by Recipient, in accordance with a mutually agreed upon transition plan for the applicable Services. If Supplier provides any access to Supplier facilities, Systems or resources in connection with the execution of such plan, Recipient will, and will cause its personnel and any third parties having such access to, comply with Supplier's security and confidentiality requirements. Notwithstanding the foregoing, Supplier shall not be required to grant access to or share any of Supplier's proprietary information, data or technology with any third party.

(b) <u>Charges</u>. ResCap will pay to AFI Charges for the Services described in <u>Section 14.5(a)</u> as provided in <u>Schedule C</u>, including during the Termination Services Periods. AFI will pay to ResCap Charges for the Services described in <u>Section 14.5(a)</u> as provided in <u>Schedule C</u>, including during the Termination Services Periods. In the event of any termination of Services by Supplier for cause, if approved by the Bankruptcy Court, Charges for the related Termination Assistance Services will be paid by Recipient monthly in advance.

**14.6** Dedicated Assets. Upon the approval by the Bankruptcy Court of any Sale or as otherwise mutually agreed between the Parties to facilitate a Sale, AFI and ResCap each agree to promptly enter into good faith negotiations with the Buyer(s) in such Sale to agree on which, if any, Dedicated Assets are to be transferred to Recipient, an Affiliate of Recipient, or such Buyer(s), in connection with such Sale, and the terms and conditions of such transfer (including pricing and allocation of responsibility with respect to obtaining and paying for the associated Required Consents). Notwithstanding the allocation of responsibility for Required Consents in Section 3.5, except as otherwise agreed upon by the Parties, Recipient shall reimburse Supplier for any actual costs, losses, transfer fees or termination penalties incurred by Supplier in connection with the transfer or assignment to Recipient, an Affiliate of Recipient or such Buyer(s) of any Dedicated Assets. Supplier shall use good faith efforts to mitigate such costs including, to the extent feasible, by not entering into (or extending) during the Term such Supplier contracts that contain transfer fees or early termination penalties without the written consent of Recipient. For the avoidance of doubt, while each Party agrees to negotiate in good faith with respect to the transfer of Dedicated Assets, neither Party shall be obligated to transfer any Dedicated Assets, except as otherwise mutually agreed upon in writing by the Parties.

14.7 <u>Transition Services Agreement</u>. Upon the approval by the Bankruptcy Court of any Sale or as otherwise mutually agreed between the Parties to facilitate a Sale, AFI and ResCap each agree to promptly enter into good faith negotiations with the Buyer(s) in such Sale for an agreement pursuant to which AFI, ResCap and/or the Buyer(s) would provide or receive, as applicable, reasonable specified transition services in connection with such Sale (the "Transition Services Agreement"). The services to be provided to the Buyer and the terms relating to the transition of certain Services are to be addressed in the Transition Services Agreement. For the avoidance of doubt, while AFI agrees to negotiate in good faith with ResCap and/or a Buyer with respect to entering into a Transition Services Agreement, AFI is not obligated to provide services to a Buyer or after a Sale, other than Termination Assistance Services as contemplated in Section 14.5(a)(2) unless otherwise agreed by AFI.

# **FULLY INTEGRATED AGREEMENT.** The following provisions of this Section 15 are subject to the other provisions set forth in this Agreement relating to Recipient's right to increase the amount of, decrease the amount of and/or terminate any Service provided to Recipient by and/or on behalf of Supplier.

Each of the Parties represents, warrants, covenants, and agrees that this Agreement including the exhibits and schedules thereto comprise a single, unitary, indivisible, and non-severable agreement governing the operational arrangements between Supplier and its Affiliates on the one hand and Recipient and its Affiliates on the other hand. Although various schedules in this Agreement describe different services, all of the Services covered by this Agreement are highly integrated and interdependent, such that removal of any of the services would impair the delivery, value or usability of the remaining services. Accordingly, the description of those Services and the associated terms in separate documents, including separate Charges for the Services, does not signify that each constitutes a separate agreement; instead, such treatment is intended solely to facilitate articulation of the terms and conditions of the overall single, unitary and indivisible transaction. The use of expressions "single", "unitary", "indivisible", and "non-severable" to describe this Agreement is not merely for convenient reference. It is the conscious choice and express intent of the Parties to enter into a single, unitary, indivisible and non-severable transaction. Each of the Parties agrees that from an economic point of view this Agreement including all exhibits and schedules thereto reflect one indivisible and non-severable economic bargain between Supplier and Recipient, all other provisions of this Agreement and the provisions of each exhibit and schedule thereto, including such schedules and exhibits that may be executed following the Effective Date, have been negotiated and agreed to collectively as a single, composite, inseparable transaction, and that any one component of the transaction would not have been entered into other than as a part of the overall transaction. Except as expressly provided in this Agreement or any exhibit or schedule thereto for specific isolated purposes (and in such cases only to the extent expressly so stated, it otherwise being presumed that this paragraph is applicable), (i) all provisions of this Agreement including all exhibits and schedules thereto, including definitions, commencement and expiration dates, monetary provisions, use provisions, breach, default, setoff, recoupment, enforcement and termination provisions, and assignment, are integral to the entire transaction and are not severable; (ii) the economic terms of the transaction would have been substantially different had separate transactions been acceptable

to the Parties; and (iii) the provisions of this Agreement including all exhibits and schedules thereto will at all times be construed, interpreted and applied such that the intention of all Parties to effect a unitary, indivisible transaction will be preserved and maintained.

- Without limiting the foregoing <u>Section 15.1</u>, the Parties further agree that for all purposes, including any transfer, assignment, rescission, assumption, or rejection of this Agreement under Section 365 of the Bankruptcy Code or any amendment or successor section thereof, or otherwise, this Agreement including all exhibits and schedules thereto constitutes one indivisible, integrated and non-severable agreement dealing with and covering one legal and economic unit which must be transferred, assigned, rescinded, assumed, or rejected (as applicable) as a whole with respect to all (and not less than all) of the obligations covered under this Agreement including all exhibits and schedules thereto.
- 15.3 The terms and provisions of this Agreement govern the provision of all Services under this Agreement.

#### 16. GENERAL.

- 16.1 <u>Acknowledgement</u>. The Parties acknowledge that the terms and conditions of this Agreement have been the subject of active and complete negotiations, and that this Agreement will be construed as if drafted jointly by the Parties and no presumption or burden of proof will arise favoring or disfavoring either Party by virtue of the authorship of any provision of this Agreement.
- 16.2 <u>Binding Effect; No Assignment</u>. This Agreement is binding upon and inures to the benefit of the Parties and their respective successors, permitted assigns and legal representatives. This Agreement is not assignable by either Party without the prior written consent of the other Party, and any other purported assignment will be null and void.
- **16.3** Counterparts. This Agreement may be executed in two or more counterparts, each of which will be deemed an original, but all of which will be considered one and the same agreement, and will become effective when one or more counterparts have been signed by each of the Parties and delivered (including by facsimile) to the other Parties.
- 16.4 Entire Agreement. This Agreement, including the Schedules hereto, which are each hereby incorporated herein and made a part hereof, contains the entire agreement between the Parties with respect to the subject matter hereof. While purchase orders, invoices or similar routine documents may be used to implement or administer provisions of this Agreement, any provisions of such documents that add to, vary, modify or are at conflict with the provisions of this Agreement will be deemed deleted and will have no force or effect on either Party's rights or obligations under this Agreement.

#### **16.5** Force Majeure.

(a) "Force Majeure Event" means any event beyond the reasonable control of the Party affected that significantly interferes with the performance by such Party of its obligations under this Agreement, including acts of God, strikes, lockouts or industrial disputes or disturbances, civil disturbances, arrests or restraint from rulers or people,

interruptions by government or court orders or present and future valid orders of any regulatory body having proper jurisdiction (other than any such interruption arising from the failure by the Party claiming force majeure to comply with any applicable regulation or to obtain and comply with any required permit), acts of the public enemy, wars, riots, blockades, insurrections, inability to secure labor or secure materials upon terms deemed practicable by the Party affected (including by reason of allocations, voluntary or involuntary, promulgated by authorized governmental agencies), epidemics, landslides, lightning, earthquakes, fire, storm, floods, washouts, explosions, breakage or accident to machinery.

- (b) If a Force Majeure Event is claimed by either Party, the Party making such claim will orally notify the other Party as soon as reasonably possible after the occurrence of such Force Majeure Event and, in addition, will provide the other Party with written notice of such Force Majeure Event within five (5) days after the occurrence of such Force Majeure Event.
- (c) Except for a Party's obligations to make payments hereunder (including those obligations of <u>Section 10.3</u> hereof), neither Party hereto will be liable for any nonperformance or delay in performance of the terms of this Agreement when such failure is due to a Force Majeure Event and the Charges payable by Recipient shall be equitably adjusted such that Recipient is not be obligated to pay any Charges for any Services to the extent not performed due to a Force Majeure Event. If either Party relies on the occurrence of a Force Majeure Event as a basis for being excused from performance of its obligations hereunder, such Party relying on the Force Majeure Event will (i) provide an estimate of the expected duration of the Force Majeure Event and its probable impact on performance of such Party's obligations hereunder and (ii) provide prompt notice to the other Party of the cessation of the Force Majeure Event.
- (d) Upon the occurrence of a Force Majeure Event, the same will, so far as possible, be remedied as expeditiously as possible using commercially reasonable efforts. It is understood and agreed that nothing in this **Section 16.5(d)** will require the settlement of strikes, lockouts or industrial disputes or disturbances by acceding to the demands of any opposing party therein when such course is inadvisable in the discretion of the Party having the difficulty.
- **16.6** <u>Further Assurances</u>. Each Party covenants and agrees that, subsequent to the execution and delivery of this Agreement and without any additional consideration, each Party will execute and deliver any further legal instruments and perform any acts that are or may become necessary to effectuate the purposes of this Agreement.

#### 16.7 Governing Law; Jurisdiction and Forum; Waiver Of Jury Trial.

- (a) This Agreement will be governed by and construed in accordance with the laws of the State of New York without reference to the choice of law principles thereof.
- **(b)** Each Party irrevocably submits to the jurisdiction of the Bankruptcy Court in any action arising out of or relating to this Agreement, and hereby irrevocably agrees that

all claims in respect of such action may be heard and determined in the Bankruptcy Court. Each Party hereby irrevocably waives, to the fullest extent it may effectively do so, the defense of an inconvenient forum to the maintenance of such action or proceeding. The Parties further agree, to the extent permitted by law, that final and unappealable judgment against any of them in any proceeding contemplated above will be conclusive and may be enforced in any other jurisdiction within or outside the United States by suit on the judgment, a certified copy of which will be conclusive evidence of the fact and amount of such judgment.

- (c) Each Party waives, to the fullest extent permitted by Law, any right it may have to a trial by jury in respect of any action, suit or proceeding arising out of or relating to this Agreement. Each Party certifies that it has been induced to enter into this Agreement by, among other things, the mutual waivers and certifications set forth above in this <u>Section 16.7</u>.
- 16.8 <u>Headings</u>. The Section headings contained in this Agreement are inserted for convenience of reference only and do not affect the meaning or interpretation of this Agreement. All references to Sections contained herein mean Sections of this Agreement unless otherwise stated and except in the Schedules hereto, wherein references to Sections mean Sections of such Schedule unless otherwise stated.
- 16.9 <u>Independent Contractors</u>. Supplier is an independent contractor, with all of the attendant rights and liabilities of an independent contractor, and not an agent or employee of Recipient. Any provision in this Agreement, or any action by Recipient, that may appear to give Recipient the right to direct or control Supplier in performing under this Agreement means that Supplier will follow the desires of Recipient in results only.
- 16.10 <u>Notices</u>. All notices and other communications to be given to any Party hereunder are sufficiently given for all purposes hereunder if in writing and delivered by hand, courier or overnight delivery service or three days after being mailed by certified or registered mail, return receipt requested, with appropriate postage prepaid, or when received in the form of a facsimile or e-mail transmission and will be directed to the physical address, facsimile number or e-mail address set forth below (or at such other address or facsimile number as such Party will designate by like notice):

#### IF TO AFI:

Ally Financial Inc. 200 Renaissance Center, MC 200-B09-B-11 Detroit, Michigan 48265 Attention: Chief Financial Officer E-Mail Address: james.mackey@ally.com

With a copy to: Ally Financial Inc. Legal Staff 200 Renaissance Center, MC 200-B09-B-11 Detroit, Michigan 48265 Attention: IT Counsel

E-mail address: Thomas.lynch@ally.com

#### IF TO RESCAP:

Residential Capital, LLC 1100 Virginia Drive Fort Washington, PA 19034 Attention: James Whitlinger

E-Mail Address.: jim.whitlinger@gmacm.com

With a copy to: Residential Capital, LLC 1100 Virginia Drive Fort Washington, PA 19034 Attention: Tammy Hamzehpour

E-mail Address: tammy.hamzehpour@gmacm.com

- 16.11 Public Announcements. Except as otherwise required by law, each of AFI and ResCap will consult with the other and obtain the prior written consent of the other before issuing, or permitting any agent or Affiliate to issue, any press releases or otherwise making, or permitting any agent or Affiliate to make, any public statements with respect to this Agreement or the transactions contemplated hereby; provided, however, that the prior written consent of the other Party is not required hereunder with respect to any press release, public announcement or communication that is substantially similar to a press release, public announcement or communication previously issued with the prior written consent of the other Party.
- Amendments and Waivers. This Agreement may not be modified or amended except by an instrument or instruments in writing signed by both Parties. The Parties may, only by an instrument in writing, waive compliance by the other Party with any term or provision of this Agreement on the part of such other Party to be performed or complied with. The waiver by either Party of a breach of any term or provision of this Agreement will not be construed as a waiver of any subsequent breach. Except as otherwise expressly provided herein, no failure to exercise, delay in exercising or single or partial exercise of any right, power or remedy by either Party, and no course of dealing between the Parties, will constitute a waiver of any such right, power or remedy.
- 16.13 Severability. If any provision of this Agreement is held invalid, illegal or unenforceable, the validity, legality or enforceability of the other provisions of this Agreement will not be affected thereby, and there will be deemed substituted for the provision at issue a valid, legal and enforceable provision as similar as possible to the provision at issue.
- 16.14 <u>No Third Party Beneficiaries.</u> This Agreement is entered into solely between, and, except with respect to the rights of the Supplier Parties and the Recipient Parties under **Section 13**, may be enforced only by, Recipient and Supplier. This Agreement does not

# 12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 62 of 126

create any rights or causes of action in or on behalf of any third parties, including employees, suppliers, customers, Affiliates or Subcontractors of a Party, or create any obligations of a Party to any such third parties, except that either Party is entitled to include as part of its Damages under this Agreement the Damages incurred by its Affiliates or Subcontractors in connection with a breach by the other Party of this Agreement.

16.15 Order of Precedence. In the event of a conflict, this Agreement takes precedence over the Schedules.

[Signature Page Follows]

# 12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 63 of 126

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed by their authorized representatives, to be effective as of the Effective Date.

Ally Fi	nancial Inc.	Residential Capital, LLC
Ву: _	15 M	Ву:
Name:	Michael A. Carpenter	Name:
Title:	Chief Executive Officer	Title:

# 12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 64 of 126

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed by their authorized representatives, to be effective as of the Effective Date.

Ally Financial Inc.	Residential Capital, LLC
Ву:	By: Mu Mun
Name:	Name: Thomas Marano
Title:	Title: Chief Executive Officer

### **Schedule A-1**

### **Parent Services**

## The following Statements of Work are governed by the Agreement.

# AFI to ResCap Statements of Work by Functional Service Area

	F 4 10		SOW
	Functional Service Area	SOW Name	Number (if applicable)
1	Audit	Internal Audit Services	1G
2	Capital Markets	Capital Markets Services	1C
3	Communications	Communications and Investor Relations Services	-
4	Communications	eCommerce Services	1H
5	Compliance	Global Security Services	10
6	Compliance	Global Privacy Services	1N
0	Compliance	Global Filvacy Scrvices	111
	(includes Consent		
7	Order Costs)	Global Compliance Services	1M
8	Facilities	Facilities Services	1Ka
9	Finance Tax	Finance Shared Services	1a04
10	Finance Tax	Tax Services	1a08
11	Finance Tax	General Accounting Services	1A5
	Government		
12	Relations	Government Relations	1X
13	Human Resources	Human Resources Support Services	1Qa
14	ITG	Global Information Security	1R1
		Application Development and Support – IT Corporate	
15	ITG	Services	1R9
16	ITG	Cross Functional Services	-
17	ITG	End User Computing Services	-
18	ITG	IT Technology Strategy and Architecture Services	-
19	ITG	Voice Services Mobility SOW (Mid)_AFI to ResCap	-
20	ITG	Network Services – Local Area Network	-
		Network Services – Network WAN Network Services	
21	ITG	WAN (Mid)_AFI to ResCap	-
		Hosting Operations – Hosting, Implementation and Data	
22	ITG	Center Services	-
		Voice Services - Telecommunications and Contact Center	
23	ITG	Telecom Call Center AFI to ResCap	-
24	Legal	Legal Services	1S
25	Loan Review	Loan Review	1T
26	Marketing	Global Brand & Product Marketing	1I
27	Risk	Risk Management	1J
28	Supply Chain	Supply Chain Management Services	1P
29	Treasury	Treasury Services - Global Funding and Liquidity ("GF&L")	1W1
30	Treasury	Treasury Services Structured Funding Deal Compliance and Facility Management	1W2
31	Treasury	Treasury Services, Treasury Operations	1W7

### Schedule A-2

#### **Reverse Services**

### The following Statements of Work are governed by the Agreement.

# **ResCap to AFI Statements of Work by Functional Service Area**

	Functional Service		SOW Number (if
#	Area	SOW Name	applicable)
1	Accounting	Capital Markets Accounting	14L
2	Accounting	ResCap Accounting Services	-
3	Accounting	Accounting	14F
	Business Risk and Controls and Business Excellence	Business Risk and Controls and Business	
4	Support	Excellence Support	14J
5	Capital Markets	Capital Markets	14B
6	Capital Markets	Records Services	14I
7	Compliance	Line of Business Compliance Services	14H
8	Consumer Lending	Consumer Lending Services	14M
9	Finance	Mortgage Financial Planning and Analysis	14G
10	Human Resources	HR Support Services (AFI)	1Qb
11	Human Resources	HR Support Services (Bank)	14C
12	ITG	IT Resource Services	-
13	ITG	Application Support	14A
14	Legal	Legal Services (AFI)	1S
15	Legal	Legal Services (Bank)	14E
16	Masterservicing	Master Servicing for Ally Auto - Bond Administration	-
17	Masterservicing	Master Servicing for Ally Auto - Bond Modeling	-
18	Risk	Risk Services Reporting and Special Assets	-
19	Risk	Risk Management & Data Collection	14D
20	Risk	Client Repurchase	14N

**Schedule A-3** 

**Planned IT Projects** 

C	ľ	)
<	1	
(	1	)
3		5
7	Ç	5
9	1	)
7		5
ì	7	١

			Allo	Allocated Project Costs	ıts				
	Selected	2012 Total 3rd Party Expense	arty Expense	2012 People Expense	e Expense	2012 Total Project Expense	ject Expense	Mortgage Allocation	Allocation
AFI ==> Rescap	Driver	Monthly \$	Annual \$	Monthly \$	Annual \$	Monthly \$	Annual \$	Monthly \$	Annual \$
Global Functions Infrastructure	33.1%	128,835	1,546,021	13,760	165,120	142,595	1,711,141	47,199	566,388
Corporate Infrastructure	33.1%	34,614	415,364	3,645	43,736	38,258	459,100	12,664	151,962
Security	33.1%	20,000	000,009	22,054	264,646	72,054	864,646	23,850	286,198
Finance	33.1%	53,333	640,000	26,100	313,203	79,434	953,203	26,293	315,510
Treasury	33.1%	97,583	1,171,000	122,259	1,467,106	219,842	2,638,106	72,768	873,213
Compliance	33.1%	35,333	424,000	7,378	88,532	42,711	512,532	14,137	169,648
Legal	33.1%	60,083	721,000	50,439	605,273	110,523	1,326,273	36,583	438,996
HR	24.9%	52,083	625,000	43,152	517,822	95,235	1,142,822	23,707	284,485
Communications	24.9%	12,500	150,000	24,584	295,009	37,084	445,009	9,231	110,777
Supply Chain	33.1%	39,583	475,000	75,825	968'606	115,408	1,384,896	38,200	458,401
Capital Markets	33.1%	60,083	721,000	50,079	600,950	110,162	1,321,950	36,464	437,565
Technology Operations	16.1%	166,327	1,995,928	192,734	2,312,805	359,061	4,308,733	57,809	693,706
Total		790,359	9,484,314	632,008	7,584,098	1,422,368	17,068,412	398,904	4,786,850
					Memo: Direct Ch	Memo: Direct Charged Project Expense	nse		
						Shared Services - RTCoE Project	RTCoE Project		401,380
						Technology Ops. Project Expense	Project Expense		73,185
						Finance - Mortgage Ledger Risk Project	ge Ledger Risk Pr	oject	893,630
						Direct Charged Infrastructure Shared Services - TI\$	frastructure Shar	ed Services - TI \$	221,928
						Mortgage Infrastructure Project Expense	ructure Project Ex	chense	1,690,480
					Total Projects Dir	Total Projects Direct and Shared Expense (excl ResCap DW)	oense (excl ResCa	\$ (MQ d	\$ 8,067,453
				_	Rescap Data War	Rescap Data Warehouse Apr-Dec (Capital)	apital)		3,823,528
					ResCap Data War	ResCap Data Warehouse Apr-Dec (Expense)	xpense)		158,235
					Hardware purchased to date	sed to date			2,748,351

Schedule A-3 Planned IT Projects

~	٠
٣	J
◁	•
ч	
_	۹
u	,
=	=
_	7
=	-
_	j
ā	١
u	ı
	-
÷	
•	٦
~	:
allibada	1
٠.	•

	12-	12020-mg	Doc 41	Filed 05/		Entere	ed 05/14 126	/12 14:	50:46	Main Doo	cument
	v Spend Officia	\$36 Green \$2 Green \$0 Green \$0 Green \$0 Green \$79 Green \$79 Green	\$0 Green \$323 Green \$81 Green \$0 Green \$0 Green \$45 Green	\$20 Green \$200 Green \$20 Green \$0 Green \$0 Green \$0 Green \$0 Green	\$0 Green \$0 Green \$0 Green \$0 Green \$0 Green	\$0 Green \$0 Green \$0 Green \$0 Green \$0 Green	\$633 Green \$7,149 Green \$0 Green \$0 Green	\$150 Green \$150 Green \$150 Green \$150 Green \$115 Green	\$700 Green \$20 Green \$1 Green \$0 Green \$0 Green \$0 Green	\$50 Green \$24 Green \$0 Green \$181 Green \$45 Green \$63 Green \$63 Green \$63 Green \$63 Green	\$25 Green \$47 Green \$2 Green \$0 Green
	ference in Plan b/w and Capital										
	roject Expense (3rd Dif y + People Factor)	\$26 \$121 \$121 \$12 \$13 \$8 \$135 \$135 \$135 \$135 \$135 \$135 \$135	\$203 \$203 \$51 \$2 \$2 \$2 \$57 \$57 \$60 \$10	\$506 \$367 \$449 \$445 \$124 \$150 \$150 \$853	\$71 \$18 \$37 \$135 \$413	\$183 \$212 \$105 \$77 \$261	\$279 \$262 \$260 \$75 \$723 \$662	\$338 \$439 \$0 \$1,160 \$701 \$865	\$14 \$1.8 \$1.8 \$1.8 \$7 \$4	\$76 \$13 \$4 \$113 \$28 \$1 \$3,874 \$3,874 \$3,874	\$5 \$34 \$3 \$168 \$16 \$16 \$3 \$3 \$3 \$3 \$3 \$3 \$3 \$3 \$3 \$3 \$3 \$3 \$3
	le Expense Dollar Total Pr	\$5 \$5 \$5 \$5 \$5 \$5 \$5 \$5 \$5 \$5 \$5 \$5 \$5 \$	850 850 89 89 89 89	\$230 \$167 \$204 \$295 \$42 \$21 \$26 \$26 \$26 \$2313	\$32 \$8 \$17 \$61 \$187	\$83 \$97 \$48 \$35 \$119	\$70 \$74 \$74 \$21 \$475 \$435	\$188 \$244 \$0 \$645 \$390 \$265	8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	67 60 611 63 63 63 63 63 63 63 63 63 63 63 63 63	2. 2. 2. 2. 2. 2. 2. 2. 2. 2. 2. 2. 2. 2
	ple Expense Factor People (varies by Global Function)	10.7% 10.7% 10.7% 10.7% 10.7% 10.7% 10.7%	10.7% 10.7% 10.7% 10.7% 10.7% 10.7%	83.3% 83.3% 196.7% 20.9% 20.9% 20.9%	82.9% 82.9% 82.9% 82.9% 82.9%	83.9% 83.9% 83.9% 83.9% 83.9%	39.7% 39.7% 39.7% 39.7% 191.6%	125.3% 125.3% 125.3% 125.3% 125.3%	44.1% 10.7% 10.7% 10.7% 10.7%	10.7% 10.7% 10.7% 10.7% 10.7% 10.7% 10.7%	10.7% 10.7% 10.7% 10.7% 10.7% 10.7% 10.7% 10.7%
	Z IT Project Spend People Expense (vari	\$23 \$10 \$11 \$11 \$17 \$122 \$20 \$868	\$7 \$183 \$46 \$5 \$5 \$52 \$69 \$60	\$276 \$200 \$245 \$150 \$100 \$124 \$640	\$39 \$10 \$20 \$74 \$226 \$456	\$110 \$110 \$57 \$42 \$142	\$200 \$200 \$186 \$186 \$54 \$227	\$150 \$150 \$0 \$515 \$311 \$600	86 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	\$69 \$11 \$102 \$28 \$3 \$3,50 \$3,50 \$2,50	\$5 \$30 \$13 \$143 \$143 \$143 \$143 \$143 \$143 \$143
	2012 IT Project Spend Subtotal Subtotal (calculated IT Project Spend Expense + IT roject Spend Software	11 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	206 127 127 167 167 52 54 58	276 400 445 150 200 104 124 640	39 10 20 74 226	100 110 115 42 442 344	200 200 892 2,335 54 248 227	212 345 286 865 426 1,200	700 33 61 61 4	69 84 4 77 71 167 250	30 77 143 12 2 3 3 6 6 129 175
	Budget Allocated To	infrastructure S Infras	Infrastructure \$ Infras	Corp - Capital Markets S  Corp - Communications S  Corp - Communications S  Corp - Communications S  Corp - Compilance S  Corp - Compilance S  Corp - Compilance S  Corp - Compilance S		Corp. HR Sor Corp. Legal Sor C	Corp - teagan Corp - Risk Corp - Risk Corp - Risk Corp - Risk Corp - Sizk Corp - Sizk Corp - Supply Chain Corp - Supply Chain	Corp - Treasury S	Information Security Sinfrastructure Infrastructure	Infrastructure S Infras	infrastructure similaristructure similaristructu
	BU/GF Name	Global Functions Infrastructure	clobal Functions Infrastructure Global Functions Infrastructure Global Functions Infrastructure Global Functions Infrastructure Global Functions Infrastructure Global Functions Infrastructure Global Functions Infrastructure	Capital Markets Capital Markets Capital Markets Compliance Compliance Compliance Compliance Compliance Compliance Compliance Compliance Compliance			hain nain			Cop infrastructure	icture
Consolidated 2012 Project List	Projecu Initiative Name	EU EU! Enterprise Windows 7 Test Environment EU EU Winness LAN EU EU Winness LAN EU EU Brand Leise Lei	SEC EV Security Evergreen - Friewalls, Proxy, IPS - NADD (20%) SEC EV Security Evergreen - Friewalls, Proxy, IPS - NADD (20%) SEC EV Security Evergreen - Friewalls, Proxy, IPS - NADD (20%) SEC EV Security Evergreen - Friewalls, Proxy, IPS - NADD (20%) SEC EV Security Evergreen - Friewalls, Proxy, IPS - NADD (20%) SEC EV Security Evergreen - Friewalls, Proxy, IPS - NADD (20%) SEC EV Security Evergreen - Friewalls, Proxy, IPS - NADD (20%) NADD - Service Marria Service - Friewalls - Service - S	aces Reteases 22 Enhancement Reli Initiatives se program aining rade rade Resease Program	2017 Compensation Cycle Project 2012 HT Talent Mgmt & TA Release Pool 2012 HT Total Rewards Release Pool 2012 HT Workforce Admin and ER&P Release Pool 2013 Compensation Cycle Project 2013 Compensation Cycle Project	Author Program Implementation eDiscovery Release eDocs DM, Upgrade Team Room Release	rearitouritour, opgade Capen Release Pool Open Pages - Release Pool Open Pages - Technology Upgade Polytramche Upgade (Accutapel/Aploytraunche/NEV Tech Upgrade) Polytramcha Release Program Supplier Risk Management Program	Quartum Resease Program Security Compliance and Remediation Security Compliance and Remediation Sienra, Intelinated, SSR Release Master WebSeries Release Program WebSeries Release Program	Emerpries AMM R2 EU EU Firengries Windows 7 Test Environment EU EU Windess LAN Global Database Upgrade HC F5 Load Banding HT TITSS Storage and BUR Capacity Optimization	NW D2D - Telecom/Network - LAN - Corporate Port Admission (CPA) Transformation NW EV Datacenter LAN Expansion and Upgrade SEC EV Security Evergreen - Frewalls, Proxy, IPS - NADD (20%) SEC EV Security Evergreen - Frewalls, Proxy, IPS - NADD (20%) SEC EV Security Evergreen - Frewalls, Proxy, IPS - NADD (20%) SEC EV Security Evergreen - Frewalls, Proxy, IPS - NADD (20%) SEC EV Security Evergreen - Frewalls, Proxy, IPS - NADD (20%) The Service Mgmt Suite for Mortgage / LV The admical Support Resources Transformation	Vulnerability Scanning - Expansion for Webinspect / AMP EU EUT Empergines Windows 7 Test Environment Clear Universes LAM Global Darkhase Lipprade HT EV 2011 Evergreen-Win & UNIX Servers HT EV 5011 Evergreen-Win & UNIX Servers HT EV Balon on Rails 2.0.x + Mongel 1.x Upgrades and Replacement HT EV StanChe HT IT SS Stonge and BUR Capacity Optimization NW EV Dassagn Febriory Evergreen 2015

_	-12020-mg Doc 41	Filed 05/14/12 Entered 05/14/12 14:50:46 Pg 70 of 126	Main Document
and Capital	\$150 Green \$421 Green \$421 Green \$50 Green \$50 Green \$50 Green \$50 Green \$50 Green		
Differen	\$277 \$10 \$264 \$65 \$3 \$565 \$75 \$13 \$13 \$77 \$13 \$7779		
ioliar Total Project Expense (3rd Party + People Factor)	\$17.72 \$2.5 \$6.5 \$6.3 \$7.7 \$1.779		
People Expense Dollar			
People Expense Factor (varies by Global Function)	10.7% 10.7% 10.7% 10.7% 10.7% 10.7% 10.7% 29.7%		
2012 IT Project Spend Expense	\$20 \$238 \$238 \$60 \$60 \$67 \$1,800 \$6,000 \$6,000		
2012 IT Project Spend Subtoral (calculated IT Project Spend Expense + IT Project Spend Software	S		
Budget Allocated To	Infrastructure TO - CPMO TO - CPMO TO - RTOOE		
BU/GF Name	Mortgage - Infrastructure Tech Ops - SPNO Tech Ops - STOGE		
idated 2012 Project List Project Intrative Name	WI ANVoice Transition  with Sergreen - Firewalls, Proxy, IPS - NADD (20%)  with Sergreen - Firewalls, Proxy, IPS - NADD (20%)  with Sergreen - Firewalls, Proxy, IPS - NIDD (20%)  with Sergreen - Firewalls, Proxy, IPS - NIDD (20%)  with Sergreen - Firewalls, Proxy, IPS - NIDD (20%)  with Sergreen - Firewalls, Proxy, IPS - NIDD (20%)  working agreen Program  working Sergreen Program  recrification and the Expension for Webinspect / AMP  recrease Pool  aged Resources for BU NADD  ing - Projects to be Direct Charged at 100%  g - Panially Direct Charged  ding - Projects to Charged at 0%		

#### Schedule B

#### **Governance Model**

#### 1. INTRODUCTION

This <u>Schedule B</u> (this "Schedule B") is incorporated by reference and attached to the Shared Services Agreement by and between ResCap and AFI dated May\_\_\_, 2012 (the "Agreement").

#### 2. AGREEMENT COVERAGE

This **Schedule B** sets out the service management model for Supplier and Recipient (the "**Service Management Model**"), related to the Services provided under the Agreement.

#### 3. OVERVIEW

Supplier and Recipient will each appoint responsible, knowledgeable persons to serve as their representatives to oversee the performance of the Agreement (such persons and each Party's Relationship Manager shall collectively constitute the "Service Management Team").

The Service Management Team members will work informally on a regular basis, but the Relationship Managers, and such additional Service Management Team members as the Relationship Managers may designate from time to time, will meet formally at least monthly. These monthly meetings will review such topics as significant operational concerns, performance metrics as developed by the Service Management Team, future service needs, transition service progress and pricing. In addition, the Relationship Managers may from time to time designate Service Management Team members to meet in sub-teams either on a temporary or ongoing basis, to deal with specific issues under the Agreement. The Relationship Managers will share the responsibility for organizing the meetings and ensuring the effectiveness of the meetings.

Each Party will staff its Service Management Team as necessary to represent the areas of services offered.

#### 4. SCOPE

The Service Management Team will be responsible for overseeing the overall execution of the Agreement to ensure that each Party performs its responsibilities as expected.

The Service Management Team's responsibilities shall include the following:

(a) <u>Strategy and Direction</u>. The Service Management Team may make recommendations as to strategic direction and provide critical input for business planning decisions between Supplier and Recipient.

- (b) General Service Management. The Service Management Team may review the status of projects and prioritize work. The Service Management Team will also be a forum for discussions between Supplier and Recipient regarding service changes and opportunities for Additional Services, and will coordinate with management and service delivery teams to ensure appropriate execution of such service changes or Additional Services.
- (c) <u>Issue Resolution</u>. The Service Management Team will be responsible for resolving problems, concerns or inefficiencies in their execution as they arise, and will escalate unresolved issues as described below.

#### 5. ESCALATION

Issues that cannot be resolved within a reasonable period of time (which period of time shall not exceed twenty (20) days) by the Parties' respective Service Management Team members responsible for the applicable subject area must be first be escalated to the Relationship Managers. If the Relationship Managers are unable to resolve an issue within ten (10) days, then either Party's Relationship Manager may, upon prior written notice to the other Relationship Manager, escalate such issue to Recipient's and Supplier's executive management representative for resolution. Neither Party may initiate any formal legal proceedings for resolution of such issue until ten (10) business days after such issue has been escalated to each Party's executive management representatives.

The provisions and time periods specified in this <u>Section 5</u> will not be construed to prevent a Party from instituting, and a Party is authorized to institute, formal proceedings earlier to (A) avoid the expiration of any applicable limitations period, (B) preserve a superior position with respect to other creditors, or (C) address a claim arising out of the breach of a Party's obligations under <u>Section 9</u> of the Agreement or a dispute with respect to Intellectual Property Rights.

#### Schedule C

#### **Pricing Methodology**

#### 1. INTRODUCTION

This <u>Schedule C</u> (this "Schedule C") is incorporated by reference and attached to the Shared Services Agreement by and between ResCap and AFI dated May\_\_, 2012 (the "Agreement").

- 1.1 <u>Charges</u>. This <u>Schedule C</u> describes the Charges, and the methodologies for their calculation, with respect to the Services.
- 1.2 <u>Order of Precedence</u>. The Parties acknowledge that certain obligations may be set forth in both this Schedule and elsewhere in the Agreement, and in the event of a conflict, such conflict shall be resolved in accordance with <u>Section 16.15</u> of the Agreement.
- 1.3 <u>Section and Article References</u>. Unless otherwise specified, Section references in this <u>Schedule C</u> refer to the Sections of this <u>Schedule C</u>.
- 1.4 <u>Definitions</u>. Capitalized terms have the meanings assigned below. Any other capitalized terms used and not otherwise defined in this <u>Schedule C</u> have the meanings given them in the Agreement.
  - (a) "Business Change Event" means any change in the products or services offered by Recipient or the expansion or closure of any Facility of any Recipient, or a decrease or increase in Recipient's level of consumption of the Services from the historical usage levels of its business, in either case that has a material impact on the cost to Supplier of providing any impacted Services.
  - (b) "Charges" means:
    - (i) any Total Shared Services Charge;
    - (ii) amounts to be paid by Recipient as Pass-Through Expenses;
    - (iii) the charges for Additional Services that may be agreed upon by the Parties pursuant to a signed Supplement from time to time;
    - (iv) the charges for the Customized Services that may be agreed upon by the Parties pursuant to a signed Supplement from time to time; and
    - (v) the charges for Termination Assistance Services.

- (c) "Monthly Fixed Charges" means, for a given month and Service, the amount reflected in <u>Schedule C-1b</u> or <u>C-2b</u> for such Service. This amount does not include amounts to be paid by Recipient as Monthly Variable Charges or Pass-Through Expenses.
- (d) "Monthly Variable Charges" means, for a given month and Service, the (i) total actual costs incurred by Supplier for the Service, multiplied by (ii) the Recipient cost allocation percentage for such Service that was agreed upon between the Parties, as reflected in <u>Schedule C-1b</u> or <u>C-2b</u>, as applicable. This amount does not include amounts to be paid by Recipient as Pass-Through Expenses.
- (e) "Pass-Through Expenses" as referenced in Schedule C-1b and C-2b means those third-party, out-of-pocket expenses actually incurred by Supplier in connection with a given Service (e.g., external audit fees, training for consent orders, consent order direct vendor costs), which will be either (i) approved by Recipient prior to being incurred by Supplier; provided, that if Recipient does not approve any such expense Supplier will have no obligation to procure or provide the goods or services associated with such expense, (ii) approved by Recipient in advance as being within a category or for a Service already approved by the Parties as for a billed-as-incurred expense, or (iii) for a good or service reasonably necessary to allow Supplier to deliver the Services contemplated under this Agreement.
- (f) "Price Adjustment Event" means any of the following:
  - (i) a Business Change Event occurs; or
  - (ii) a Service has been terminated pursuant to <u>Section 3</u> or <u>Article 14</u> of the Agreement.
- (g) "Price Adjustment Process" means the process described in Section 3.4 of this Schedule C.
- (h) "Total Monthly Shared Services Charge" means, in respect of any individual Service, the monthly Charge for that Service, which shall be applicable from the Effective Date until the termination date for such Service, as such Charges are listed on Schedule C-1a or Schedule C-2a and may be adjusted in accordance with any Price Adjustment Event and/or Price Adjustment Process. The Total Shared Services Charge represents the aggregate of the Total Base Costs, Third Party Costs and IT Projects as referenced on Schedule C-1a and Schedule C-2a. The Total Base Costs are equal to an amount representing Employee Costs, IT Costs,

Platform Costs and the Indirect Support Costs for that Functional Service Area, as described below:

- (i) Employee Costs means compensation, benefits, travel and other non-compensation related personnel costs;
- (ii) IT Costs means those costs associated with IT services under Schedule A-1 and Schedule A-2;
- (iii) Platform Costs means the Charges to the Recipient for use of the IT services on Supplier's platform related to depreciation and amortization in respect of IT infrastructure; and
- (iv) Indirect Support Costs means, for AFI, an amount equal to the percentage of the total of Employee Costs, IT Costs and Platform Costs set forth in Schedule C-1b, and for ResCap means an amount equal to the percentage of the total of Employee Costs, IT Costs and Platform Costs set forth in Schedule C-2b. Such percentages in each case are those referenced in the formulas in the Pricing Methodology Spreadsheets titled "Shared Services Global Functions Exhibit v.11.pdf" (AFI to ResCap) and "ResCap to AFI Shared Services Pricing 05-13-12.pdf" (ResCap to AFI).

These amounts are set forth in <u>Schedule C-1a</u> and <u>Schedule C-2a</u> on a monthly basis as the Total Monthly Shared Services Charge. The Total Monthly Shared Services Charges do not include any (i) Charges for Additional Services or Customized Services, (ii) Termination Assistance Services or (iii) Pass-Through Expenses.

Pricing for the Services set forth on <u>Schedule C-1a</u> and <u>Schedule C-2a</u> can change on a monthly basis, depending upon whether the Service is priced on the basis of Monthly Fixed Charges or Monthly Variable Charges, as indicated on <u>Schedule C-1b</u> and <u>Schedule C-2b</u>. Any such calculations will be made by reference to and in accordance with the formulas in the Pricing Methodology Spreadsheets titled "Shared Services Global Functions Exhibit v.11.pdf" (AFI to ResCap) and "ResCap to AFI Shared Services Pricing 05-13-12.pdf" (ResCap to AFI).

#### 1.5 General.

- (a) Unless specifically stated otherwise, each Party will be financially responsible for all costs and expenses associated with performing its responsibilities under this Agreement.
- (b) Unless the Parties otherwise agree, for the purpose of calculating Charges, any Services that start on a day other than the first day of a month or are

CONFIDENTIAL ny-1040886

1.6 Attachments. Attached to this **Schedule C** are the following Attachments:

**Schedule C-1a** – Pricing for Parent Services

**Schedule C-2a** – Pricing for Reverse Services

**Schedule C-1b** –Billing Method by Service – Parent Services

<u>Schedule C-2b</u> –Billing Method by Service – Reverse Services

#### 2. COMMENCEMENT OF CHARGES.

The Total Shared Services Charges will begin on the Effective Date and are set forth in **Schedule C-1a** and **Schedule C-2a**. Charges for Additional Services, Customized Services, and Termination Assistance Services shall begin on the date such services begin to be performed by Supplier.

#### 3. PRICING METHODOLOGY.

- 3.1 Total <u>Monthly Shared Services Charges</u>. All Total Monthly Shared Services Charges will only be adjusted in accordance with the Price Adjustment Process.
- 3.2 <u>Additional Services</u>. Charges for Additional Services will be set forth in the applicable Supplement for such Services. To the extent the Additional Services are ongoing, the Charge may also or in the alternative be added as a separate item in <u>Schedule C-1a</u> or <u>Schedule C-2a</u>, as applicable and, if so included as a separate item, will be subject to the Price Adjustment Process just as any other Total Shared Services Charges would be.
- 3.3 <u>Customized Services</u>. Charges for Customized Services will be set forth in the applicable Supplement for such Services. To the extent the Customized Services are ongoing, the Charge may also or in the alternative be added as a separate item in <u>Schedule C-1a</u> or <u>Schedule C-2a</u>, as applicable and, if so included as a separate item, will be subject to the Price Adjustment Process just as any other Total Shared Services Charges would be.
- 3.4 This section is intentionally left blank.
- 3.5 <u>Price Adjustment Process</u>. The Price Adjustment Process described in this <u>Section 3.5</u> will be followed by the Parties upon either Party's written request in the event of a Price Adjustment Event. The Price Adjustment Process for Business Change Events is subject to any approval right or termination right that Supplier may have with respect to such change in Services pursuant to <u>Section 3</u> or <u>14</u> of the Agreement.
  - (a) The Price Adjustment Process will be initiated immediately following a Price Adjustment Event and the resulting change to the Charges will take

CONFIDENTIAL Schedule C Page 4 ny-1040886 effect as soon as practical and will be retroactive to the point in time that the Price Adjustment Event occurs (or, in the case of adjustments under **Section 3.5(c)** below, retroactive to the point in time that the New Cost Allocation is determined).

- (b) Unless the Price Adjustment Process is initiated for a Business Change Event, during the Price Adjustment Process the Parties will mutually determine in good faith the cost impact of the Price Adjustment Event through the methodologies set forth in the exhibits to Schedule C including all of the sub-attachments. When a Service is being terminated by a Party, the starting assumption is that Recipient will no longer be charged for that Service; provided, however, that Supplier may be entitled to reimbursement from Recipient of all costs previously paid by Recipient for such Service that Supplier is not able to eliminate, for example, costs for physical assets which cannot be reduced (i.e. leasing expenses or rental expenses which cannot be eliminated), third party costs (i.e. minimum revenue commitments which are required under a third party agreement) or any other non-internal costs ("Stranded Costs") which the Parties using appropriate due diligence and commercially reasonable efforts can not eliminate. If, after using such efforts the Parties are unable to eliminate 100% of any Stranded Costs, the Parties will meet and discuss the remaining Stranded Costs to be reimbursed by Recipient to Supplier.
- (c) The Total Shared Services Charges set forth in Schedule C-1a and Schedule C-2a were determined by allocating to Recipient a portion of Supplier's underlying costs of performing the applicable Service (such cost allocation, the "Initial Cost Allocations"). In the event the Price Adjustment Process is initiated solely due to a Business Change Event and does not involve a termination of the applicable Service, then during the Price Adjustment Process, Supplier will (i) use the cost allocation methodology that was used to determine the Initial Cost Allocation for such Service, to determine a new cost allocation reflecting Recipient's then-current level of consumption of such Service, taking into account material increases or decreases (if any) in Supplier's overall costs that are a direct result of the change in Recipient's level of consumption of such Service (a "New Cost Allocation"), and (ii) if the New Cost Allocation differs from the Initial Cost Allocation for such Service, adjust the Total Shared Services Charge for such Service accordingly. otherwise agreed by the Parties, the cost allocation determination described above shall only be performed for the area impacted by the Business Change Event and such cost allocation determination per Functional Service Area shall not be performed more than once per month. At Recipient's request, Supplier will provide Recipient with all supporting calculations of the effects of such Business Change Event on Supplier's costs of performing the Services.
- 3.6 Charges for Extensions of Term. The pricing in this **Schedule C** as of the

Effective Date includes only the pricing for the Services during the Initial Term. Any extensions of the Term are subject to the Parties reaching agreement of the pricing during such extension period.

3.7 <u>Termination Assistance Services</u>. If requested by Recipient, Supplier will provide Termination Assistance Services in accordance with <u>Section 14.5</u> of the Agreement. The price of Termination Assistance Services shall be documented in a Supplement to <u>Schedule A</u>. In addition to any Charges otherwise provided for in the Agreement or <u>Schedule C</u>, Recipient will reimburse Supplier for all incremental additional resource and other costs and expenses required or incurred by Supplier to provide Termination Assistance Services.

Schedule C-1
Pricing for Parent Services (1)

Parent Services Pricing		Monthly Total Sha	red Services Charge	(US\$)
	Total Base			Total Shared
<b>Functional Service Area</b>	Costs (2)	Third Party Costs	IT Projects	Services
Finance / Tax	\$ 1,340,595	\$ -	\$ 26,293	\$ 1,366,888
Audit	12,845	67,168	-	80,013
Treasury	820,790	-	72,768	893,558
Risk	799,225	29,000	-	828,225
Insurance Premium	-	1,267,250	-	1,267,250
Compliance	403,948	-	14,137	418,086
Consent Order	189,720	150,000	-	339,720
Loan Review	33,577	-	-	33,577
HR	367,188	468,747	23,707	859,642
Legal	127,175	-	36,583	163,758
Communications / IR	296,414	-	9,231	305,645
Capital Markets	464,577	-	36,464	501,041
Marketing	253,271	-	-	253,271
Facilities	115,956	119,436	-	235,392
Supply Chain	608,376	-	38,200	646,576
Gov't Relations	31,651	3,000	-	34,651
ITG	1,879,177	-	141,521	2,020,698
<b>Total Monthly Charge</b>	\$ 7,744,484	\$ 2,104,601	\$ 398,904	\$ 10,247,989

<sup>&</sup>lt;sup>(1)</sup>Approximate costs of services - refer to 3 supporting Exhibits. C-1a AFI Shared Services Workstream, C-1b Monthly Billing detail and Exhibits in C-1c for Pricing calculations

 $<sup>^{(2)}</sup>$  Includes employee costs, IT services, IT platform costs, and 13.8% of Indirect Support

**AFI Shared Services Workstream** ResCap, LLC - Pricing Summary Schedule C-1a

			BASE COSTS	(0		BILLED	BILLED AS INCURRED	RED	As of May 10, 2012	202
AFI Shared Service	Employee (1)	e IT Costs	Platform Costs	Indirect Support Costs	Total Base Costs	Third Party Costs I	T Projects	Total As Incurred	Total Shared Services	Total Shared O Services A Monthly
	[A]	[8]	[0]	[D] = 13.8% of [A]+[B]+[C]	[E] = [A]+[B]+[C]+[D]	E	[6]	[H] = [F]+[G]	[I] = [E]+[H]	Doc
Finance / Tax	\$	5.9 \$ 5.4	4 \$ 2.9	\$ 2.0	\$ 16.1	ı	0.3	\$ 0.3	\$ 16.4	\$4,4.1
Audit	0			0.0		0.8	1			
Treasury	3	3.9 4.3	3 0.5	1.2	9.8	•	0.0	0.9	10.7	<b>L</b> 6:0
Risk	2	2.8 4.7	7 0.9	1.2	9.6	0.3	1	0.3	6.6	ile 80
Insurance Premium	'	1	•	1	1	15.2	,	15.2	15.2	1.3 <b>5</b>
Compliance	2	2.3 1.5	5 0.5	9.0	4.8	1	0.2	0.2	5.0	0.40
Consent Order	2	2.0 -	1	0.3	2.3	1.8	1	1.8	4.1	5/ <b>1</b> E:0
Loan Review	0			0.0	0.4	1	,	1	0.4	<b>4/</b> 0:0
HR	2	2.9 0.9	0.0	0.5	4.4	5.6	0.3	5.9	10.3	12 60
Legal	0	0.3 0.9	9 0.1	0.2	1.5	1	0.4	0.4	2.0	P69
Communications	1	1.4 1.6	6 0.1	0.4	3.6	1	0.1	0.1	3.7	E <sub>2</sub> 8
Capital Markets	3	3.4 1.4	4 0.1	0.7	5.6		0.4	0.4	0.9	nt G
Marketing	2	2.7	•	0.4	3.0		,	1	3.0	er off
Facilities	1	1.0 0.2	2 0.0	0.2	1.4	1.4	,	1.4	2.8	ed 12
Supply Chain	2	5.6 0.7	7 0.1	0.0	7.3		0.5	0.5	7.8	0 26
Gov't Relations	0	0.3	ı	0.0	0.4	0.0	1	0.0	0.4	<b>5/</b>
ITG <sup>(3)</sup>		17.5	5 2.4	2.7	22.6	ı	1.7	1.7	24.2	1 <b>4</b> / 70:
										12
<b>Total Shared Services</b>	\$ 34.9	1.9 \$ 39.1	1 \$ 7.7	\$ 11.3	\$ 92.9	\$ 25.2	\$ 4.8	\$ 30.0	\$ 123.0	\$ 10.27

Filed 05/14/12 Entered 05/14/12 14:50:46

 $^{(2)}$  Direct / Third Party costs represent pass through costs based on diligence and negotiations

	10.8	8.4	2.0	2.9	24.2
					❖
	0.7	0.3		0.7	1.7
	0.7	0.3		0.7	1.7 \$
	1	ı			\$ -
					\$
	10.1	8.2	2.0	2.2	22.6
	1.2	1.0	0.2	0.3	2.7 \$
	0.8	4.1	0.1		2.4 \$
	8.1	5.8	1.7	2.0	17.5 \$
					\$
(3) ITG Breakdown	Tech Infrastructure	Security	Architecture	Tech & Ops	Total ITG

Main Document

## 12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 81 of 126

Schedule C-1b Pricing for Parent Services

rent Services Prici	ng				
Function	SOW Link	Notes	Service	Monthly Total Shared Service Charge (US\$)	Monthly Billing Methodology
Finance/Tax	1a4	Headcount	Finance Shared Services - Payroll	\$33,260	Monthly Variable
	1a4	Purchase Orders	Finance Shared Services - P2P	\$64,648	Monthly Variable
	1a4	Reconciliations	Finance Shared Services - Accounting	\$17,379	Monthly Variable
	1a4	Expenses Accounts	Finance Shared Services - T&E	\$17,895	Monthly Variable
	1a5	Time Study	Accounting Policy Team	\$48,854	Monthly Fixed
	1a5	Time Study	Benefits Accounting	\$5,949	Monthly Fixed
	1a5	Time Study	Accounting Leadership and Oversight - Accounting and Policy	\$17,753	Monthly Fixed
	1a8	Time Study	Tax - Direct Team Costs	\$212,846	Monthly Fixed
	1a8	Time Study	Tax Indirect Services	\$56,717	Monthly Fixed
	1a8	Time Study	Tax Leadership	\$48,543	Monthly Fixed Monthly Variable - Bi
	NA	pass through	External Audit Fees Financial Controls - Consolidated and Global Functions	\$0	Incurred
	1a5	Time Study	(Grennan Team)  ITG Finance - FP&A Support for ResCap and Global	\$0	Monthly Fixed
	1a4	Time Study	Functions (Kubitz Team)	\$32,448	Monthly Fixed
	1r9	OPEX share less SAP exclusions	Finance - Sustain	\$507,713	Monthly Variable
	1r9	OPEX share less SAP exclusions	Finance - IT Platform	\$276,591	Monthly Fixed
	1r9	OPEX share of selected projects	Finance - IT Projects	\$26,293	Monthly Variable
			Finance Total	\$1,366,888	
Audit		Time Study			
	1g		Direct Team Related to ResCap  Training for the Consent Order - signed with PwC -	\$0	Monthly Fixed
	1g	Time Study	training for all audit staff - already expensed - design classes; customize the services	\$0	Monthly Variable - Bi Incurred
	1g	pass through	Co-Sourcing for the Consent Order - Subject Matter experts for demonstrating enhanced coverage - planning activities make sure we cover the right audits	\$66,667	Monthly Variable - Bi Incurred
	1g	pass through	Indirect costs for additional Audit work from Fisette Team - Help Mgmt reporting, audit deck etc	\$9,046	Monthly Variable
	1g	pass through	TeamMate - Audit software to hold workpapers	\$502	Monthly Variable - Bi Incurred
	1r9	OPEX share	Audit - IT Sustain	\$3,799	Monthly Variable
	119	OF LA STIGITE	Audit Total	\$80,013	Widitilly Valiable
Treasury	1w1	Time Study			Monthly Fixed
,		,	ST Liquidity and Funding Planning	\$16,933	Monthly Fixed
	1w1	Time Study	LT Liquidity	\$11,288	Monthly Fixed
	1w1	Time Study	Risk Controls	\$3,327	Monthly Fixed
	1w1	Time Study	Liquidity Executive	\$8,877	Monthly Fixed
	1w1	Time Study	OH Allocation - Facilities and Travel	\$4,209	Monthly Fixed
	1w1	Time Study	Leadership Direct	\$7,864	Monthly Fixed
	1w2	Time Study	Global Funding Team	\$97,558	Monthly Fixed
	1w2	Time Study	OH Allocation - Facilities and Travel	\$8,862	Monthly Fixed
	1w2	Time Study	Leadership Allocation - Direct	\$14,202	Monthly Fixed
	1w7	Time Study	Direct Costing	\$147,214	Monthly Fixed
		Time Study	Facilities and Travel	\$16,173	Monthly Fixed
	1w7	Title Study			
	1w7 1w7	Time Study	Leadership Allocation	\$30,789	Monthly Fixed
		1	Leadership Allocation Treasury - ITG Sustain	\$30,789 \$405,076	,
	1w7	Time Study			
	1w7 1r9 1r9	Time Study EOP Assets EOP Assets OPEX share of selected	Treasury - ITG Sustain Trasury - ITG Platform	\$405,076 \$48,418	Monthly Variable  Monthly Fixed
	1w7 1r9	Time Study EOP Assets EOP Assets	Treasury - ITG Sustain	\$405,076	Monthly Variable
	1w7 1r9 1r9	Time Study EOP Assets EOP Assets OPEX share of selected	Treasury - ITG Sustain Trasury - ITG Platform Treasury - IT Projects	\$405,076 \$48,418 \$72,768	Monthly Variable Monthly Fixed
Risk (Includes Model Governance)	1w7 1r9 1r9	Time Study EOP Assets EOP Assets OPEX share of selected	Treasury - ITG Sustain Trasury - ITG Platform Treasury - IT Projects	\$405,076 \$48,418 \$72,768	Monthly Variable Monthly Fixed
	1w7 1r9 1r9 1r9	Time Study EOP Assets EOP Assets OPEX share of selected projects	Treasury - ITG Sustain Trasury - ITG Platform Treasury - IT Projects Treasury Total	\$405,076 \$48,418 \$72,768 \$893,558	Monthly Variable Monthly Fixed Monthly Variable
	1w7 1r9 1r9 1r9	Time Study EOP Assets EOP Assets OPEX share of selected projects Time Study Time Study	Treasury - ITG Sustain Trasury - ITG Platform Treasury - IT Projects Treasury Total  Leadership Administration Enterprise Risk (includes Analytics from Seth S. Group)	\$405,076 \$48,418 \$72,768 \$893,558 \$101,341 \$73,041	Monthly Variable Monthly Variable  Monthly Variable  Monthly Fixed  Monthly Fixed
	1w7 1r9 1r9 1r9	Time Study EOP Assets EOP Assets OPEX share of selected projects Time Study	Treasury - ITG Sustain Trasury - ITG Platform Treasury - IT Projects Treasury Total  Leadership Administration	\$405,076 \$48,418 \$72,768 \$893,558 \$101,341	Monthly Variable Monthly Fixed  Monthly Variable  Monthly Fixed

5/12/2012

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 82 of 126

			Pg 82 of 126		
Function	SOW Link	Notes	Service	Monthly Total Shared Service Charge (US\$)	Monthly Billing Methodology
		Reserved seats at	Business Continuity Planning - Sungard Expenses 3rd		Monthly Variable - Bill as
	1j 1j	contigency site Headcount	party  Corporate Insurance Promiums	\$29,000	Incurred
			Corporate Insurance Premiums	\$1,267,250	Monthly Fixed
	1r9	Risk Time Study	Risk - IT Sustain	\$448,279	Monthly Variable
	1r9	Risk Time Study	Risk - Platform	\$89,453	Monthly Fixed
			Risk Total	\$2,095,475	
5 Compliance (Includes Consent Order Costs)	10	Time Study	Global Security	\$42,747	Monthly Fixed
	1n	Time Study	Privacy Services	\$0	Monthly Fixed
	1m	Time Study	Leadership Administration	\$45,046	Monthly Fixed
	1m	Time Study	AML	\$47,424	Monthly Fixed
	1m	Time Study	Program Management	\$56,744	Monthly Fixed
	1m	Time Study	Global Functions	\$22,916	Monthly Fixed
	1m	100% to ResCap	Consent Order Direct Costs - 3rd Party	\$150,000	Monthly Variable - Bill as Incurred
	1m	100% to ResCap			Monthly Fixed
		· ·	Consent Order Staffing	\$189,720	·
	1r9	OPEX share	Compliance - IT Sustain	\$143,865	Monthly Variable
	1r9	OPEX share OPEX share of selected	Compliance - Platform	\$45,206	Monthly Fixed
	1r9	projects	Compliance - IT Projects	\$14,137	Monthly Variable
			Compliance Total	\$757,806	
6					
Loan Review 7	1t	% of total loans reviewed	Loan Reviews	\$33,577	Monthly Fixed
Human Resources	1q	Time Study	Benefits Administration	\$32,410	Monthly Fixed
	1q	Time study	Compensation	\$67,169	Monthly Fixed
	1q	ResCap specific BPO costs	Employee Relations	\$15,005	Monthly Fixed
	1q	Time Study	HR Business Partner Support	\$91,370	Monthly Fixed
	1q	Time Study	Operations - Base Costs	\$39,485	Monthly Fixed
	1q	Time Study	Operations - Third party	\$440,167	Monthly Fixed
	1q	Time Study	Staffing - Base Costs	\$31,247	Monthly Fixed
	1q	Time Study Higher of 11.75% of	Staffing - Third Party	\$28,580	Monthly Fixed
	1q	ResCap base pay or claim experience	Medical Dental, Life Insurance, Disability Charges	\$0	Monthly Variable
	1q	actual contributions	401K Employer Contributions	\$0	Monthly Variable
	1r9	headcount share	HR - IT Sustain	\$87,360	Monthly Variable
	1r9	headcount share	HR - IT Platform	\$3,142	Monthly Variable
	1r9	headcount share	HR - IT Projects	\$23,707	Monthly Variable
			HR - Total	\$859,642	
8 Legal	1s	Time Study	HR and Employment	\$10,022	Monthly Fixed
	1s	Time Study	Procurement	\$2,181	Monthly Fixed
	1s	Time Study	Law Department Management Support	\$5,527	Monthly Fixed
	1s	Time Study	Proportion of Attorneys & Paralegals	\$14,798	Monthly Fixed
	1r9	OPEX share	Legal - IT Sustain	\$82,841	Monthly Variable
	1r9	OPEX share	Legal - IT Platform	\$11,805	Monthly Fixed
		OPEX share of selected			,, . mea
	1r9	projects	Legal - IT Projects	\$36,583	Monthly Variable
			Legal Total	\$163,758	
9 Communications and Investor Relations	SOW*	Time Study	Employee Communications	\$2,016	Monthly Fixed
esco. Neidelons	SOW*	Time Study	Community Relations	\$0	Monthly as Incurred
	sow*	Time Study	Digital Communications	\$10,149	Monthly Fixed
	sow*	headcount share	Investor Relations	\$124,433	NA

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 83 of 126

	1		Pg 83 of 126		
Function	SOW Link	Notes	Service	Monthly Total Shared Service Charge (US\$)	Monthly Billing Methodology
	1r9	headcount share	Communications - IT Sustain	\$150,324	Monthly Variable
	1r9	headcount share	Communications - IT Platform	\$9,493	Monthly Fixed
	1r9	headcount share	Communications - IT Projects	\$9,231	Monthly Variable
			Communications Total	\$305,645	
10 Conital Markets and IM	SOW*	Time Study , Dischards	lauratus ant Managaran	Ć205 570	Manual In Chand
Capital Markets and IM	SOW*	Time Study + Blackrock  1 analyst	Investment Management  MSR Analyst	\$305,570 \$14,229	Monthly Fixed  Monthly Fixed
	1r9	OPEX share	Capital Markets - IT Sustain	\$136,423	Monthly Variable
	1r9	OPEX share	Capital Markets - IT Platform	\$8,355	Monthly Fixed
	1.3	OPEX share of selected	Capital Markets 11 - Iatoliii	φο,333	moneny rinea
	1r9	projects	Capital Markets - IT Projects	\$36,464	Monthly Variable
			Capital Markets - Total	\$501,041	
11					
Marketing	<b>1</b> i	Time Study	Strategic Marketing Planning & Strategy Development	\$53,206	Monthly Fixed
	1i	Time Study	Creative Asset Design Production & Mgmt	\$46,856	Monthly Fixed
	1i	Time Study	Marketing performance evaluation & optimization activities	\$13,686	Monthly Fixed
	1	Time Study	Management of day-to-day marketing operational	Ģ13,000	Worlding Fixed
	<b>1</b> i	Time Study	responsibilities	\$12,839	Monthly Fixed
	1i	Time Study	Facilities, Office Communications	\$8,298	Monthly Fixed
	1i	Time Study	Travel	\$3,781	Monthly Fixed
	1i	Time Study	eCommerce Product Mgmt Services	\$50,190	Monthly Fixed
	1i	Time Study	Customer Experience Design & Development	\$49,137	Monthly Fixed
	<b>1</b> i	Time Study	Website Optimization	\$3,676	Monthly Fixed
	1i	Time Study	Facilities, Office Communications	\$7,969	Monthly Fixed
	1i	Time Study	Travel	\$3,632	Monthly Fixed
			Marketing Total	\$253,271	
12 Facilities	1ka	Square Footage	Facility Services	\$90,479	Monthly Fixed
	1ka	Square Footage	3rd party JLL Services	\$119,436	Monthly Fixed
	IKG	Square rootage	Current rent payment process and allocation to	\$119,430	Worlding Frized
			business/function based on occupied space will stay in	4.0	
	1ka	Square Footage	effect	\$0	Settled Monthly
	1r9	Square Footage	Facilities - IT Sustain	\$22,293	Monthly Variable
	1r9	Square Footage	Facilities - IT Platform	\$3,184	Monthly Fixed
	113	Square rootage			Monthly Fixed
			Facilities Total	\$235,392	
13 Supply Chain	1kb	Time Study			
	1kb	NA	Sourcing, Monitoring and Off-boarding of Vendors	\$535,930	Monthly Fixed
	IND		Pricing covers the Maintenance of the Risk Partner Relationships and all work associated with meeting the		
		Manadan Cu	Consent Order requirements and other services	\$0	NA
	1r9	Vendor Spend	Supply Chain - IT Sustain	\$63,513	Monthly Variable
	1r9	Vendor Spend	Supply Chain - IT Platform	\$8,933	Monthly Fixed
	1r9	OPEX share of selected projects	Supply Chain - IT Projects	\$38,200	Monthly Variable
	1.5	F. 5,000	Supply Chain - Total	\$646,576	
14 Executive Leadership					
LACCULIVE ECUACISHIP				\$0	NA
15 Government Relations	1x		Government Relations	\$34,651	Monthly Fixed
16 ITG			ITG Sustain		
	All IT SOWs excl				
	1r1, 1r9, SOW for IT Resources, IT				
		OPEX share less non-			
		Mortgage NIDD exclusions	Corporate TI (includes NIDD)	\$844,182	Monthly Variable
		OPEX share less non-			
	1r1	Mortgage Privacy and Project Sustain	Security	\$679,516	Monthly Variable
	1 111	i roject sustain	Security	9075,510	wionally variable

5

5/12/2012

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 84 of 126

Function	SOW Link	Notes	Service	Monthly Total Shared Service Charge (US\$)	Monthly Billing Methodology
	IT	OPEX share less non			
		Mortgage application and Project Sustain	Architecture	\$169,203	Monthly Variable
	Cross Functional	OPEX Share	Technology Operations Group	\$186,276	Monthly Variable
			IT Projects for Mortgage		
		OPEX share for selected			
	1r9	projects OPEX share for selected	TI Global Functions NADD	\$47,199	Monthly Variable
	1r9	projects	TI Corporate NADD	\$12,664	Monthly Variable
	1r9	OPEX share for selected projects	Security	\$23,850	Monthly Variable
	1r9	Project spend	Technology Operations Group	\$57,809	Monthly Variable
			Projects for Mortgage	\$0	Monthly Variable
			ITG - Total	\$2,020,698	

Total Monthly Shared Service Charge \$10,247,989

Annual Total Shared Service Charge \$122,975,884

5/12/2012

<sup>\*</sup> SOW is not numbered

## 12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 85 of 126

Schedule C-2 Pricing for ResCap Services (1)

ResCap to AFI				Mon	thly Total Shared	Services CI	harges (US	\$)	
		Total I	Base Costs						Total Shared
Functional Service Area	Statement of Work		(2)	Th	ird Party Costs	IT Pro	jects		Services
Human Resources	HR Support Services (AFI)	\$	74,326	\$	-	TB	BD	\$	74,326
Legal	Legal Services (AFI)		21,481		-	TB	BD		21,481
ITG	IT Resource Services		1,015,068		-	TB	BD		1,015,068
Compliance	Line of Business Compliance Services		79,088		-	TB	BD		79,088
Risk	Risk Services - Reporting and Special Assets		23,342		-	TB	BD		23,342
Accounting	ResCap Accounting Services		163,437		-	TB	BD		163,437
Master Servicing	Master Servicing for Ally Auto - Bond Modeling		15,436		15,000	TB	BD		30,436
Master Servicing	Master Servicing for Ally Auto - Bond Administration		26,274		-	ТВ	BD		26,274
Subtotal		\$	1,418,453	\$	15,000	\$	-	\$	1,433,453

ResCap to AFI for Bank			<b>Monthly Total Shared</b>	Services Charges (U	S \$)
		Total Base Costs			Total Shared
Functional Service Area	Statement of Work	(2)	Third Party Costs	IT Projects	Services
ITG	Application Support	\$ 444,495	\$ 222,122	TBD	\$ 666,617
Capital Markets	Capital Markets	428,678	-	TBD	428,678
Human Resources	HR Support Services (Bank)	32,388	-	TBD	32,388
Risk	Risk Management and Data Collection	501,014	141,858	TBD	642,872
Legal	Legal Services (Bank)	20,317	-	TBD	20,317
Accounting	Accounting	78,159	-	TBD	78,159
Finance	iviongaye řinanciai rianning anu Analysis	101,984	-	TBD	101,984
Capital Markets	Records Services	49,829	-	TBD	49,829
Business Risk and Controls and Business Excellence	Business Risk and Controls and Business Excellence	105,454	8,381	TBD	113,835
Accounting	Capital Markets Accounting	50,377	-	TBD	50,377
Consumer Lending	Consumer Lending Services	817,156	-	TBD	817,156
Risk	Client Repurchase Management	6,638	-	TBD	6,638
Subtotal		\$ 2,636,488	\$ 372,361	\$ -	\$ 3,008,850
Subtotal		\$ 4,054,941	\$ 387,361	\$ -	\$ 4,442,302

 $<sup>^{(1)}</sup>$  Approximate costs of services  $\,$  - refer to supporting exhibts C2-a and C2-b  $\,$ 

<sup>(2)</sup> Includes Employee Costs and 2.9% indirect support

	Services (1)
e C-2a	or ResCap
Schedul	Pricing f

ResCap to AFI - Functional Area   Statement of Work   Wo			mployee (1) IT Costs	_	Platorm Costs	Indirect Support Costs	Total Base Costs	Third Party Costs	IT Projects	Total As Incurred	Total Shared Services
HR Support Services (AFI)	ResCap to AFI - Functional Area	Statement of Work	(A)	(B)			(A) = (A) + (B) + (C) + (D)	(F)	(9)	(H) = (F) + (G)	(H) + (H) = (I)
He Support Services (AF)											
HR Support Services (AF)   2866 926	ResCap to AFI										
Transference Services (API)	Human Resources	HR Support Services (AFI)	866,926	,		24,991	891,918	•	٠		891,918
Tresource Services   11839,510	Legal	Legal Services (AFI)	250,555			7,223	257,777				257,777
Park Services   922.464   26.592   949,057   1.86,230   1.96,1243   1.96,226   1.96,1243   1.96,226   1.96,1243   1.96,236   1.96,1243   1.96,236   1.96,1243   1.96,236   1.96,1243   1.96,236   1.96,1243   1.96,236   1.96,1243   1.96,236   1.96,1243   1.96,236   1.96,1243   1.96,236   1.96,1243   1.96,236   1.96,1243   1.96,236   1.96,1243   1.96,236   1.96,1243   1.96,236   1.96,	ITG	IT Resource Services	11,839,510			341,302	12,180,812				12,180,812
Risk Services - Reporting and Special Assets   1,905,296   1,0244   1,91,12	Compliance	Line of Business Compliance Services	922,464	,		26,592	949,057		•		949,057
ResCap Accounting Services   1.906.290	Risk	Risk Services - Reporting and Special Assets	272,258	,		7,848	280,107				280,107
Master Services (Bank)	Accounting	ResCap Accounting Services	1,906,290	,		54,953	1,961,243				1,961,243
16,339,455   -	Master Servicing	Master Servicing for Ally Auto - Bond Modeling	174,996			10,234	185,230	180,000	•	180,000	365,230
Bank         Application Support         5,109,805         -         -         224,141         5,333,945         2,665,462         -         2,665,4	Master Servicing	A desirable to the second of t	306,456			8,834	315,290				315,290
Bank         Application Support         5,109,805         - 224,141         5,333,945         2,665,462         - 2,665,462           Capital Markets         Capital Markets         5,000,000         - 144,137         5,144,137         - 2,665,462         - 2,665,462           Risk Management and Data Collection         5,766,014         - 243,804         - 1,702,299         - 1,702,299           Risk Management and Data Collection         5,786,014         - 243,804         - 243,804         - 1,702,299         - 1,702,299           Andrage Financial Planning and Analysis         1,185         - 34,291         - 34,291         - 1,702,399         - 1,702,299           Records Services         1,185         - 34,291         - 34,291         - 1,223,804         - 1,702,299         - 1,702,299           Records Services         1,185         - 34,291         - 34,291         - 1,223,804         - 1,203,804         - 1,203,804         - 1,205,44         - 100,575         - 100,575           Records Services         1,227,169         - 1,227,169         - 1,265,44         100,575         - 10,675         - 10,675           Capital Markets Accounting         587,589         - 1,011,684         31,637,889         - 1,468,336         - 1,468,336         - 1,468,336         - 1,468,336         - 1,468,336<	Total AFI		16,539,455	•		481,979	17,021,434	180,000		180,000	17,201,434
Application Support 6,109,805 - 224,141 5,333,945 2,665,462 - 2,665,462 Capital Markets Capital Markets S,000,000 - 144,137 5,144,138,555 - 1,265,444 1,702,299 1 1,702,299 1 1,702,299 1 1,702,299 1 1,702,299 1 1,702,299 1 1,702,299 1 1,702,299 1 1,702,299 1 1,702,299 1 1,702,299 1 1,702,299 1 1,223,804 1 1,	ResCap to AFI for Ally Bank										
Harden	ITG	Application Support	5,109,805	,		224,141	5,333,945	2,665,462		2,665,462	7,999,408
HR Support Services (Bank)	Capital Markets	Capital Markets	5,000,000	,	,	144,137	5,144,137		٠	. '	5,144,137
Risk Management and Data Collection         5,796,011         -         216,157         6,012,168         1,702,299         -         1,702,299           Legal Services (Bank)         236,973         -         -         26,280         937,908         -         -         -           Accounting         Mortgage Financial Planning and Analysis         1,189,513         -         -         26,280         937,908         - <td>Human Resources</td> <td>HR Support Services (Bank)</td> <td>377,765</td> <td>,</td> <td>,</td> <td>10,890</td> <td>388,655</td> <td>•</td> <td>•</td> <td></td> <td>388,655</td>	Human Resources	HR Support Services (Bank)	377,765	,	,	10,890	388,655	•	•		388,655
Legal Services (Bank)	Risk	Risk Management and Data Collection	5,796,011	,		216,157	6,012,168	1,702,299	•	1,702,299	7,714,467
Accounting Accounting Mortgage Financial Planning and Analysis Records Services  Records Services  Business Risk and Controls and Business  Excellence Capital Markets Accounting Consumer Lending Services  Consumer Lending Services  A7,419  A7,419  A7,416,630  A4,686,336  A4,686,336  A4,686,336  A4,686,336  A4,686,336  A4,689,336  A4,689,336  A4,648,336	Legal	Legal Services (Bank)	236,973	,		6,831	243,804	•			243,804
Mortgage Financial Planning and Analysis   1,189,513   34,291   1,223,804	Accounting	Accounting	911,628			26,280	937,908				937,908
Records Services         581,189         -         16,754         597,943         -	Finance	Mortgage Financial Planning and Analysis	1,189,513	,		34,291	1,223,804	•			1,223,804
rols and Business         Husiness Risk and Controls and Business         1,227,169         -         38,275         1,265,444         100,575         -         100,575         -         100,575         -         100,575         -         100,575         -         100,575         -         100,575         -         100,575         -         100,575         -         100,575         -         100,575         -         100,575         -	Capital Markets	Records Services	581,189			16,754	597,943				597,943
Excellence       587,589       -       -       16,939       604,528       -       -       100,575         Capital Markets Accounting       9,531,114       -       -       274,757       9,805,872       -       -       -         Cilient Repurchase Management       77,419       -       -       1,011,684       31,637,889       -       -       4,468,336       -       -       4,468,336       -       -       4,648,336       -       -       4,648,336       -       -       4,648,336       -       -       4,648,336       -       -       4,648,336       -       -       4,648,336       -       -       4,648,336       -       -       4,648,336       -       -       4,648,336       -	Business Risk and Controls and Business	Business Risk and Controls and Business	1,227,169	,		38,275	1,265,444	100,575			
Capital Markets Accounting         587,589         -         -         16,939         604,528         - <td>Excellence</td> <td>Excellence</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>100,575</td> <td>1,366,019</td>	Excellence	Excellence								100,575	1,366,019
Consumer Lending Services 9,531,114 - 274,757 9,805,872	Accounting	Capital Markets Accounting	587,589	,		16,939	604,528	•			604,528
Client Repurchase Management         77,419         -         2,232         79,651         -	Consumer Lending	Consumer Lending Services	9,531,114	,		274,757	9,805,872	•			9,805,872
30,626,175     -     -     1,011,684     31,637,859     4,468,336     -     4,468,336       47,165,630     -     -     1,493,663     48,659,293     4,648,336     -     4,648,336	Risk	Client Repurchase Management	77,419			2,232	79,651				79,651
47,165,630 - 1,493,663 48,659,293 4,648,336 - 4,648,336	Total AFI for the Bank		30,626,175	•	·	1,011,684	31,637,859	4,468,336	•	4,468,336	36,106,196
oppingt.	Total Bank and AEI		47.165.630			1.493.663	48.659.293	4.648.336		4 648 336	53 307 629
	I Otal Dalin allu Ari		200620		1	,	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	2206		20,01	220,100,00

(1) Employee Costs include Compensation and Benefits plus costs for T&E, Facilities, Office and Communications, Training and other employee related expenses (2) Direct third party costs represent pass-through costs based on diligence and negotiations

## 12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 87 of 126

Schedule C-2b Pricing for ResCap Services

		ResCap to AFI Pricing		
Function	Statement of Work	Service	Monthly Total Shared Service Charge (US\$)	Monthly Billing Methodology
1. Human Resources	HR Support Services (AFI)	All Services	\$74,326	Monthly Fixed
2   2	Lord Consisse (AEI)	Internal Counsel	\$21,481	Monthly Fixed
2. Legal	Legal Services (AFI)	External Counsel	TBD	Monthly Variable As incurred
3. ITG	IT Descures Comisses	IT Sustain	\$1,015,068	Monthly Variable As incurred
3.110	IT Resource Services -	IT Projects	TBD	Monthly Variable As incurred
4. Compliance	Line of Business Compliance Services	All Services	\$79,088	Monthly Fixed
5. Risk	Risk Services - Reporting and Special Assets	All Services	\$23,342	Monthly Fixed
6. Accounting	ResCap Accounting Services	All Services	\$163,437	Monthly Fixed
7. Master Servicing	Master Servicing for Ally Auto - Bond Modeling	All Services	\$30,436	Monthly Fixed
8. Master Servicing	Master Servicing for Ally Auto - Bond Administration	All Services	\$26,274	Monthly Variable As incurred

	ResC	ap to AFI for the Bank Prici	ng	
Function	Statement of Work	Service	Monthly Total Shared Service Charge (US\$)	Monthly Billing Methodology
14a ITG	Application Support	Sustain	\$666,617	Monthly Variable based on departmental expenses
		Projects	TBD	Monthly Variable As incurred
4.4h Conital Manusata	Comital Manhata	All Services	\$428,678	Monthly Fixed
14b Capital Markets	Capital Markets	Document Custody 3rd Party	TBD	Monthly Variable As incurred
14c. Human Resouces	HR Support Services (Bank)	All Services	\$32,388	Monthly Fixed
14d Risk	Risk Management and Data Collection	All Services	\$642,872	Monthly Variable based on departmental expenses
Maland.	Legal Services (Bank)	Internal Counsel	\$20,317	Monthly Fixed
14e Legal		External Counsel	TBD	Monthly Variable As incurred
14f. Accounting	Accounting	All Services	\$78,159	Monthly Fixed
14g Finance	Mortgage Financial Planning and Analysis	All Services	\$101,984	Monthly Fixed
400 7114 1		All Services	\$49,829	Monthly Fixed
14i Capital Markets	Records Services	3rd Party Costs	TBD	Monthly Variable As incurred
14j Business Risk and Controls and Business Excellence	Business Risk and Controls and Business Excellence	All Services	\$113,835	Monthly Fixed
14L Accounting	Capital Markets Accounting	All Services	\$50,377	Monthly Fixed
14M Consumer Lending	Consumer Lending Services	All Services	\$817,156	Monthly Variable Based on Consumer Loans funded
14N Risk	Client Repurchase Management	All Services	\$6,638	Monthly Fixed Needs to be reviewd quarterly to establish a new ratio of recoveries processed

Total Monthly Shared Service Charge \$ 4,442,302
Total Annual Shared Service Charge \$ 53,307,630

### Schedule D-1

List of AFI and Supported Facilities

As kept by the Corporate Real Estate Office as of the Effective Date

#### Schedule D-2

List of ResCap and Supported Facilities

As kept by the Corporate Real Estate Office as of the Effective Date

### Schedule E

	Form of Supplement
	Supplement [X]  Supplement (this "Supplement") is made and entered into as of the day of
Dela	, 2012 (the "Supplement Effective Date") by and between Ally Financial Inc., a tware corporation ("AFI") and Residential Capital, LLC, a Delaware limited liability pany ("ResCap") (together with AFI, the "Parties").
1.	BACKGROUND
	This Supplement is entered into pursuant to the terms of the Shared Services Agreement between AFI and ResCap dated [•] (the "Agreement") and constitutes a Supplement under the Agreement. Capitalized terms used but not defined in this Supplement have the meanings assigned to those terms in the Agreement.
2.	SERVICES DESCRIPTION AND CHARGES
	Supplier will provide [ <i>Insert brief description of the services here.</i> ] services as [Additional Services]/[Customized Services][, which shall be described in more detail in, for the Charges set forth [therein]/[in <u>Schedule C</u> to the Agreement].
3.	CHANGES
	[Refer to the clauses in the Agreement relating to Services and incorporate here as appropriate and agreed as the clauses in the Agreement do not apply to the Additional Services or Customized Services being added pursuant to this Supplement.]
4.	TERM AND TERMINATION
	[Term and notice provisions to be inserted as appropriate.]
5.	[OTHER TERMS
	The Parties further agree:
	[Insert any other terms and conditions applicable to the Additional Services or Customized Services to be performed under this Supplement.]

### 6. MISCELLANEOUS

This Supplement is incorporated by reference into the Agreement. In the event of

## 12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 91 of 126

any conflict between the terms of this Supplement and the Agreement, the terms of this Supplement shall only prevail to the extent that this Supplement expressly states that it is intended to override a term of the Agreement.

SPACE BELOW INTENTIONALLY BLANK – SIGNATURE PAGE FOLLOWS

## 12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 92 of 126

**IN WITNESS WHEREOF**, the Parties have caused this Supplement to be executed by their respective duly authorized representatives as of the Supplement Effective Date.

Ally F	inancial Inc.
By:	
25.	Name:
	Title:
Reside	ential Capital, LLC
By:	
,	Name:
	Title:

### **EXHIBIT B**

# 12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 94 of 126 Summary of Statements of Work From AFI to ResCap

	Statement of Work (AFI to ResCap)	Functional Service Area	Scope of Services
1	Finance Shared Services (1a04)	Finance Tax	AFI provides various finance services to ResCap including Purchase-to-Pay (P2P) in various ResCap locations. These P2P services include use of the using various applications through AFI's vendor management system including purchase requisition, invoice reconciliation, create and update vendor master records, and processing invoices for payment to vendors. AFI also processes ResCap's payroll needs through its various locations, handles travel and expense management services, account reconciliation to the general ledger, fixed assets accounting by entering necessary journal entries and reclaimed property and escheatment processing through AFI's database.
2	Tax Services (1a08)	Finance Tax	AFI prepares and validates ResCap's tax forecast, provides tax planning advice to meet ResCap's needs as well as preparing supporting schedules necessary to substantiate tax journal entries. AFI also prepare account reconciliations on the for various aspects of ResCap's tax needs including current taxes, deferred taxes, non-income taxes, tax contingency amounts, and other comprehensive income (OCI) tax related accounts. These services also include preparing, reviewing and validating estimated tax calculations and, as appropriate, communicate adjustments to tax accruals for estimated tax payments and compliance with the provisions of the Tax Allocation Agreements. AFI prepares and files ResCap's federal income tax return(s) and schedules as well as any non-US income and non-income tax returns and schedules and state income/non-income tax returns and schedules.
3	General Accounting Services (1A5)	Finance Tax	AFI provides significant transaction and accounting advisory services to ResCap as well as GAAP monitoring and implementation. AFI supports the accounting standard setting process with various organizations on behalf of ResCap including the Financial Accounting Standards Board ("FASB") and supports ResCap in the implementation of changes in accounting standards.  On a monthly basis, AFI calculates accrual and compensation expense related to ResCap's compensation plans. On a monthly, it provides basis record journal entries and performs account reconciliations related to pension account balances. It provides support relating to adequacy of reserves, funds transfers pricing, variance analysis, Sarbanes-Oxley IT Testing and derivative accounting services.
4	Capital Markets (1C)	Capital Markets Services	AFI manages ResCap's Non-Core Mortgage Portfolio & Servicing Capital Group based on portfolio targets and balance sheet sensitivities as defined by ResCap's CEO, and other member of ResCap's management team, as directed. AFI executes ResCap hedge transactions for MSR & Treasury related exposures as well providing trading liquidity for Pipeline hedging as well as providing reports concerning the foregoing. It performs counterparty relationship management and negotiation of required trading documentation.

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 95 of 126

Statement of Work	Functional	Scope of Services
(AFI to ResCap)	Service Area	
	Audit	AFI internal audit services to ResCap are divided into two classes of services: The Class A
(1G)		Services include providing targeted audits of various aspects of ResCap's business lines,
		including mortgage originations, servicing and capital markets activities. Class B services
		includes utilizing AFI's standard risk assessment methodology and tools to conduct the
		annual risk assessment used as the basis for the ResCap audit plan. This also includes compiling quarterly reports for the ResCap Audit Committee relating to information on
		new, past due and re-opened audit issues, progress and results from completed audits, and
		any changes to AFI's audit plan that may impact audit coverage of ResCap, conducting
		AFI's quality assurance and improvement program (QAIP) for ResCap as well performing
		general accounting tasks such as ad-hoc analysis and research as needed to support the
		Services.
eCommerce (1H)	Communications	AFI manages online sales and service capabilities for ResCap by developing and
, ,		documenting business requirements to define the scope of specific initiatives, verifying
		that all internal reviews, governance and approval processes required by ResCap policies,
		procedures and governance structure are completed and documented prior to moving
		forward on any ResCap projects, processes and contracts, working with ResCap to
		provide that the entire online experience/capability is in full compliance with all regulatory
		guidelines and providing that ResCap's online security tools/methods remain up to date.
Global Brand & Product	Marketing	AFI supports ResCap management of ResCap's origination (home lending) product
Marketing (1I)		offering by providing strategic marketing planning & strategy development through its
		marketing platform development and evolution, messaging strategy, and communications
		planning activities. AFI helps ResCap develop its creative assets to meet agreed upon
		strategies and media/marketing plan commitments and performs marketing evaluation and
		creative optimization activities. AFI provides analytics of modeling associated with targeting. This management of day-to-day marketing operational responsibilities includes
		managing financial processes related to ResCap marketing expenditures with adherence to
		ResCap approved budgets as well as managing all agency resources and associated asset
		development status activities related to ResCap.
	(AFI to ResCap)  Internal Audit Services (1G)  eCommerce (1H)	(AFI to ResCap)  Internal Audit Services (1G)  Communications  Global Brand & Product  Marketing

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document
Pg 96 of 126

**Scope of Services Functional** Statement of Work (AFI to ResCap) **Service Area** Risk Management (1J) AFI as a global enterprise provides risk management services to ResCap. AFI's risk Risk 8 committee will advise on an as-needed basis on risk reporting provided through governance channels. This involves providing a policy framework (including the Global Policy on Policies) used to manage the policy lifecycle process for ResCap. It provides support on the new product approval process. AFI prepares risk analytics and reporting for ResCap's financial and regulatory reporting obligations as well as enterprise stress testing to evaluate the impact of severe yet plausible stress scenarios on risk, financial, capital and liquidity positions, but AFI enterprise stress testing builds on but does not replace autonomous stress tests conducted by ResCap covering specific exposures, portfolios or risks. In additions, AFI is responsible for providing ResCap with the operational risk management framework and business continuity. AFI also provides insurance coverage for ResCap such as Directors & Officers, Errors & Omissions, Fiduciary, Employment Practices Liability, Crime Bond, Property, Cyber Liability, Surety, Domestic & International Casualty – Workers' Compensation, General Liability and surety bonds. In conjunction with ResCap's market risk, AFI creates and monitors risk limits for MSR and performs economic stress testing and other analytical support for the MSR asset as needed, and provides analytical support relating to ResCaps HFI & HFS portfolios, whole loan sales and representations and warranties through use of AFI's models. **Facilities Services (1Ka)** AFI provides facilities services to ResCap, either directly or through outsource partner 9 **Facilities** Jones Lang LaSalle (JLL) or other 3rd party service provider as selected by AFI. The facilities services will be at differing levels at each location depending upon the respective lease terms and conditions but may include AFI providing managing use and access of facilities for ResCap employees including the costs and services the cost of the space, security services, building repair and maintenance obligations which are identified as tenant's obligations under the respective leases, equipment repair and maintenance, food services, mail services, print management Services, janitorial services, utilities, maintaining current levels of access to the facility for ResCap employees (i.e., badges, readers, access and security in all offices maintained by AFI). In addition, AFI provides space planning and project management services for any ResCap offices that require construction or renovation, moves, or other significant changes. Global Compliance AFI develops and maintains a compliance policy and procedure framework to support **Compliance** 10 ResCap in complying with relevant laws, rules and regulations, including, but not limited Services (1m) (includes to applicable federal consumer protection laws and regulations. This framework will also **Consent Order** establish the standards for the development, review process, publishing, and ongoing Costs) validation of compliance related policies and procedures across the ResCap and the Global Compliance and Regulatory Affairs organization. AFI develops, conducts, and maintains

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 97 of 126

Statement of Work	Functional	Scope of Services
(AFI to ResCap)	Service Area	
		a comprehensive compliance risk assessment process as a means to identify and measure the compliance risks to ResCap that may arise from violations of laws or regulations, ineffectiveness and/or weaknesses in the control environment, or from noncompliance with internal compliance policies, procedures, or standards. Compliance risk assessments serve as the foundation for the development and execution of the institution's compliance monitoring and testing program. AFI performs compliance training, monitoring, testing and reporting. In addition, AFI oversees ResCap's customer complaint process so that the appropriate tools and processes are in place, the capture and handling of complaints results in timely responses and escalation to ResCap's management, and reporting and analytic support to enable appropriate root-cause analysis to identify weaknesses and trends that may be indicative of systemic problems or violations. AFI performs an oversight role relating to key consumer practices / consumer protection / fair lending strategies and standards and alerting ResCap to relevant risk issues and emerging trends, and providing consultative guidance on such issues and support with regulatory exams as needed. AFI coordinates the completion of a compliance review of contract terms, assessment of compliance impacts of the service provided, a review of the compliance risk management program of the vendor including the data security environment, a determination of the appropriate service levels if applicable, and performance of ongoing compliance related oversight in coordination with the third party vendor management.  AFI provides a robust AML/OFAC program to support to ResCap and monitor regulatory compliance and provides detailed and accurate AML related reporting Identification Program; Customer Due Diligence and Enhanced Due Diligence; Transaction Monitoring; Customer Risk Scoring; Unusual Activity Escalation and Investigation; and FinCEN reporting. AFI notifies ResCap of regulatory and compliance policy changes. Thi
		information security. AFI will provide oversight and management of Consent Order

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document

Pa 98 of 126 **Scope of Services Functional** Statement of Work (AFI to ResCap) **Service Area** deliverables to promote compliance with the Order. **Global Privacy Services Compliance** AFI delivers services to ResCap through its Global Privacy function, which generally 11 provides such services by developing, implementing, and maintaining appropriate AFI (1N)policies and procedures so that consistent operating practices are in place to support ResCap in complying with applicable privacy laws and regulations, both directly and through the use of third-party providers. AFI arranges for and conducts appropriate privacy training for ResCap's employees to provide a sufficient level of understanding of the privacy laws and regulations that impact job responsibilities as well facilitating global privacy risk self-assessments in conjunction with ResCap's Data Protection Officers (DPOs). Privacy risk self-assessments are developed and maintained as well monitoring and testing of areas of greatest privacy risk for ResCap as determined through a documented risk assessment on an annual basis. AFI documents all events and incidents and reports impact assessed in the "Incident Notification" report provided to ResCap management and the appropriate Data Privacy Officers and Compliance Officer. AFI coordinate privacy Notices, consents, and opt-outs to ResCap's customers on behalf of ResCap as per applicable privacy policy. AFI performs security functions across its enterprise. AFI performs the following services 12 **Global Security Services Compliance** for ResCap: drafts and updates the Corporate Anti-Fraud Policy while assisting ResCap (10)with monitoring compliance and enforcing the policy. AFI maintains a case management system for the tracking, reporting and disposition of all security incidents. AFI is responsible for the investigation of all internal security incidents including but not limited to all categories of fraud, ethical/integrity misconduct, theft, security breaches, policy violations, and threats. AFI assists ResCap in managing the risk of external fraud and conducting all external fraud investigations and coordinating with law enforcement as appropriate to seek the recovery of lost assets. Global Security provides support to ResCap in assessing fraud risks in control environments, systems, and processes as well as assistance in assessing fraud prevention and detection tools for use in the ResCap organization. AFI develops content and provides instructors to perform training in live, WebEx, and on-line formats. AFI performs a security review on annual basis on all facilities housing a ResCap operation, and a written report is issued detailing findings, issues resolved, issues pending, and recommendations. AFI also assists ResCap management in providing procedures designed to enhance security and security awareness. AFI also provides security expertise and consultation to ResCap. Global Security analyzes internal and external threats of workplace violence and provides strategies in response including enhanced security measures, investigation, and coordination with law enforcement when appropriate. Global Security also provides training directed at minimizing the likelihood of violence and providing the appropriate response to threats.

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 99 of 126

	Statement of Work	Functional	Scope of Services
	(AFI to ResCap)	Service Area	
			Global Security provides travel security information to business travelers with destinations in high risk countries, as well as business intelligence regarding global locations where business opportunities are being considered. AFI maintains an internal and external hotline for issues relating to ResCap.
13	<b>Supply Chain Management</b>	Supply Chain	AFI gathers ResCap requirements, develops an initial list of potential suppliers, and
	Services (1P)		provide input into the sourcing process based on knowledge of supplier sourcing and risk assessment. AFI provides expertise regarding supplier sourcing including insight into current and past contracts for similar products or services. AFI will assist in development of contracts, negotiate contracts and Service Level Agreements on behalf of ResCap for suppliers ResCap has selected through the Sourcing process and assists with access and maintenance of the contract management application. AFI provides financial risk reviews and supplier risk assessments during the sourcing process in accordance with ResCap's direction. AFI assists ResCap as mutually agreed between the parties with the transition plan and termination notification during the off-boarding process.
14	<b>Human Resources Support</b>	Human	AFI provides many HR Support Services on behalf of ResCap and based on guidance and
	Services (1Qa)	Resources	oversight provided by ResCap. It provides recruiting support, provides systems and processes to track HR data as well as benefits design and administration of benefits through AonHewitt Benefits Service Center as appropriate, according to the terms of the Ally Financial Inc. Welfare Benefit Plan and the Ally Financial Inc. Retirement Savings Plan. Benefits include under these plans currently include: Medical benefits (Anthem BCBS), Prescription Drug benefits(Express Scripts), Expert Consultation service (Best Doctors), Wellness Programs (Alere), Health Savings Accounts (ACS Mellon), Dental benefits (Delta Dental of Minnesota), Vision benefits (VSP), Life Insurance coverage (Minnesota Life), Disability benefits (The Hartford), Flexible Spending Accounts, Employee Assistance Program (Empathia), and Pre-Tax Commuter benefits. It administers Tuition Reimbursement, Adoption Assistance, and Executive benefits through Accenture, as appropriate and COBRA administration.  Subject to reimbursement, AFI will provide and pay on behalf of all eligible ResCap employees to participate in and receive the benefits under the Ally Financial Inc. Welfare Benefit Plan and the Ally Financial Inc. Retirement Savings Plan. Benefits include under these plans currently include: Medical benefits (Anthem BCBS), Prescription Drug benefits(Express Scripts), Expert Consultation service (Best Doctors), Wellness Programs (Alere), Health Savings Accounts (ACS Mellon), Dental benefits (Delta Dental of Minnesota), Vision benefits (VSP), Life Insurance coverage (Minnesota Life), Disability benefits (The Hartford), Flexible Spending Accounts, Employee Assistance Program (Empathia), and Pre-Tax Commuter benefits.  Subject to reimbursement, AFI will continue to offer to eligible ResCap employees certain

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 100 of 126

	Statement of Work	Functional	Pg 100 of 126 Scope of Services
	(AFI to ResCap)	Service Area	
			other employee benefits, including Tuition Reimbursement, Adoption Assistance, Executive benefits, and other related benefits offered as of the date hereof, and make all payments to or on behalf of ResCap employees for these benefits. Subject to reimbursement, AFI will make monthly payments of all premiums, administrative fees, claims and other invoices, as applicable to all insurance companies and third party administrators relating to the plans and benefits listed herein. AFI designs and provides governance for compensations as well providing performance management design, administration, and support. AFI administers training programs, employee
15	Legal Services (1S)	Legal	relations, manages headcount analysis and tracking as well new hire orientation and off-boarding of staff.  AFI provides legal advice and counseling, including regarding changes in laws and regulations applicable to ResCap's business, as may be necessary from time to time, in the
			substantive areas of employment law and federal regulation of bank holding companies and their affiliates as well as drafting agreements pertaining thereto. AFI provides law department management support, consisting of identification and management of law department information management software applications including administration of law firm engagement, oversight, evaluation, e-billing and payment processes and other services provided by the outside counsel relations team.
16	Loan Review (IT)	Loan Review	AFI assists ResCap with establishing and updating an Annual Loan Review Plan. For each individual loan review, AFI performs data analytics and physical file reviews to assess compliance with internal credit policies and loan administration procedures. AFI provides reports with respect to compliance with lending policies and loan administration procedures, assessment of credit policy and procedures, credit risk exposure, provides an annual review of the Allowance for Loan Leases Losses ("ALLL") qualitative reserves including the components of the qualitative reserve calculations and evaluates the activities of lending personnel including their compliance with lending policies and the quality of their loan approvals, monitoring and risk assessment.
17	Treasury Services – Global Funding and Liquidity (1W1)	Treasury	AFI performs a variety of treasury related functions pertaining to short-term liquidity and funding Planning, produces daily liquidity forecast for a rolling 90 to 120-day period, maintain ResCap's unrestricted cash balance at targeted levels given by ResCap Treasury management, facilitates and conducts forecasting discussions with ResCap's business areas to confirm reasonableness of forecast assumptions, provide regular or ad-hoc reporting, within mutually agreeable timeframes, to ResCap's management and board of directors. In addition, AFI provides long term liquidity forecasting and funding planning, tracks and monitors open audit related issues (i.e., internal audit, SOX, etc.), oversees and coordinate Treasury Business Continuity Plans, provides liquidity strategy and liquidity risk oversight and advisory services as well as ad hoc analysis performed as requested by

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document

Pg 101 of 126 Scope of Services **Functional** Statement of Work (AFI to ResCap) **Service Area** ResCap. **Treasury Services Treasury** AFI identifies and build triggers into the monitoring process at the inception of all new 18 deals coordinates internal resources for internal and external audits as necessary, supports **Structured Funding Deal** the development of investor / servicer reports with Securitization Accounting and Legal as **Compliance and Facility** necessary, coordinates timely completion and delivery of daily, monthly, quarterly, and Management (1W2) annual requirements for all ResCap secured facilities, e.g., USAP, Regulation AB, SEC filings, Officer Certificates (monthly certification process), audited/unaudited financials, collateral reports, etc. Pool builds and collateral management. In conjunction with Capital Markets, as applicable, manages all collateral and coordinates the process for selecting assets to be sold or financed in secured transactions and performs debt structuring and negotiation, modeling/analytics as well producing all data requested by rating agencies and ad hoc support and analysis. Treasury Services, **Treasury** AFI's Capital Markets Operations supports the capital markets business which executes 19 fixed income, equity, FX and derivative trades by providing daily cash forecasting, trading **Treasury Operations** product management relating to management of trading product collateral/margin, (1W7)management of new counterparty set up, management of new products and processes, and management of quality of data. This involves performing collateral / margin calls with external and internal counterparties, counterparty management for Master Agreements for ISDA (derivatives), MSFTA (TBA), Repo, and Futures Clearing (futures/options), new product and process management:, data integrity management. Other treasury services include bank account administration for management and delivery of cash management services from banking . AFI performs cash and wire desk services. **Government Relations (1X)** AFI will work with ResCap's Voice of Customer Team in responding to congressional Government 20 inquiries on mortgage related issues. AFI will work with internal and external legal Relations counsel in connection with ResCap's responses to governmental investigations involving ResCap. AFI will support ResCap with the following tasks: manage government relations aspects of investigations, including communications with Congressional staff, GAO official and SIGTARP staff and manage the retention and work of external government relations consultants and coordinate the distribution of responsive materials. AFI will provide advice to ResCap from time to time, on political developments and issues and government actions that are relevant to ResCap. AFI's Communications area will deliver to ResCap employee and executive Communications and 21 **Communications** communications services. It also performs services relating to digital communications, investor Relations Services and investor relations. AFI provides Security Architecture and Consulting Services to verify if ResCap is 22 **Global Information ITG** compliant with Information Security Policy standards and where applicable, baseline

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document

Pg 102 of 126 Scope of Services **Functional** Statement of Work (AFI to ResCap) **Service Area** Security (1R1) configuration requirements. AFI Identities and accesses management Services. AFI provides operational support of ResCap's firewalls and intrusion detection capabilities and will monitor these devices. AFI provides computer security incident response support. AFI provides monthly reporting on Information Security Governance. In addition, AFI will provides supervision for IT-related services to monitor that there are proper segregation of duties (i.e. firewall rules proposed by platform support team are approved prior to implementation by IBM); AFI is responsible for the delivery of technology projects and the maintenance of 23 **Application Development ITG** applications worldwide. AFI functions as a strategic adviser to the enterprise and to and Support - IT ResCap to provide new system capabilities while efficiently facilitating day-to-day **Corporate Services (1R9)** operations. The Services can be organized into two broad categories including application development and enhancement and application support and maintenance. ITG AFI's Enterprise Architecture team and the underlying architecture development processes **IT Technology Strategy** 24 (e.g., from requirements through planning, testing and execution). The group works to and Architecture Services verify that all projects and related architecture are aligned with the overall strategic IT direction set forth by the IT Standards Council and the IT Architecture Review Board. The group also creates and maintains the Architecture templates defined in the Ally Project Execution (APEX) methodology. The templates are completed by the Enterprise Architecture team (or by an outsourced third party). **Cross Functional Services** ITG AFI performs a variety of IT related services for ResCap. It involves different aspects of 25 IT related management services including availability management, IT continuity management, incident management, problem management, service level management. Other services include providing management and reporting of common platforms and systems including Clarity (project management tool) and GEAR (enterprise architecture tool), Datastage, Informatica, Business Objects, Cognos and EFT (Enterprise File Transfer), testing services including performance testing, automation, security testing, test data masking and test tool management and support, providing database architecture, database standards and governance, database change and incident technical management, database security compliance and database administration, supporting the delivery of data between applications and from/to a third party vendor; provide extraction, transformation and load of data between applications, providing management and leadership of functional infrastructure areas, including team goals, priorities and work assignments, providing asset tracking service including both automated and manual process for tracking information systems hardware equipment and support configuration management database linkages between the assets, providing authorized user ID administration, ensuringe standardized methods, processes and procedures are used for all changes; maintain the balance between the need for change and the potential detrimental impact of changes.

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 103 of 126

	Statement of Work	Functional	Pg 103 of 126 Scope of Services
	(AFI to ResCap)	Service Area	
26	End User Computing Services	ITG	AFI provides ResCap service desks for all business application services; provide incident management services, provides executive and critical site support; provide general installs, moves, adds and changes of equipment and applications, manages all aspects of projects related to business unit branch openings or relocations and provides logistical coordination of EUC shipments, manages vendors in accordance with contractual obligations, perform service improvement meetings, and ensure properly defined performance improvement plans, supports and perform audits, manage outstanding audit issues and define remediation plans as requested.
27	Voice Services Mobility	ITG	AFI assists ResCap so that it conforms to AFI's security standards; support in the development of mobility standards and design guidelines; perform all mobility service activities, plans and services. AFI manages the mobile environment, provide technology improvements where possible, provide portable network device administration and management of mobile phones
28	Network Services – Local Area Network	ITG	AFI provides ResCap with general LAN management services general project management, provides third party management and coordination and provide information for audit and performance reviews; provides planning and designing of LAN projects to ensure a high quality network LAN environment; provides general monitoring and management of the LAN environment and provide enterprise operations monitoring support; provides predictive analysis, incident management and problem management, manages LAN capacity; reports and analyze LAN performance; plans, manages and perform IMACs including hard, soft, expedited and project IMACs; conducts and documents Site surveys; maintain locations lists, LAN diagrams, configurations and other LAN documentation; provide equipment space, power and security; provide and manage cabling and wiring; manages third parties and software upgrades and test all new LAN software; provides incidental network hardware; manage relationship with third party suppliers that manage network devices and the general LAN environment; provides LAN equipment and software, implement LAN connections, and provide various other LAN services; engages in transport voice and video protocols over the IP LAN, maintain LAN connectivity, provide a multi-protocol LAN network and provide various other technical LAN services; manages and control LAN connectivity between all sites and external networks; provides wireless telecommunication systems and support for office connectivity.
29	Network Services – Network WAN Services	ITG	AFI provide general WAN management services; third party management and coordination and provide information for audit and performance reviews; provides planning and designing of WAN projects to ensure a high quality network WAN environment; supports equipment provisioning, equipment retirement and asset management support; provides general monitoring and management of the WAN

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 104 of 126

	Statement of Work	Functional	Pg 104 of 126 Scope of Services
	(AFI to ResCap)	Service Area	
			environment, provide enterprise operations monitoring support; provides proactive analysis and network incident management; manages network capacity; report and analyze network performance; provides transport support and engineering services, manages and controls connectivity between all sites and external networks; manage and operate all network equipment and software; plans, manages and performs IMACs including hard, soft, expedited and other IMACs; conducts and document Site surveys; provides equipment space, power and security; provide and manage cabling and wiring; provides incidental network hardware; manage the relationship with third party suppliers that manage network devices and the general WAN environment; and supports with general technical requirements, support with system network architecture (SNA), support with routing and data interchange protocols and provide transport for network time protocol servers.
30	Voice Services – Telecommunications and Contact Center	ITG	AFI's provides Telecom call center support to ResCap by establishing security standards; supports in the development of telephony standards and design guidelines; performs all voice service activities, plans and services; manages the voice network including equipment and software; manages third parties; manages voice station headsets, manages unified messaging and provides turret voice services and remotes worker telecommunication services; manages contact center applications, manages and coordinate third parties, assists in planning and design, manages equipment and software, provide fault management, provide capacity and configuration management and assist in performance optimization; provides AFI personnel properly trained to provide services.
31	Hosting Operations – Hosting, Implementation and Data Center Services	ITG	AFI provides ResCap with end to end operation services and provides monitoring and management of AFI's regional operation centers responsible for services in ResCap's service zone; provides all integration, design, management and support for equipment and software; provides and manage a VLAN and premise network environment in hosting centers; provides architecting, implementing ad support of storage management solutions; maintains and provides support for job and batch scheduling and provides enterprise operations monitoring support; manages all components of the server environment including equipment, software and related infrastructure; provides performance statistics and provide technical support and advice for applications; provides ResCap identifier service, non-Windows LDAP, and meta directories; deploys all approved patches to all servers following the change management process development; provides normal business hours for on-site system administrative support for the pre-production environment application operations support; tests and clarifies server builds based on templates and specifications provided; provides email server refreshment, mail management, server management and collaborative application management; supports and maintains the configuration of the interface to LDAP global internet mail relay; provides DHCP and

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 105 of 126

Statement of Work	Functional	Scope of Services
 (AFI to ResCap)	Service Area	
		BootP services and provide internal DNS and DDNS services with active directory
		integration; provides documented recovery action plans, basic disaster recovery and
		management of centralized backup; manages; domain name service (DNS), server name
		registration and Windows domain requests; support global DNS maintenance, NIS
		integration and network time protocol (NTP); supports in development environment, pre-
		production environment and production environment; provides SMTP mail services, gold
		build deployment and facility management services.

# 12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 106 of 126 Summary of Statements of Work From ResCap to AFI

	Statement of Work (ResCap to AFI)	Functional Service Area	Scope of Services
1	Human Resources Support Services (1Qb)	Human Resources	ResCap provides support for appropriate organizational design based on the business strategies of the organization, including: (i) provide appropriate functional and process alignment; (ii) define reporting relationships; (iii) evaluate spans of control; (iv) provide appropriate role definition and scope; and (v) support change impacting the organization.  ResCap manages headcount analysis and tracking, provides management reporting and analysis to AFI, provides project management services in support of business initiatives, provides management and support of HR programs including Recruiting, Comp Plan Administration, and Performance Management, provides support for audit, regulatory, legal, and ad hoc requests as required and administers position approval process.
			ResCap provides the following compensation plan support: (i) facilitate annual compensation plan review process for AFI working with AFI's Affiliate's Global Capital Markets and Broker Dealer, HR Business Partners, and the AFI Compensation Team; and (ii) administrative oversight for variable plans have been reviewed by appropriate parties and payment modeling has been performed.  ResCap provides support for AFI's performance management process including individual development plans, mid-year performance reviews, and year-end performance reviews and provides assistance in setting objectives/requirements, establishing timing, and providing support and expertise to AFI as related to the above mentioned performance management processes.
2	Legal Services (1)	Legal	ResCap provides Ad-Hoc Consultative Services  During the Term, ResCap will provide the following Services to AFI in connection with its business activities when requested by AFI and consistent with current and historical practice: (i) Legal advice and counseling, including litigation management and support, including service of process, e-discovery, discovery responses, and litigation holds, as may be necessary or required by AFI from time to time, including without limitation, making available relevant systems and software that may be developed by or for the e-discovery team; (ii) drafting legal agreements and other legal documents in connections with AFI's capital markets functions, including, but not limited to, funding facilities, whole loan sales, and securitizations; (iii) obtaining and maintaining licenses required for AFI's and its subsidiaries' respective businesses, to the extent required; (iv) assistance with legal aspects of investigations and examinations in connection with oversight of the mortgage business by federal, state, and local regulatory bodies; (v) retention, oversight, and management of outside counsel representing AFI as determined by ResCap's legal staff to be appropriate to conduct litigation and/or provide advice, counseling and other legal support as to legal matters of AFI, including in respect of capital markets activities; and (vi) make available to AFI for its use the applications listed in Section VIII below.

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 107 of 126

	Statement of	Functional	Pg 107 81 126
	Work	Service	Scope of Services
	(ResCap to AFI)	Area	
3	IT Resource Services	ITG	AFI is responsible for the delivery of applications and services worldwide for its business. ResCap provides support in the nature of personnel, assets, facilities and tools ("ResCap's Resources") that facilitates AFI's day-to-day operations. ResCap's Resources assists AFI's ability to support the delivery of applications to existing applications per AFI's enterprise standards and methodology.
4	ResCap Accounting Services	Accounting	During the Term of the Agreement, ResCap will provide the following services to AFI: (i) General Accounting Operations; (ii) Financial and Regulatory Reporting; and (iii) subject matter expertise and support activities to comply with certain of AFI's Global Financial Control Policies:
5	Master Servicing Bond Modeling (Auto 1)	Master Servicing	ResCap will perform monthly bond administration functions compliant to securitization deal documentation and as instructed by AFI including: (i) performing model validations on all securitizations within 1 business day after completion of distribution calculations and prior to reporting and distribution of data and investor remittance amounts; (ii) performing initial modeling and ongoing bond administration for newly originated and assigned deals; (iii) coordinating with AFI in addressing audit requests, issues or concerns as raised; (iv) indentifying and incorporating process improvements surrounding existing monthly distribution processes and integrated workbooks including; (v) coordinating with ITG in resolution of spreadsheet deficiencies as reported on monthly defect logs; (vi) reviewing and recommending logic changes to data queries; access and excel macros and related data exchanges; (vii) reviewing and modifying waterfall validation spreadsheets incorporating excel and access enhancements; and (viii) distributing reports to Investors.
6	Master Servicing Bond Administration (Auto II)	Master Servicing	ResCap will receive monthly, various reports and data files from AFI (or its agent) encompassing the automotive servicing activity for the respective security transactions for the prior month. ResCap will process the activity and calculate amounts owed to and from various parties to the whole loan transaction. Mark to Market deals will be completed and delivered to AFI by EOD on the 6 <sup>th</sup> Business day of the month. Non Mark to Market deals will be completed and delivered on the 10 <sup>th</sup> business day of the month.
7	Compliance Services (14H)	Compliance	ResCap will deliver Services through its Compliance and Licensing functions. The following is a general description of the services that may be performed –
			Compliance Oversight and Support for the Mortgage LoB: ResCap will provide sufficient compliance resources that are dedicated to oversee and support the Client's compliance needs regarding its Mortgage LoB, which will include Client providing a dedicated Mortgage Compliance Officer and team. Such dedicated Mortgage Compliance Officer and team will provide appropriate Compliance integration with and oversight of the Client's mortgage business operations, including input on product design and development, projects and key initiatives, operational processes, business strategies and marketing efforts.
			Licensing: ResCap will provide sufficient licensing resources that are dedicated to applying for and maintaining

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 108 of 126

			Pg 108 01 126
	Statement of Work	Functional Service	Scope of Services
	(ResCap to AFI)	Area	
			corporate, branch and individual licenses needed to support the Client (based upon the advice of Client's Legal staff as to licenses needed), which shall include a dedicated Licensing staff. Such dedicated Licensing staff will ensure timely filing of applications to ensure that the Client is in good standing and is appropriately licensed to conduct its business activities.
8	Risk Services Reporting and Special Assets	Risk	ResCap will provides risk reporting, as well as support AFI in the valuation of assets for governance and reporting purposes, seek to liquidate assets at the most economical price for AFI's benefit, evaluate AFI's strategic alternatives and disposition of assets, support AFI with management services of direct personnel and 3 <sup>rd</sup> party brokers and contractors, provide valuation and risk rating support for tracking of criticized credits and other assets, provide approval of work out resolution recommendations and any loan or asset advance requests and timely and accurate responses to issues that arise on SAG accounts.
9	Application Support (14A)	ITG	ResCap facilitates day-to-day operations by maintaining existing system capabilities for AFI. The Services are application enhancement, and application support and maintenance.  ResCap maintains and supports the various application environments including configuration, processing, monitoring, maintenance, trouble shooting, bug correction, application hosting, application maintenance, meeting uptime requirements, responding to incidents and problems as well as disaster recovery. Services also include support for SOX and Enterprise Access Controls (EAC) compliance and responding to audit and other management inquiries. These services can be summarized into application maintenance, application support, information management and reporting, compliance and controls and support for other ad-hoc requests.
10	Capital Markets (14B)	Capital Markets	ResCap will (i) support AFI Lending Channels and Management with product updates and implementation; (ii) negotiate all agreements with Government Sponsored Enterprises (GSE's) and Mortgage Insurance (MI) partners; and coordinate any new contracts and / or updates to existing contracts.  Loan Pooling, Securities/Loan Delivery, and Settlement: ResCap will obtain commitment authority money and pool numbers. Identify loans in "targeting lending areas" using thresholds set by GNMA, FNMA and FHLMC. Pooling process in accordance with the criteria set by agency / investor. Send notification to GNMA, FNMA and FHLMC when pool is certified. Settlement Sheet Notification. Delivery of AFI owned loans to the agency or investor. Require that all loans are eligible for delivery to the agency or investor. Pipeline Hedging, Trading and Pricing. Assist in the facilitation of the following at the direction of AFI or Ally Investment Management (AFI IM). Hedging and subsequent sale of every originated/purchased loan (Jumbo HFI – excluded for the purposes of hedging and trading). Implementation and Execution of "best execution" model. Advise AFI regarding margins, product development, credit guidelines, and operational processes. Work with RESCAP Treasury and RESCAP IM to obtain and maintain counterparty relationships. Provide base daily pricing to AFI inclusive of all intra day re-pricing. Provide daily pricing (including margins) for rate sheet flow, direct trade, Bulk Bid, as well as AOT pricing. Consult with AFI Lending

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 109 of 126

	Pg 109 of 126				
	Statement of Work	Functional Service Area	Scope of Services		
	(ResCap to AFI)	Alea			
			Channels and Management regarding the following: Current market conditions or changes in market conditions.  Origination volumes and pricing, and potential management decisions, actions or steps to keep pricing and volumes aligned with management objectives. The selection of counterparties for funding and hedging. Produce pricing necessary for daily mark to market. Maintain hedge model (QRM) to ensure all market/GSE assumptions are current. Facilitate purchase of loans via bulk bid/direct trade with AFI Correspondents. Capital Markets Program Office. Coordinate IT/Business strategies and policies across Capital Markets Platforms. Application Management. Portfolio Management/Analytics. Loss Mitigation strategy. Manage repurchased assets: Asset valuation modeling:		
Len Gov Ope wor			During the Term of the Statement of Work, RESCAP will provide the following services to AFI: (i) support AFI Lending Channels and Management with product updates and implementation, (ii) negotiate all agreements with Government Sponsored Enterprises (GSE's) and Mortgage Insurance (MI) partners, (iii) participate in Mortgage Credit Operation Risk Committee (MCORC) where all product changes/implementations are reviewed and approved, (iv) work closely with Trading and Credit Risk to require products are safe and sound, (v) create and maintain product support documentation, (vi) Coordinate any new contracts and / or updates to existing contracts.		
11	HR Support Services (14C)	Human Resources	ResCap will provide the following HR Support Services: Organizational Design,		
			Employee Relations, Headcount Management, Compensation Design & Administration Services, Staffing, Performance Management/Talent Management/Engagement Services and Enterprise Training Requirements, Workforce Planning & Analytics, and Miscellaneous Human Resource Services		
12	Risk Management and Data Collection (14D)	Risk	ResCap will develop, maintain, and update all lending product reference material (product summaries), document matrices, prepare communications and memos to external constituencies, coordinate, on behalf of Bank, relationships with agencies (FNMA, FHLMC, FHA, VA, USDA, and GNMA) on credit related issues, maintain and update the credit chapters of Bank's Guide for all product types, support the development, implementation and maintenance of credit risk support tools, training modules, delegated underwriting authority levels, exclusionary list and condominium management, underwrite performance oversight, develop rules and support maintenance of existing product eligibility and credit risk guidelines within the proprietary automated underwriting system (AUS) (Custom DU) and Engenious Product Engine, provide support to Underwriting/ Product Development on risk related loan transactional/product policy concerns, provide exception oversight on loan eligibility & credit risk exceptions associated with individual loan transaction, bulk transactions and channel process changes, provide administrative support to Mortgage Credit Operational Risk Committee.		
			Credit Quality Assurance and Fraud. Loan audits are performed on loans for Bank, for compliance with investor and company loan programs, underwriting guidelines, policies, procedures and fraud detection.		
			Quality Assurance will make additional targeted discretionary loan selections to address identified trends or other		

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 110 of 126

Statement of Work (ResCap to AFI)	Functional Service Area	Pg 110 of 126  Scope of Services
		higher risk areas. Delinquent Audits: Delinquent loan reviews encompass a sample of loans from all Channels.
		Mortgage Fraud Risk Operations (MFRO) provides fraud awareness training. The training program is designed to inform participants of existing, new and emerging fraud schemes, how to identify red flags and elevating concerns. Fraud training is designed to include origination and servicing employees. Fraud Audit: Reporting and alerts identify lenders with the highest risk thresholds.
		Appraisal Review: Review of specific appraisals, assessment of Appraisal Management Companies (AMCs), validation of appraisal assessments, vendor monitoring.
		Servicing Risk Management, including (i) daily, monthly and quarterly Mortgage Servicing Rights (MSR) valuation, (ii) MSR asset and portfolio reporting, (iii) creation of loan and tranche data tapes, (iv) documentation and performance of SOX controls, (v) support data and information requests for MSR, and (vi) MSR benchmarking.
		Credit Risk Modeling - Credit risk analytics and benchmarking, including Ally Bank HFI reserve calculation, ResCap performs the following activities on a monthly basis:
		Perform a "first look" analysis to verify the reasonableness of the monthly WestPat and SGM (see "Systems Applications") cycles.
		Perform SOX controls related to the ALLL calculation process, and provide evidence to the SOX team.
		Bank HFI ALLL risk analytics and reporting: RESCAP performs the following activities on a monthly basis
		Bank HFI ALLL governance and financial reporting: RESCAP performs the following activities on a quarterly (or more frequent, as needed) basis.
		Calculation of HFS valuation levels (to Bank Accounting) for quarterly financial disclosure; participation in meetings with Bank and capital markets representatives to review results.
		Analysis and presentation of potential updates to WestPat and SGM behavioral model assumptions to the appropriate Bank management and/or Board level Committee.
		Preparation and presentation of quarter-end levels, analytics and benchmarking for the appropriate Bank management and/or Board level Committee approval.
		Bank HFI internal and regulatory analytics and reporting: ResCap performs the following activities as needed (typically quarterly), including analysis and presentation in response to Regulator requests (as needed), Bank HFI loss model implementation: Analytics and Controls / Governance / Regulatory presentation related to the implementation of new behavioral and/or calculation models.

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 111 of 126

	Statement of	Functional					
	Work	Service	Scope of Services				
	(ResCap to	Area					
	AFI)						
			Investor Repurchase Review: The Investor Repurchase team operating at the sole discretion and authority of a designated Officer of AFI, reviews, responds to, and negotiates on repurchase or indemnification requests from investors for loans where Bank owns the Mortgage Servicing Rights (MSR). In addition, Investor Repurchase review Notices of MI Rescission put forth by MI providers for loans where Bank owns the MSR. Investor Repurchase review and responds to investor findings, and provides Repurchase Reporting and Analytics.				
			Risk Applications -				
			a. Risk Application Business Design and Maintenance:				
			b. Application Access:				
			c. Business Recovery Planning: Coordination of Business Resiliency Activities to enable compliance with the Bank Business Continuity Policy.				
			d. Operational Risk Management: Coordination of ORM activities.				
			Counterparty and Market Risk -				
			a. Calculating counterparty exposures for derivative transactions, operating balances, forward settling transactions, and other relevant transactions and producing reporting to test compliance with established limits by counterparty.				
			b. Market Risk Limit monitoring across all portfolios.				
13	Legal Services (14E)	Legal	During the Service Term, ResCap shall maintain its current level of service and support and will provide the following services to AFI in respect of AFI's residential mortgage activities:				
			Where requested by AFI, provide legal analysis and support as may be necessary or required by AFI from time to time, including without limitation, support for the mortgage correspondent, warehouse, and wholesale lending lines				
			Inform and instruct AFI management regarding changes in laws and regulations affecting AFI;				
			Examination support;				
			Review marketing materials;				
			Fair lending and fair credit support, supplemented by AFI's legal staff;				
			Licensing matters not otherwise covered;				
			Retain and manage outside counsel as determined to be appropriate to conduct litigation and/or provide assistance in				

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 112 of 126

	Statement of Work (ResCap to AFI)	Functional Service Area	Scope of Services
			other specialized matters; and
			Other services as requested from time to time by AFI
14	Accounting	Accounting	During the Term of the Agreement, ResCap will provide the following services to Ally Financial Inc.:
	(14F)		Accounting Operations: ResCap provides accounting services and subject matter expertise related to domestic residential mortgage loan activities to AFI, including services pertaining to:
			Mortgage Loan Accounting: (i) Mortgage Loan Inventory, and (ii) Servicing Operations.
			Real Estate Owned (REO) Accounting.
			Corporate Accounting.
			Transaction Support.
			Financial and Regulatory Reporting.
			Financial Systems Support.
			Ad-Hoc Audit Support.
		ResCap provides financial information that is needed for AFI's consolidated forecast and business plan activities. ResCap also provides support as requested, and jointly agreed to between ResCap and AFI, for ad-hoc analysis and reporting (e.g., FED / SEC requests.). Services include:	
	Analysis (14G)		A. GAAP / Managerial forecasts at the product and portfolio levels
			<ol> <li>Financial forecast and planning services for AFI's Affiliate, Ally Bank's Mortgage business including balance sheet (new loan volume, loan sales, asset liquidation/termination, allowance for loan loss and other asset balances) and income statement items (revenue by product and expenses including depreciation, interest, provision, operating, etc.).</li> </ol>
			2. Forecasts are provided, at a minimum at the direction of Ally Corporate FP&A, typically quarterly for detailed forecasts, monthly for executive forecasts, or more frequently as requested by AFI (for more frequent requests, delivery timeframe jointly agreed to between ResCap and AFI on a case by case basis).
			B. GLR / cash flow forecast
			<ol> <li>Liquidity forecast services for AFI's Affiliate, Ally Bank's Mortgage business including cash- related balance sheet items (new loan volume, loan sales, asset liquidation/termination) and cash-</li> </ol>

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document
Pg 113 of 126

	Statement of Work (ResCap to AFI)	Functional Service Area	Scope of Services			
			related income statement items (revenue and expenses).			
			<ol> <li>GLR Forecasts are provided at the direction of AFI's Affiliate, Ally Bank Treasury, typically monthly (for more frequent requests, delivery timeframe jointly agreed to between ResCap and AFI on a case by case basis)</li> </ol>			
			C. Portfolio stress tests			
			<ol> <li>Provide impacts to baseline forecasts under various stress scenarios (declining HPI, increasing market delinquencies) as defined by Ally Risk (delivery timeframe and reasonability of stress scenarios jointly agreed to between ResCap and AFI on a case by case basis)</li> </ol>			
			<ol> <li>Completed at the direction of Ally Corporate FP&amp;A and Ally Risk, usually quarterly (for more frequent requests, delivery timeframe jointly agreed to between ResCap and AFI on a case by case basis)</li> </ol>			
			D. Ad hoc reporting and analytics			
			<ol> <li>Support AFI's Affiliate, Ally Bank's requests for reporting and analytics related to AFI's Affiliate, Ally Bank's Mortgage assets, typically at a product or portfolio level. Product is defined as Vintages and Asset Classes which includes but is not limited to Held for Investment (HFI), Mortgage Servicing Rights (MSR) and Warehouse.</li> </ol>			
			<ol> <li>Ad hoc reporting and analytics provided to AFI's Affiliate, Ally Bank typically includes items such as detailed forecast assumptions, and variance explanations (forecast vs. forecast, actual vs. forecast, etc.), portfolio performance summaries, reserve analysis</li> </ol>			
			<ol> <li>Ad-hoc reporting and analytics in support of special projects as jointly agreed to between ResCap and AFI</li> </ol>			
			4. Ad-hoc support is provided upon request from AFI and jointly agreed to by ResCap			
16	Record Services	Capital	During the Term of the Statement of Work, ResCap will provide the following services to AFI:			
	(141)	(141) Markets	Facilitates Treasury and Capital Markets in the determination/confirmation of custodial file locations and updates custodial locations in LoanServ & the Custodial File Database and performs research functions to reconcile custodial locations for files that are shown as not on hand.			
			Process requests for files/documents that are received from internal and external clients for loans sales, litigation requests, audits and reconciliation projects and deliver files/images to requestor in a mutually agreeable format that is			

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pa 114 of 126

	Statement of Work (ResCap to AFI)	Functional Service Area	Pg 114 of 126  Scope of Services	
	AII)		suitable to their request (i.e. electronic format).	
			Test and coordinate modifications to a system/database when needed.	
1=			Responsible for auditing Vendors for compliance to the agreed upon Service Level Agreement (SLA).	
17 Business Risk and ResCap will:  Risk and Controls Risk and ResCap will:		·		
	and Business	Controls	Perform testing to ensure key operational procedures are performed timely and are adequately documented;	
	Excellence	and	Perform independent testing/monitoring to identify gaps in controls;	
	Support (14J)	Business Excellence Support	Monitor new and/or revised business practices/strategies, system changes, policies and procedures to ensure that associated risks are identified and addressed;	
		Support	Provide cross functional knowledge to identify risks, controls, and best practices;	
			Partner with AFI and Operational Risk Management in the development of the annual key process assessment and respective targeted risk assessments;	
			Maintain QA program for MERS;	
			Provide Data Privacy incident management;	
			Issue Escalation and Issue Tracking/Action Plan status reporting;	
			Regulatory and Business Continuity Coordination;	
			Centralized Variable Compensation Administration; and	
			Audit Management.	
18	Capital Markets Accounting (14L)	Accounting	ResCap will provide accounting & reporting support for the Bank Pipeline Swap, including prepare journal entries associated with the Bank Pipeline swap (inclusive of the pipeline and inventory mark) for recordation by Bank, create and maintain procedural documentation in support of the Bank Pipeline Swap., Submit derivative details into the accounting reporting tool - for SEC disclosures and regulatory reporting, and prepare, review and submit Bank disclosures via an upload to the shared AFI Reporting team rooms:	
			ResCap will also provide accounting & reporting support for the Total Return Swap (TSR), which includes calculating the daily valuation of the Total Return Swap as defined by the legal confirmation (daily and monthly components) as directed by AFI, preparing journal entries associated with the Total Return Swap to be recorded by AFI, including detailed support and assumptions behind the daily exchange, on a daily basis, creating and maintain procedural documentation in support of the Total Return Swap, submitting derivative details into the accounting reporting tool -	

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 115 of 126

	Statement of Work (ResCap to AFI)	Functional Service Area	Scope of Services	
			for SEC disclosures and regulatory reporting, and preparing, reviewing and submitting Bank disclosures via an upload to the shared AFI Reporting team rooms:	
			ResCap will also provide Mark to Market Accounting.	
19	Consumer Lending Services (14M)	Consumer Lending	ResCap's consumer lending operations shall provide the Services in connection with loans brokered to AFI by ResCap on the Eclipse loan origination system including: (i) State audit review utilizing Global Compliance State Matrix; send special flood hazard notice to borrower when applicable; (ii) obtaining title commitment and review for clear to close; (iii) ensuring that closing agent is approved through Closing Agent Verification Process; and (iv) schedule loan foreclosing.	
			ResCap also provides: (i) loan Closing Services; (ii) Loan Funding Services; (iii) Wire Desk services; and (iv) RESPA Curative Services.	
20	Client Repurchase (14N)	Risk	Loss recovery: negotiates repurchase/recovery from correspondent and/or broker on Ally Bank owned loans to mitigate potential loss to AFI's Affiliate, Ally Bank and recapture actual losses on loans that have violated one or more of the loan level representation, warranties, and/or covenants outlined in the applicable contract between AFI's Affiliate, Ally Bank and the correspondent and/or broker. ResCap Client Repurchase develops and distributes repurchase-related communications to clients associated with the repurchase/make whole process and takes action necessary to enforce said obligations (e.g., Repurchase Demand, Notice of Default, Indemnification Agreements, Payment Plan/Settlement Agreements, etc) as directed under the sole discretion by an Officer of AFI's Affiliate, Ally Bank.	
			Loss recovery accounting: on behalf of AFI's Affiliate, Ally Bank, calculates total amounts due for recovery on a loan level basis for repurchase or make whole. ResCap provides reconciliation of monies received from clients as "repurchase funds" or "make whole funds" due AFI's Affiliate, Ally Bank.	
			Loss recovery controls and protocol: provides to AFI's Affiliate, Ally Bank clients who have breached the protocol by publishing the Early Warning Report.	

## **EXHIBIT C**

UNITED STATES BANKRUPTCY COURT	I
SOUTHERN DISTRICT OF NEW YORK	

	)	
In re:	)	Case No. 12-
	)	
RESIDENTIAL CAPITAL, LLC, et al.,	)	Chapter 11
	)	-
Debtors.	)	Jointly Administered
	)	•

INTERIM ORDER UNDER BANKRUPTCY CODE SECTIONS 105(a) AND 363(b) AUTHORIZING RESIDENTIAL CAPITAL, LLC TO ENTER INTO A SHARED SERVICES AGREEMENT WITH ALLY FINANCIAL INC. NUNC PRO TUNC TO THE PETITION DATE FOR THE CONTINUED RECEIPT AND PROVISION OF SHARED SERVICES NECESSARY FOR THE OPERATION OF THE DEBTORS' BUSINESSES

Upon the motion (the "Motion")<sup>1</sup> of the Debtors for entry of interim and final orders, under Bankruptcy Code sections 105(a) and 363(b), authorizing the Debtors to enter into a shared services agreement (the "Agreement") with Ally Financial Inc. ("AFI") *nunc pro tunc* to the Petition Date for the continued receipt and provision of shared services necessary for the operation of the Debtors' businesses; and upon the Whitlinger Affidavit; and it appearing that this Court has jurisdiction to consider the Motion pursuant to 28 U.S.C. §§ 157 and 1334; and it appearing that venue of these Chapter 11 cases and the Motion in this district is proper pursuant to 28 U.S.C. §§ 1408 and 1409; and it appearing that this proceeding is a core proceeding pursuant to 28 U.S.C. § 157(b); and sufficient notice of the Motion having been given under the particular circumstances; and it appearing that no other or further notice need be provided; and it appearing that the relief requested by the Motion is in the best interests of the Debtors' estates,

Capitalized terms not otherwise defined herein shall have the meanings ascribed to such terms in the Motion. Creditors and parties-in-interest with questions or concerns regarding the relief granted herein may refer to <a href="http://www.kccllc.net/rescap">http://www.kccllc.net/rescap</a> for additional information.

their creditors, and other parties in interest; and after due deliberation thereon; and sufficient cause appearing therefor, it is hereby

## ORDERED, ADJUDGED, AND DECREED THAT:

- 1. The Motion is GRANTED on an interim basis, as set forth herein.
- 2. The Agreement, attached as Exhibit A to the Motion, is approved on an interim basis and the Debtors are authorized and directed to perform under the terms of the Agreement as of the Petition Date pursuant to sections 105(a) and 363(b) of the Bankruptcy Code, unless and until the Agreement is terminated under its own terms or pursuant to the terms of this Order, pending a final hearing on the Motion.
- 3. The Debtors and AFI are bound by all of the terms and conditions of the Agreement as of the Petition Date pending a final hearing on the Motion.
- 4. AFI shall be granted limited relief from the automatic stay to the extent required to effectuate the terms and conditions of the Agreement and this Order, including to terminate the Agreement or any Service (as defined in the Agreement) provided thereunder and to transmit any notices required under the Agreement.
- 5. Notwithstanding any provision in the Agreement or this Order to the contrary, AFI and/or the Debtors may immediately terminate the Agreement without further relief from the Court upon (a) entry by the Court of an order appointing a trustee or an examiner with expanded powers, (b) acceleration of the obligations due under the Debtors' debtor in possession credit facility, (c) conversion of any of the Debtors' Chapter 11 cases to a case under Chapter 7 of the Bankruptcy Code, (d) the failure of the final order substantially in the form attached to the Motion as Exhibit B (the "Final Order") to be entered by the Court within fifty

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 119 of 126

(50) days of the Petition Date, or (e) this Order or the Final Order having been stayed, reversed or otherwise rendered ineffective.

- 6. Notwithstanding any provision in the Bankruptcy Code, including under section 365 of the Bankruptcy Code, to the contrary, the Debtors may not, without the consent of AFI, assign the Agreement.
- The Debtors shall serve a copy of the Motion and this Interim Order by United States mail, first class postage, on (a) the Office of the United States Trustee for the Southern District of New York; (b) the office of the United States Attorney General; (c) the office of the New York Attorney General; (d) the office of the United States Attorney for the Southern District of New York; (e) the Internal Revenue Service; (f) the Securities and Exchange Commission; (g) each of the Debtors' prepetition lenders, or their agents, if applicable, (h) each of the indenture trustees for the Debtors' outstanding notes issuances; (i) Ally Financial Inc. and its counsel; (j) counsel for the administrative agent for the Debtors' proposed providers of debtor in possession financing; (k) Nationstar Mortgage LLC and its counsel; and (l) the parties included on the Debtors' list of fifty (50) largest unsecured creditors, within two (2) days of entry of this Order.
- 8. The Final Hearing, if required, to consider the Motion and proposed final order is scheduled for \_\_\_\_\_\_\_, 2012 at \_\_:\_\_\_\_.m. (prevailing Eastern Time) before the Court. Any objections or responses to the Motion must be filed with the Clerk of the Bankruptcy Court and served upon and received by: (proposed counsel for the Debtors, Morrison & Foerster LLP, 1290 Avenue of the Americas, New York, NY 10104 (Attn: Larren M. Nashelsky, Gary S. Lee, and Lorenzo Marinuzzi); (b) the Office of the United States Trustee for the Southern District of New York, 33 Whitehall Street, 21st Floor, New York, NY 10004 (Attn: Tracy Hope Davis,

- 9. The Court finds and determines that the requirements of Bankruptcy Rule 6003 are satisfied and that the relief requested is necessary to avoid immediate and irreparable harm.
  - 10. The requirements set forth in Bankruptcy Rule 6004(a) are satisfied.
- 11. Notwithstanding Bankruptcy Rule 6004(h), to the extent applicable, this Order shall be effective and enforceable immediately upon entry hereof.
- modify or affect the terms and provisions of, nor the rights and obligations under, (a) the Board of Governors of the Federal Reserve System Consent Order, dated April 13, 2011, by and among AFI, Ally Bank, ResCap, GMAC Mortgage, LLC, the Board of Governors of the Federal Reserve System, and the Federal Deposit Insurance Corporation, (b) the consent judgment entered by the District Court for the District of Columbia, dated February 9, 2012, (c) the Order of Assessment of a Civil Money Penalty Issued Upon Consent Pursuant to the Federal Deposit Insurance Act, as amended, dated February 10, 2012, and (d) all related agreements with AFI and Ally Bank and their respective subsidiaries and affiliates (excluding ResCap and its subsidiaries).

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 121 of 126

- 13. The Debtors are authorized to take all actions necessary to effectuate the relief granted pursuant to this Order in accordance with the Motion.
- 14. The relief granted by this Order shall apply to any Future Debtor in these jointly-administered cases.
- 15. This Court shall retain jurisdiction with respect to all matters relating to the interpretation or implementation of this Order.

Dated:	, 2012								
	New York, New York								
		UNITED STATES BANKRUPTCY JUDGE							

## EXHIBIT D

UNITED STATES BANKRUPTCY COUR	T.
SOUTHERN DISTRICT OF NEW YORK	

	)	
In re:	)	Case No. 12-
	)	
RESIDENTIAL CAPITAL, LLC, et al.,	)	Chapter 11
	)	-
Debtors.	)	Jointly Administered
	)	•

FINAL ORDER UNDER BANKRUPTCY CODE SECTIONS 105(a) AND 363(b) AUTHORIZING RESIDENTIAL CAPITAL, LLC TO ENTER INTO A SHARED SERVICES AGREEMENT WITH ALLY FINANCIAL INC. NUNC PRO TUNC TO THE PETITION DATE FOR THE CONTINUED RECEIPT AND PROVISION OF SHARED SERVICES NECESSARY FOR THE OPERATION OF THE DEBTORS' BUSINESSES

Upon the motion (the "Motion")<sup>1</sup> of the Debtors for entry of interim and final orders, under Bankruptcy Code sections 105(a) and 363(b), authorizing the Debtors to enter into a shared services agreement (the "Agreement") and with Ally Financial Inc. ("AFI") *nunc pro tunc* to the Petition Date for the continued receipt and provision of shared services necessary for the operation of the Debtors' businesses; and upon the Whitlinger Affidavit; and the Court having entered an interim order (the "Interim Order") on May \_\_\_, 2012 granting the Motion on an interim basis; and it appearing that this Court has jurisdiction to consider the Motion pursuant to 28 U.S.C. §§ 157 and 1334; and it appearing that venue of these Chapter 11 cases and the Motion in this district is proper pursuant to 28 U.S.C. §§ 1408 and 1409; and it appearing that this proceeding on the Motion is a core proceeding pursuant to 28 U.S.C. § 157(b); and sufficient notice of the Motion having been given; and it appearing that no other or further notice need be provided; and upon the record of the Final Hearing; and it appearing that the relief requested by

Capitalized terms not otherwise defined herein shall have the meanings ascribed to such terms in the Motion. Creditors and parties-in-interest with questions or concerns regarding the relief granted herein may refer to <a href="http://www.kccllc.net/rescap">http://www.kccllc.net/rescap</a> for additional information.

the Motion is in the best interests of the Debtors' estates, their creditors, and other parties in interest; and after due deliberation thereon; and sufficient cause appearing therefore, it is hereby

## ORDERED, ADJUDGED, AND DECREED THAT:

- 1. The Motion is GRANTED, as set forth herein.
- 2. The Agreement in the form attached hereto as Exhibit A is approved and the Debtors are authorized and directed to enter into the Agreement *nunc pro tunc* to the Petition Date pursuant to sections 105(a) and 363(b) of the Bankruptcy Code.
- 3. The Debtors are authorized to perform under the terms of the Agreement until the Agreement is terminated under its own terms or pursuant to the terms of this Order.
- The Debtors and AFI are bound by all of the terms and conditions of the Agreement.
- 5. AFI shall be granted limited relief from the automatic stay to the extent required to effectuate the terms and conditions of the Agreement and this Order, including to terminate the Agreement or any Service (as defined in the Agreement) provided thereunder and to transmit any notices required under the Agreement.
- 6. Notwithstanding any provision in the Agreement or this Order to the contrary, AFI and/or the Debtors may immediately terminate the Agreement without further relief from the Court upon (a) entry by the Court of an order appointing a trustee or an examiner with expanded powers, (b) acceleration of the obligations due under the Debtors' debtor in possession credit facility, (c) conversion of any of the Debtors' Chapter 11 cases to a case under Chapter 7 of the Bankruptcy Code, or (d) this Order having been stayed, reversed or otherwise rendered ineffective.

- 7. Notwithstanding any provision in the Bankruptcy Code, including under section 365 of the Bankruptcy Code, to the contrary, the Debtors may not, without the consent of AFI, assign the Agreement.
- 8. The Debtors are authorized to execute, deliver, implement and fully perform any and all obligations, instruments, and documents and to take any and all actions reasonably necessary or appropriate to consummate the Agreement and perform any and all obligations contemplated therein, including entering into any amendments to the Agreement without further order of the Court.
- 9. Notwithstanding anything herein to the contrary, this Order shall not modify or affect the terms and provisions of, nor the rights and obligations under, (a) the Board of Governors of the Federal Reserve System Consent Order, dated April 13, 2011, by and among AFI, Ally Bank, ResCap, GMAC Mortgage, LLC, the Board of Governors of the Federal Reserve System, and the Federal Deposit Insurance Corporation, (b) the consent judgment entered by the District Court for the District of Columbia, dated February 9, 2012, (c) the Order of Assessment of a Civil Money Penalty Issued Upon Consent Pursuant to the Federal Deposit Insurance Act, as amended, dated February 10, 2012, and (d) all related agreements with AFI and Ally Bank and their respective subsidiaries and affiliates (excluding ResCap and its subsidiaries).
  - 10. The requirements of 6004(a) are satisfied.
- 11. Notwithstanding Bankruptcy Rule 6004(h), to the extent applicable, this Final Order shall be effective and enforceable immediately upon entry hereof.
- 12. The relief granted by this Order shall apply to any Future Debtor in these jointly-administered cases.

	13.	This Court shall retain jurisdiction with respect to all matters relating to
the interpreta	ation or	implementation of this Order.
Dated:	New	York, New York
		LINITED STATES DANIEDIDTOV HIDGE

12-12020-mg Doc 41 Filed 05/14/12 Entered 05/14/12 14:50:46 Main Document Pg 126 of 126